

Chow–Witt Groups, Tensor Triangular Geometry, and Milnor–Witt Motives

Zur Erlangung des akademischen Grades Dr. rer. nat. dem Fachbereich Mathematik und Naturwissenschaften der Bergischen Universität Wuppertal vorgelegte

Dissertation

von

Jule Marie Hänel

aus Bonn

2024

Can't is a word none should speak without blushing. Edgar Albert Guest

Contents

1	Intr	oduction	7		
	1.1	Motivation	7		
	1.2	Outline	8		
	1.3	Acknowledgements	10		
2	Tensor Triangular Geometry and Dualities				
	2.1	Foundations	13		
	2.2	Rigid Tensor Triangulated Categories	19		
	2.3	Dimension Functions and Filtrations	22		
3	Grothendieck–Witt and Witt Groups				
	3.1	(Grothendieck–)Witt Groups of Triangulated Categories	27		
	3.2	(Grothendieck–)Witt Groups of Exact Categories	30		
	3.3	Localization	35		
4	Tensor Triangular Chow–Witt Groups				
	4.1	Tensor Triangular Chow Groups	39		
	4.2	Chow–Witt Groups of a Scheme	44		
	4.3	Tensor Triangular Chow–Witt Groups	51		
		4.3.1 Idea	51		
		4.3.2 Construction and Definition	53		
	4.4	Agreement with Algebraic Geometry	54		
		4.4.1 Reformulation of Classical Oriented Cocycles	55		
		4.4.2 Reformulation of Classical Oriented Coboundaries	56		
		4.4.3 Agreement of Chow–Witt Groups	60		
	4.5	Functoriality	65		
	4.6	Chow(–Witt) Groups for Tensor Triangular Categories with Models	69		
5	Cor	nputations in Modular Representation Theory	73		
	5.1	The Stable Module Category	73		
		5.1.1 Recollection on Group Rings	73		
		5.1.2 Construction of kG -stab	75		
	5.2	The (Grothendieck–)Witt Group of kG -stab	79		
	5.3	Tensor Triangular Chow–Witt Groups for $G = \mathbb{Z}/p^n\mathbb{Z}$	83		
	5.4	Tensor Triangular Chow Groups for $G = \mathbb{Z}/p\mathbb{Z} \times \mathbb{Z}/p\mathbb{Z}$	87		

	5.5	Tensor Triangular Chow–Witt Groups for $G = \mathbb{Z}/p\mathbb{Z} \times \mathbb{Z}/p\mathbb{Z}$	91		
6	The	Spectrum of Artin Motives	97		
	6.1	The Derived Category of Motives	97		
	6.2	Artin Motives, Mackey Functors, and Permutation Modules	101		
7	The	Derived Category of Artin Milnor–Witt Motives	107		
	7.1	Milnor–Witt Motives	107		
		7.1.1 Milnor–Witt Correspondences	108		
		7.1.2 Construction of the Derived Category of Milnor–Witt Motive	s112		
	7.2	Sheaves with Milnor–Witt Transfers on Zero-Dimensional Schemes	114		
	7.3	Artin Milnor–Witt Motives	116		
8	Cohomological Milnor–Witt Mackey Functors 12				
	8.1	Span Categories	123		
	8.2	Milnor-Witt Mackey Functors	136		
9	The	Balmer Spectrum of Artin Milnor–Witt Motives	139		
	9.1	Algebraically Closed Fields and First Computations	139		
	9.2	The Base Field \mathbb{R} and Coefficients in \mathbb{F}_2 \ldots \ldots \ldots \ldots	141		
Bi	Bibliography				

Chapter 1

Introduction

1.1 Motivation

Chow-Witt groups were introduced by Barge and Morel in [BM00]. The motivating question when a projective A-module is isomorphic to the direct sum $Q \oplus A$ for some projective A-module Q led to the construction of the Euler class $\tilde{c}_n(E)$ of a vector bundle E. It is a refinement of the top Chern class $c_n(E)$ in the theory of Chow groups, see e.g. [Ful98]. To answer the above question, one first asks whether this Euler class is 0, see [BM00, Fas08].

Let X be a regular scheme. The refinement from Chow groups CH(X) to Chow–Witt groups $\widetilde{CH}(X)$ includes replacing Milnor K-groups at certain points of the construction with Minor–Witt K-groups. In degree zero, this is equivalent to replacing K_0 -groups by Grothendieck–Witt groups GW.

We apply the idea of such refinements, i.e. of "decorating with tilde", to the construction of tensor triangular Chow groups $CH_{\Delta}(\mathcal{T})$ introduced in [Bal13] to define tensor triangular Chow-Witt groups $\widetilde{CH}_{\Delta}(\mathcal{T})$, where \mathcal{T} is a sufficiently "nice" tensor triangulated category. This way, we obtain a generalization of tensor triangular Chow groups, but, on the other hand, these groups generalize the definition of classical Chow-Witt groups of a scheme as introduced in [BM00]. This refinement clears the way for new areas apart from algebraic geometry. Thus, it is only natural to ask what Chow-Witt groups of tensor triangulated categories such as the stable module category kG-stab might be for initial cases of k and G. In future work, a refinement of the Euler class would be an interesting direction. For this, the notion of a vector bundle would have to be generalized to the tensor triangular setting first.

In [BC+22], Bachmann, Calmès, Déglise, Fasel, and Østvær refine Voevodsky's derived category of motives DM(k; R) (see [MVW06, Voe00]) and construct the *derived category of Milnor–Witt motives* $\overline{DM}(k; R)$. In a first attempt to understand the structure of this category, restriction to Milnor–Witt motives of 0-dimensional schemes – the *Artin Milnor–Witt motives* – will already yield interesting results that are motivated by the non-oriented results from [BG23a]. These results then serve as a basis to compute first Balmer spectra of subcategories of DM(k; R), in this case of the subcategory of Artin Milnor–Witt motives. Other subcategories are promising topics for future work. For instance, first steps in the direction of Tate Milnor–Witt motives are taken in [FY23].

1.2 Outline

We provide an outline of this work.

To begin with, **Chapter 2** introduces the reader to Balmer's tensor triangular geometry ([Bal02, Bal05b, Bal07]). Basic concepts of rigid tensor triangulated categories, such as the Balmer spectrum and filtration by dimension of support, are recalled. In particular, we prove that certain constructions such as shifted dualities, restriction to filtration components, idempotent completion, and a decomposition statement behave well with duality (e.g. Lemma 2.3.5, Lemma 2.3.7). Moreover, we recall that a rigid tensor triangulated category can be turned into a triangulated category with duality (Example 2.2.6 (i)).

In Chapter 3, the theory of (Grothendieck–)Witt groups in the setting of triangulated categories with duality ([Bal00, Wal03, Sch17]) and of exact categories with duality (and weak equivalences, respectively) ([Sch10a, Sch10b, FS09]) is recalled. We focus on the various localization statements by Balmer and Walter, as well as by Schlichting, Fasel, and Srinivas, where the latter two proved these sequences of the different settings to be equivalent (Theorem 3.3.4).

Building the core of this work's first part, Chapter 4 gives the central construction of tensor triangular Chow–Witt groups after tensor triangular Chow groups ([Bal13, Kl16a]) and classical Chow–Witt groups ([BM00, Fas08]) are recalled. In Section 4.3, tensor triangular oriented cocycles, coboundaries, and finally Chow–Witt groups CH_{Δ} are defined (Definition 4.3.3, Definition 4.3.5) using the localization sequence of (Grothendieck–)Witt groups of triangulated categories with duality. The central theorem (Theorem 4.4.5) then follows mandatorily in Section 4.4: We show that for the rigid tensor triangulated category $D^{perf}(X)$, X being a regular scheme, its tensor triangular Chow–Witt group coincides with the classical Chow–Witt group of X introduced by Barge and Morel. The main tool here will be the comparison between the filtration by dimension of support in the tensor triangular setting and the filtration by codimension of homological support in the algebro-geometric setting. After presenting a computation for X = Spec(k)by hand, we prove functorial properties of the given construction in Section 4.5. Dropping some assumptions on the tensor triangulated category, one can obtain a more general definition of tensor triangular Chow–Witt groups, but at the price of introducing models. We sketch this idea in Section 4.6.

Tensor triangular geometry grants, in particular, access to the field of representation theory. Hence, **Chapter 5** discusses the application of tensor triangular Chow–Witt groups to the stable module category kG-stab for a field k dividing the order of G. At first, basic notions and results are recalled. In Section 5.2, the most prominent result is the computation of the Grothendieck Witt group of kG-stab as GW(kG-mod) divided by the subgroup generated by projective kG-modules (Theorem 5.2.3). Deriving a dévissage result from various results in the literature (Proposition 5.2.6), we will moreover be able to reduce the calculation to a quotient of the Grothendieck–Witt group of the base field. We prove the same results for Witt groups.

Let $p \neq 2$ and k of characteristic p. The first concrete example of $\operatorname{CH}_{\Delta}(kG\operatorname{-stab})$ treats the cyclic group $G = \mathbb{Z}/p^n\mathbb{Z}$. Using previous results, we show that the 0-th tensor triangular Chow–Witt group of $kG\operatorname{-stab}$ is $\mathbb{Z}/p^n\mathbb{Z}$ and vanishes in all other degrees if k is finite or algebraically closed (Example 5.3.3, Example 5.3.4). In particular, in these cases, Chow and Chow–Witt groups coincide. For $G = \mathbb{Z}/p\mathbb{Z} \times \mathbb{Z}/p\mathbb{Z}$, the 0-th Chow–Witt group is $\mathbb{Z}/p^2\mathbb{Z}$ and all degrees $\neq 1$ vanish if k is algebraically closed or finite (Example 5.5.2, Example 5.5.3), just as in the case of Chow groups. For degree 1, we give partial results.

From **Chapter 6** on, we dive into the world of motives. First, Section 6.1 reviews the construction of Voevodsky's derived category of motives DM(k; R) following [MVW06, Voe00]. Then, we recall the main results and tools of [BG23a] in Section 6.2, which essentially breaks down to the equivalence of the derived categories of Artin motives DAM(k; R), of permutation modules $D Perm(\Gamma; R)$, and of cohomological Mackey functors $D(Mack_R^{coh}(\Gamma))$ (Corollary 6.2.12).

In Chapter 7, we begin to set up the stage to refine the results from [BG23a] to the oriented setting. For this, Section 7.1 recalls the construction of the derived category of Milnor–Witt motives $\widetilde{DM}(k; R)$ from [BC+22] before investigating the special case of finite Milnor–Witt correspondences generated by 0-dimensional schemes \widetilde{Cor}_{k}^{0} in Section 7.2. Then, Section 7.3 establishes a first refinement of [BG23a], namely the equivalence between the tensor triangulated categories $\widetilde{DAM}^{gm}(k; R)$ and $K_b((\widetilde{Cor}_{k}^{0})^{\natural})$ (Proposition 7.3.3). The general version for non-geometric motives is shown in the case that Conjecture 7.3.10 holds (Corollary 7.3.12).

The remaining equivalence fitting into the diagram

$$\widetilde{\mathrm{DAM}}(k; R) \xleftarrow{\simeq} \mathrm{D}(\mathrm{Sh}_{\mathrm{Nis}}(\widetilde{\mathrm{Cor}}_{k,R}^{0}))$$
$$\downarrow \simeq$$
$$\mathrm{D}(\widetilde{\mathrm{Mack}}_{R}^{\mathrm{coh}}(k))$$

is established in **Chapter 8**. First, Section 8.1 introduces the category $\widetilde{\Omega}_R(k)$, which is closely related to classical span categories (Construction 8.1.2). Using the construction of $\widetilde{\Omega}_R(k)$, the category of (cohomological) Milnor–Witt Mackey functors in the algebro-geometric setting $\widetilde{\text{Mack}}_R^{\text{coh}}(k)$ is defined (Definition 8.2.1) and the vertical equivalence (Remark 8.2.4) is shown in Section 8.2 under the assumption that a certain factorization is unique (Conjecture 8.2.2).

In **Chapter 9**, examples of the Balmer spectrum of $\widetilde{DAM}^{gm}(k; R)$ are computed. If k is an algebraically closed field, the spectrum is shown to coincide with the spectrum of $DAM^{gm}(k; R)$, which is merely a point (Corollary 9.1.2). In Section 9.2, we then present first steps in the computation of the spectrum of $\widetilde{DAM}^{gm}(k; K)$ for $k = \mathbb{R}$ and K a field of characteristic 2. In particular, we show that it contains the tensor prime ideal $\langle \widetilde{M}(\mathbb{C}) \rangle$, which denotes the thick tensor ideal generated by the complex given by $\operatorname{Spec}(\mathbb{C})$ concentrated in degree 0 (Lemma 9.2.8).

1.3 Acknowledgements

First of all, I would like to thank my supervisor Jens Hornbostel for supporting and guiding me through this challenging time. I sincerely appreciate his endless patience, his quick and detailed feedback, his understanding, and the experience I gained through him in teaching. Moreover, as my very first professor of mathematics, he lit the spark that led me down this path, for which I am very grateful.

I am grateful to Paul Balmer for commenting on a draft of Section 3.3, sharing ideas of future topics in the field of tensor triangular geometry, as well as for many friendly conversations.

I am grateful to Dave Benson for discussions on the connection between dimensions of kG-modules and the characteristic of k as well as for giving an inspiration for examples concerning kG-modules of different complexity.

I am grateful to Jean Fasel for clarifying details about the connection between classical and Milnor–Witt correspondences.

I am grateful to Martin Gallauer for giving details about the self-duality of permutation modules, adjoint functors, and for sharing his opinions on the role of philosophy in mathematics. Moreover, I appreciate his explanations on \mathbb{A}^1 -local objects and valuable further comments.

I am grateful to Fernando Muro for welcoming me to Seville and for interesting discussions on algebraic models. My stay was valuable and I appreciate the hospitality I experienced during this time.

I am grateful to my other committee members Kay Rülling, Britta Späth, Matthias Wendt, and Marcus Zibrowius for taking the time to asses my work.

To my colleagues Jan Hennig, Henry July, Andrea Lachmann, Georg Linden, Tommy Lundemo, and Herman Rohrbach I owe great thanks for reading parts of my work and/or always having had the patience to listen. Moreover, I would like to thank my former colleague Thomas Hudson for uplifting chats and teaching advice.

As an associated member, I would like to thank the principal investigators and other members of the *GRK 2240: Algebro-Geometric Methods in Algebra*, *Arithmetic, and Topology* for their support.

Having been part of the *Selma Meyer Mentoring Program*, I have experienced outstanding support. I would like to thank my mentors Birgit Jacob and Stephan

Kulka, the organizer Ekaterina Masetkina, and my colleagues, in particular, my peer support group.

Mama, thank you for being my greatest fan and my best friend. I know I can always count on you. Without you, I would not have been able to pull this through.

Papa, I am sorry for taking just a tad too long for you to read this. I know you would be proud of me now, because you always were. Your love, shown in your very own way, has always empowered me to do great things. I really wish I could share this moment with you.

I would also like to thank my partner Jens Bickmann for supporting me and enduring my mood swings during the demanding and difficult final year. Thank you for empowering me, leaving me the freedom of my own pace, handing me half of the spirit fruit on my low days, and sharing enthusiasm about every small step of advancement.

Finally, special thanks go to my closest friends, my hockey family, my brother Ben, my coworking group, my other colleagues, and my students.

Chapter 2

Tensor Triangular Geometry and Dualities

In this chapter, we first review Balmer's tensor triangular geometry before explaining how the concept of dualities can be included. The main references for most of the contents of this chapter are [Bal02, Bal05b, Bal07, Bal10b]. We refer to [GN03] for the concept of a dualizing pairing. For readers interested in the foundations of triangulated categories, see for example [Wei95], whereas readers who would like to read more about monoidal structures are referred to [MacL71, Chapter VII].

Convention 2.0.1. Categories we consider in this work will always assumed to be essentially small.

2.1 Foundations

First, we define the objects of study, namely the *tensor triangulated categories* and functors between them.

Definition 2.1.1. (i) A tensor triangulated category $\mathcal{T} = (\mathcal{T}, \otimes, \mathbb{1})$ is a triangulated category together with a symmetric monoidal structure $\otimes : \mathcal{T} \times \mathcal{T} \to \mathcal{T}$ with unit object $\mathbb{1}$ such that \otimes is triangulated in both variables. We usually denote the translation functor of \mathcal{T} by T.

In particular, for every a, b in \mathcal{T} , we have natural isomorphisms $T(a) \otimes b \cong T(a \otimes b)$ and $a \otimes T(b) \cong T(a \otimes b)$ such that $T(a) \otimes T(b) \cong T(T(a) \otimes b) \cong T^2(a \otimes b)$ and $T(a) \otimes T(b) \cong T(a \otimes T(b)) \cong T^2(a \otimes b)$ only differ by sign.

(ii) A triangulated functor $F : \mathcal{T} \to \mathcal{S}$ between tensor triangulated categories \mathcal{T} and \mathcal{S} is called a *tensor triangulated functor* if it commutes with the tensor product up to isomorphism and fulfills $F(\mathbb{1}_{\mathcal{T}}) = \mathbb{1}_{\mathcal{S}}$.

Remark 2.1.2. In [May01, Chapter 4], May gives a different definition of a tensor triangulated category under the name *closed monoidal category with "compatible"*

triangulation; also, Hovey, Palmieri, and Strickland give a different definition in [HPS97, Definition A.2.1]. The equivalence of the latter to May's axioms (TC1) and (TC2) is shown in [May01, Remark 4.3], but he established additional axioms (T3)-(T5) to carry more information about the product. The axioms (TC1) and (TC2) also suffice for our purposes and are equivalent to Balmer's definition above, see [Bal10b, Remark 4].

We now explain how to endow triangulated categories with a duality structure. For this, we need the following type of functors.

Definition 2.1.3. Let $\delta = \pm 1$ and \mathcal{C}, \mathcal{D} be triangulated categories. A contravariant δ -triangulated functor $F : \mathcal{C} \to \mathcal{D}$ is a contravariant additive functor satisfying $FT_{\mathcal{C}}^{-1} = T_{\mathcal{D}}F$ such that if

$$A \xrightarrow{u} B \xrightarrow{v} C \xrightarrow{w} T_{\mathcal{C}}(A)$$

is an exact triangle in \mathcal{C} , then

 $FC \xrightarrow{Fv} FB \xrightarrow{Fu} FA \xrightarrow{\delta \cdot T_{\mathcal{D}}(Fw)} T_{\mathcal{D}}(FC)$

is an exact triangle in \mathcal{D} .

Now, we can define *triangulated categories with duality* and specify functors between such.

- **Definition 2.1.4.** (i) Let $\delta = \pm 1$. A triangulated category with δ -duality $(\mathcal{T}, *, \delta, \eta)$ is a triangulated category \mathcal{T} together with a δ -triangulated functor $*: \mathcal{T}^{\text{op}} \to \mathcal{T}$ and an isomorphism of triangulated functors $\eta: \operatorname{id}_{\mathcal{T}} \stackrel{\cong}{\to} * \circ *$ such that $\eta_A^* \circ \eta_{A^*} = \operatorname{id}_{A^*}$ and $\eta_{T(A)} = T(\eta_A)$ for all objects A of \mathcal{T} . Here, * is called the *duality functor* and η is called the *double dual identification*.
 - (ii) A duality-preserving functor between triangulated categories with dualities $(\mathcal{T}, *, \delta, \eta)$ and $(\mathcal{T}', *', \delta', \eta')$ is a pair (F, μ) with $F : \mathcal{T} \to \mathcal{T}'$ a triangulated functor and $\mu : F \circ * \to *' \circ F$ an isomorphism of triangulated functors such that the diagram

$$\begin{array}{c} FA \xrightarrow{F\eta_A} F(A^{**}) \\ \downarrow^{\eta'_{FA}} \downarrow & \downarrow^{\mu_{A^*}} \\ (FA)^{*'*'} \xrightarrow{\mu_A^{*'}} F(A^*)^{*'} \end{array}$$

commutes for all objects A in \mathcal{T} .

Notation 2.1.5. We will sometimes write $(\mathcal{T}, *, \eta)$ or even $(\mathcal{T}, *)$ instead of $(\mathcal{T}, *, \delta, \eta)$ when δ and η are clear from the context (or $\delta = 1$). Often we will say *triangulated category with duality* and omit δ from the name.

Remark 2.1.6. The notion of the unit element 1 is defined in detail in [MacL71] as an element together with natural isomorphisms α , β such that, for every object

a, we have $a \otimes \mathbb{1} \stackrel{\alpha}{\cong} a$ and $\mathbb{1} \otimes a \stackrel{\beta}{\cong} a$. Here, a unit element is unique up to (not necessarily unique) isomorphism: Suppose there is another unit $\mathbb{1}'$ together with isomorphisms α' and β' as above. Then, $\mathbb{1} \stackrel{\alpha'}{\cong} \mathbb{1} \otimes \mathbb{1}' \stackrel{\beta}{\cong} \mathbb{1}'$ and $\mathbb{1} \stackrel{\beta'}{\cong} \mathbb{1}' \otimes \mathbb{1} \stackrel{\alpha}{\cong} \mathbb{1}'$.

Remark 2.1.7. There are also related notions such as *abelian categories with duality* or *exact categories with duality* which we will encounter later. In general, the concept of duality includes a duality functor compatible in a certain sense with the given structure and a double dual identification, see [Sch10a, Definition 3.1] for the definition of a *category with duality*.

- **Example 2.1.8.** (i) In many cases, the duality we consider on triangulated categories will be (derived from) $\operatorname{Hom}(-, 1)$, such as in the derived category of finitely generated *R*-modules for 1 = R, the derived category of perfect complexes with $1 = \mathcal{O}_X$, where *X* is a regular scheme (see [Sch10a, Examples 2.2 and 2.3], compare Remark 2.1.19), or the category of *kG*-modules with 1 = k. More details on these particular triangulated categories with duality will follow in Section 2.2.
 - (ii) Note that $\operatorname{Hom}_R(-, R)$ does, in general, *not* equip the category *R*-Mod of *R*-modules with a duality since we do not have a double dual identification in general. For instance, if R = k is a field, the cardinality of the dual (and therefore also the double dual) of an infinite-dimensional vector space *V* is higher than its own cardinality. Hence, $(V^*)^* \ncong V$.

Neither does $\operatorname{Hom}_R(-, R)$ equip the category of finite length R-modules with a duality in general. Let $R = \mathbb{Z}$ and n > 1. Then, $\operatorname{Hom}_{\mathbb{Z}}(\mathbb{Z}/n\mathbb{Z},\mathbb{Z})$ contains only the zero morphism. To see this, take a \mathbb{Z} -module homomorphism $f : \mathbb{Z}/n\mathbb{Z} \to \mathbb{Z}$. We have

$$0 = f(0) = f(n) = n \cdot f(1),$$

and, since \mathbb{Z} is an integral domain, f(1) = 0. As a consequence, $((\mathbb{Z}/n\mathbb{Z})^*)^* = (\{0\})^* = \{0\} \not\cong \mathbb{Z}/n\mathbb{Z}$.

(iii) If A is a local 0-dimensional ring that is a finite dimensional k-algebra for some field k, $\operatorname{Hom}_A(\operatorname{Hom}_A(M, A), A)) \ncong M$ in general, which can be fixed by choosing the duality $\operatorname{Hom}_k(-, k)$; for more details see [Eis95, Chapter 21].

Remark 2.1.9. Note that [CH09] denotes a strong duality a duality such that η from Definition 2.1.4 is an isomorphism, but we assume all triangulated categories to be equipped with such a strong duality. To understand the difference between Definition 2.1.4 and [CH09, Definition 2.1.1] in detail, the reader is referred to [CH09, Remark 2.1.2].

The following lemma is straightforward.

Lemma 2.1.10. Let $F : (\mathcal{T}, *) \to (\mathcal{L}, *')$ be a duality-preserving functor between triangulated categories with duality. Let \mathcal{J} be a triangulated subcategory of \mathcal{T} that

we consider a triangulated category with duality when equipped with the restricted duality. Let $F(\mathcal{J}) \subset \mathcal{K}$ for some triangulated subcategory of \mathcal{L} , considered a triangulated category with restricted duality. Then,

- (i) The restriction $F|_{\mathcal{J}} : \mathcal{J} \to \mathcal{L}$ is duality-preserving.
- (ii) The induced duality turns the Verdier quotient \mathcal{T}/\mathcal{J} into a triangulated category with duality.
- (iii) The functor on the quotients $\mathcal{T}/\mathcal{J} \to \mathcal{L}/\mathcal{K}$ is duality-preserving.

Now, we want to introduce *tensor triangulated categories with duality*. For this, we use the following notion from [GN03].

Definition 2.1.11. A product between triangulated categories \mathcal{K} and \mathcal{L} with codomain \mathcal{M} is a bi-covariant functor $\otimes : \mathcal{K} \times \mathcal{L} \to \mathcal{M}$ such that

- (i) the functor \otimes is 1-triangulated in both variables and
- (ii) the following diagram is skew-commutative for A in \mathcal{K} and B in \mathcal{L} , i.e. $T_{\mathcal{M}}(\mathfrak{r}_{A,B}) \circ \mathfrak{l}_{A,T(B)} = -T_{\mathcal{M}}(\mathfrak{l}_{A,B}) \circ \mathfrak{r}_{T(A),B}$

where $\mathfrak{r}_{A,B}: A \otimes T_{\mathcal{L}}(B) \xrightarrow{\cong} T_{\mathcal{M}}(A \otimes B)$ and $\mathfrak{l}_{A,B}: T_{\mathcal{K}}(A) \otimes B \xrightarrow{\cong} T_{\mathcal{M}}(A \otimes B)$.

Definition 2.1.12. Let \mathcal{K}, \mathcal{L} , and \mathcal{M} be triangulated categories with dualities and $\otimes : \mathcal{K} \times \mathcal{L} \to \mathcal{M}$ a product between their underlying triangulated categories in the sense of Definition 2.1.11. It is called a *dualizing pairing* between \mathcal{K}, \mathcal{L} , and \mathcal{M} if for all objects A in \mathcal{K} and B in \mathcal{L} there are isomorphisms $\mu_{A,B} : A^{*\kappa} \otimes B^{*\varsigma} \xrightarrow{\cong} (A \otimes B)^{*\mathcal{M}}$ functorial in A and B which make the following diagrams

$$\begin{array}{c} A \otimes B \xrightarrow{\eta_A^{\mathcal{K}} \otimes \eta_B^{\mathcal{L}}} A^{*\kappa*\kappa} \otimes B^{*\mathcal{L}*\mathcal{L}} \\ \eta_{A \otimes B}^{\mathcal{M}} \downarrow & \downarrow^{\mu_{A^*,B^*}} \\ (A \otimes B)^{*\mathcal{M}*\mathcal{M}} \xrightarrow{(\mu_{A,B})^{*\mathcal{M}}} (A^{*\mathcal{M}} \otimes B^{*\mathcal{L}})^{*\mathcal{M}} \end{array}$$

and

commute, where η denotes the double dual identification.

Definition 2.1.13. Let $\delta = \pm 1$. The tuple $(\mathcal{T}, \otimes, \mathbb{1}, *, \eta)$ is called a *tensor trian*gulated category with δ -duality if

- (i) $(\mathcal{T}, \otimes, \mathbb{1})$ is a tensor triangulated category as in Definition 2.1.1,
- (ii) $(\mathcal{T}, *, \eta, \delta)$ is a triangulated category with δ -duality as in Definition 2.1.4,
- (iii) $\otimes : (\mathcal{T}, *, \eta) \times (\mathcal{T}, *, \eta) \rightarrow (\mathcal{T}, *, \eta)$ is a dualizing pairing as in Definition 2.1.12.

Remark 2.1.14. This definition is more than we actually need later. In Subsection 4.3.2, we will merely need rigid tensor triangulated categories that can be considered triangulated categories with dualities without assuming a compatibility in the sense of a dualizing pairing.

Definition 2.1.15. A *duality-preserving functor* between tensor triangulated categories with dualities is a functor that is

- (i) a tensor functor as in Definition 2.1.1 (ii) when considered a functor between the underlying tensor categories, and
- (ii) a duality-preserving functor as in Definition 2.1.4 (ii) when considered a functor between the underlying triangulated categories with duality (thus, in particular, triangulated).

With the definition of tensor triangulated categories with duality at hand, we now turn to the main construction in the setting of tensor triangular geometry: the *Balmer spectrum* ([Bal05b, Definition 2.1]). The idea behind it is to generalize the notion of (prime) ideals of a ring to the categorical setting; this way, well-known methods from the algebro-geometric setting can be imitated.

Definition 2.1.16. Let $(\mathcal{T}, \otimes, \mathbb{1})$ be a tensor triangulated category. A thick triangulated subcategory $\mathcal{J} \subset \mathcal{T}$ is called a *tensor ideal* if $\mathcal{T} \otimes \mathcal{J} \subset \mathcal{J}$. It is called *prime* if $\mathcal{J} \neq \mathcal{T}$ and

$$A \otimes B \in \mathcal{J} \Rightarrow ((A \in \mathcal{J}) \text{ or } (B \in \mathcal{J})).$$

Definition-Lemma 2.1.17. For an essentially small tensor triangulated category $\mathcal{T} = (\mathcal{T}, \otimes, \mathbb{1})$ we define its (Balmer) spectrum as the set

 $\operatorname{Spc}(\mathcal{T}) := \{ \mathcal{P} \subset \mathcal{T} \mid \mathcal{P} \text{ is a tensor prime ideal} \}.$

It is a topological space by defining the basic closed subsets as the sets of the form

$$\operatorname{supp}(A) := \{ \mathcal{P} \in \operatorname{Spc}(\mathcal{T}) \mid A \notin \mathcal{P} \} \subset \operatorname{Spc}(\mathcal{T}),$$

for every object A in \mathcal{T} .

We now introduce the category $D^{\text{perf}}(X)$ (see [TT90, Section 2]), which will serve as our running example and is of vital importance when comparing classical Chow–Witt groups to the new construction of tensor triangular Chow–Witt groups in Section 4.3. **Definition 2.1.18.** ([SGA6, Définition 2.1]) Let X be a scheme. A *perfect complex* of \mathcal{O}_X -modules is a complex locally quasi-isomorphic to a bounded complex of algebraic vector bundles.

We denote by $D^{\text{perf}}(X)$ the derived category of perfect complexes equipped with the derived tensor product $\otimes_{\mathcal{O}_X}$, which is a tensor triangulated category by [Tho97, Theorem 3.15].

Remark 2.1.19. Recall that, if we have a regular noetherian scheme X, there is a categorical equivalence between $D^{perf}(X)$ and the bounded derived category $D^{b}(Coh(X)) =: D^{b}(X)$ of coherent sheaves on X.

Moreover, these categories are equivalent to $D^{b}(Vect(X))$ if X is regular, where Vect(X) is the exact category of algebraic vector bundles, i.e., of locally free sheaves, on X.

The category P(X) of coherent locally free sheaves on X is contained in Vect(X). From [Bal99], we know that for regular X there is a categorical equivalence $D^b(Coh(X)) \simeq D^b(P(X))$. (Here, *regular* means that every coherent \mathcal{O}_X -module has a finite resolution by locally free coherent \mathcal{O}_X -modules, see [Bal99, Remark 2.8]). In conclusion, there are equivalences

$$\mathrm{D}^{\mathrm{perf}}(X)\simeq\mathrm{D}^b(\mathrm{Coh}(X))\simeq\mathrm{D}^b(\mathrm{Vect}(X))\simeq\mathrm{D}^b(P(X))$$

when X is regular.

Example 2.1.20. ([Bal05b, Corollary 5.6]) Let X be a topologically noetherian scheme. There is a homeomorphism $X \cong \operatorname{Spc}(D^{\operatorname{perf}}(X))$ of the underlying topological space of X and the Balmer spectrum of $D^{\operatorname{perf}}(X)$, given explicitly by

$$\sigma: X \to \operatorname{Spc}(\operatorname{D}^{\operatorname{perf}}(X))$$
$$x \mapsto \{A^{\bullet} \in \operatorname{D}^{\operatorname{perf}}(X) \mid A_{x}^{\bullet} \cong 0 \text{ in } \operatorname{D}^{\operatorname{perf}}(\mathcal{O}_{X,x})\} \subset \operatorname{D}^{\operatorname{perf}}(X).$$

Under this homemorphism, the support $\operatorname{supp}(A^{\bullet})$ corresponds to the *homological* support $\operatorname{supp}(A^{\bullet})$, see Definition 4.2.5.

Remark 2.1.21. Tensor triangular geometry is a relatively new topic. It is a difficult problem to compute Balmer spectra, which is why there is still a wide range of open questions.

Ahead of time, before the language of tensor triangular geometry was introduced, Hopkins and Smith already determined the spectrum of the topological stable homotopy category of finite spectra SH^{fin} in [HS98] (compare [Bal10a, Corollary 9.5]).

Many results in this direction are due to Paul Balmer. For example, the spectrum of the bounded derived category of kG-modules was shown to be equivalent to the homogeneous spectrum of the cohomology ring $H^{\bullet}(G, k)$ in [Bal05b, Corollary 5.10]. An overview of open problems is given in [Bal10b], where Balmer describes, for example, the computation of $\operatorname{Spc}(\operatorname{SH}_{gm}^{\mathbb{A}^1}(S))$ as a "long-term challenge" ([Bal10b, p. 18]). A first step is given in [HO18, Theorem 1.1].

Recent progress in the field of motivic homotopy theory includes the computations of the spectra of the derived category of Artin-Tate motives over \mathbb{R} with integral coefficients in [BG22a] by Balmer and Gallauer or the triangulated category of Tate motives with integral coefficients over certain algebraically closed base fields by Gallauer in [Gal19, Theorem 8.6]. In [BG22b] and [BG23b], a description of the spectrum of the derived category of permutation modules over a finite group with coefficients in a field of positive characteristic is given. The extension to profinite groups has recently been published in the preprint [BG24]. In [Pe13], Peter proved that the spectrum of mixed Tate motives with rational coefficients DTM(k, \mathbb{Q}) is a point.

Further results and progress in computing Balmer spectra in the field of algebraic geometry are, for example, [DS14, Theorem 4.7], [Ste14, Theorem 7.7], and [Hal16, Theorem 1.2]. In modular representation theory, examples are the spectrum of the stable module category in [BCR97]. For more examples, the reader is referred to [Bal19].

Having recalled the basic concepts of tensor triangular geometry, we can direct ourselves in more detail toward the matter of duality in the following subsection.

2.2 Rigid Tensor Triangulated Categories

In Definition 2.1.13, we introduced triangulated categories with duality. We mentioned that, in most cases, the duality we are interested in is inherited from some Hom-functor. This idea is captured by the notion of *rigid* tensor triangulated categories. These will be our main object of study and are introduced in this subsection.

Our overall goal of defining tensor triangular Chow–Witt groups is inspired by the construction of tensor triangular Chow groups by Balmer and Klein [Bal13], [Kl16a]. This idea is led by the concept of a filtration of the given tensor triangulated category by dimension of support (cf. Section 2.3)

$$\mathcal{T}_{(-\infty)} \subset ... \subset \mathcal{T}_{(n-1)} \subset \mathcal{T}_{(n)} \subset ... \subset \mathcal{T}_{(\infty)},$$

where we define $\mathcal{T}_{(n)} := \{a \in \mathcal{T} \mid \dim(\operatorname{supp}(a)) \leq n\}$ for a dimension function dim in the sense of Definition 2.3.1 and $n \in \mathbb{Z}$.

We would like to use the same approach, but, in general, supports do not behave well with duality, i.e. the support of an object might differ from the support of its dual. Balmer shows that the assumption of the tensor triangulated category being *rigid* (Definition 2.2.1) fixes this problem. He proves that, in this case, the support of an object equals the support of its dual, which is one of the reasons why we introduce this notion (see [Bal07, Chapter 2]).

Definition 2.2.1. A tensor triangulated category $(\mathcal{T}, \otimes, \mathbb{1})$ is called *rigid* if there is a bi-triangulated functor hom : $\mathcal{T}^{\text{op}} \times \mathcal{T} \to \mathcal{T}$ such that

(i) there are natural isomorphisms

$$\operatorname{Hom}_{\mathcal{T}}(a \otimes b, c) \cong \operatorname{Hom}_{\mathcal{T}}(a, \operatorname{hom}(b, c))$$

for all objects a, b, and c in \mathcal{T} , and

(ii) every object a is *rigid*, that is, the natural morphism

 $\hom(a, 1) \otimes b \to \hom(a, b)$

is an isomorphism for all objects b of \mathcal{T} .

The functor hom is often called the *internal Hom*. We denote the rigid tensor triangulated category by $(\mathcal{T}, \otimes, \mathbb{1}, \text{hom})$.

Remark 2.2.2. In [PS14, Definition 2.1], another description of rigid (or (strongly) dualizable) objects in tensor triangulated (or closed symmetric monoidal) categories is given: An object a in $(\mathcal{T}, \otimes, \mathbb{1})$ is rigid if there exists a dual a^* together with evaluation and coevaluation maps

$$\eta : \mathbb{1} \to a \otimes a^* \text{ and } \varepsilon : a^* \otimes a \to \mathbb{1}$$

satisfying

$$(\mathrm{id}_a \otimes \varepsilon)(\eta \otimes \mathrm{id}_a) = \mathrm{id}_a$$
 and $(\varepsilon \otimes \mathrm{id}_{a^*})(\mathrm{id}_{a^*} \otimes \eta) = \mathrm{id}_{a^*}$.

The definitions are equivalent (see [PS14, p.7]) in the following sense: An object a is rigid with dual hom(a, 1) in the sense of [PS14, Definition 2.1] if and only if the map

$$\hom(a, 1) \otimes a \to \hom(a, a)$$

is an isomorphism.

Remark 2.2.3. Definition 2.2.1 (i) states that (\otimes, hom) is an adjoint couple, and we call Definition 2.2.1 (i) the *adjunction isomorphism*.

As an important example, we will now show how rigid tensor triangulated categories yield triangulated categories with duality. As a first step, we recall some notions on objects L that twist dualities.

Definition 2.2.4. An object L of a tensor triangulated category $(\mathcal{T}, \otimes, \mathbb{1})$ is called \otimes -invertible if there exists an object M of \mathcal{T} and an isomorphism $L \otimes M \xrightarrow{\cong} \mathbb{1}$.

These objects satisfy some nice technical properties.

Lemma 2.2.5. Let $(\mathcal{T}, \otimes, 1, \text{hom})$ be a rigid tensor triangulated category.

- (i) If K and L are \otimes -invertible objects, then so is $K \otimes L$.
- (ii) For a \otimes -invertible object L in a (not necessarily rigid) tensor triangulated category we have $\operatorname{supp}(L) = \operatorname{Spc}(\mathcal{T})$.

Proof. Part (i) can be seen as follows: For \otimes -invertible objects K and L there exist objects M and M' such that $K \otimes M \cong 1$ resp. $L \otimes M' \cong 1$. It then follows by associativity of \otimes that $(K \otimes L) \otimes (M' \otimes M) \cong K \otimes (L \otimes M') \otimes M \cong K \otimes 1 \otimes M \cong K \otimes M \cong 1$.

For part (ii), assume that $\operatorname{supp}(L) \neq \operatorname{Spc}(\mathcal{T})$, which is equivalent to $\{\mathcal{P} \in \operatorname{Spc}(\mathcal{T}) | L \in \mathcal{P}\}$ being nonempty, i.e., there exists a tensor prime ideal $\mathcal{P} \in \operatorname{Spc}(\mathcal{T})$ such that $L \in \mathcal{P}$. Since \mathcal{P} is a thick tensor ideal we have that, in particular, $L \otimes M \in \mathcal{P}$ for its \otimes -inverse M. But since $L \otimes M \cong \mathbb{1}$, \mathcal{P} is not proper and thus cannot be a tensor prime ideal. \Box

Example 2.2.6. This example shows how to turn rigid tensor triangulated categories into triangulated categories with dualities.

(i) Let (T, ⊗, 1, hom) be a rigid tensor triangulated category. We can turn it into a triangulated category with 1-duality as follows: Set * : T^{op} → T as * := hom(-, 1), which is 1-triangulated by Definition 2.2.1. The double dual identification η : id_T → * ∘ * is defined as the image of the evaluation map

$$\operatorname{ev}: A \otimes \operatorname{hom}(A, 1) \to 1$$

under the adjunction isomorphism Definition 2.2.1 (i). The natural transformation η is indeed an isomorphism, see [HPS97, Theorem A.2.5 (b)]. The fact that $(\mathcal{T}, \hom(-, 1))$ is a triangulated category with duality follows from [CH09, Corollary 3.2.4].

- (ii) Let L be a fixed \otimes -invertible object. More generally, we can turn the rigid tensor triangulated category $(\mathcal{T}, \otimes, \mathbb{1}, \hom)$ into a triangulated category with duality by choosing the duality $D_L := \hom(-, L)$, which amounts to tensoring $\hom(-, \mathbb{1})$ with L in the setting above since $\hom(-, \mathbb{1}) \otimes L \cong \hom(-, L)$ by rigidity. The reader is referred to [CH09, Corollary 3.2.4] for more details.
- (iii) Let X be a quasi-compact, quasi-separated scheme (e.g. noetherian). The category $D^{\text{perf}}(X)$ is a rigid tensor triangulated category with the left derived tensor product $\otimes_{\mathcal{O}_X}^L$ and the right derived Hom-sheaf R Hom =: hom as internal Hom, see [Bal07, Proposition 4.1]. In particular, it can be turned into a triangulated category with duality as above. For a regular scheme X, $D^{\text{perf}}(X)$ is equivalent to $D^b(X)$ as stated in Remark 2.1.19.
- (iv) Let k be a field of positive characteristic and G a finite group whose order is divisible by char(k). Then, the stable category kG-stab is a rigid tensor triangulated category with tensor product \otimes_k and internal Hom hom := Hom_k, see [Bal07, Proposition 4.2]. A detailed construction of the category kG-stab will follow in Section 5.1.
- (v) As stated in [Bal07, Remark 2.2], rigid tensor triangulated categories often arise as subcategories of compact objects of some bigger tensor triangulated

category. For example, if X is quasi-compact and quasi-separated, $D^{\text{perf}}(X)$ is the subcategory of compact objects in the derived category $D^+_{\text{Qcoh}}(X)$ of cohomologically bounded below complexes with quasi-coherent cohomology. The latter is an example of a tensor triangulated category that is *not* rigid.

Given a triangulated category with duality, we can construct *shifted dualities*. We first recall Balmer's definition of this notion.

Definition 2.2.7. Let $(\mathcal{T}, *, \eta, \delta)$ be a triangulated category with duality. We define the associated triangulated category with *shifted duality* as

$$T(\mathcal{T}, *, \eta, \delta) := (\mathcal{T}, T \circ *, -\delta \cdot \eta, -\delta)$$

where T is the shift functor of \mathcal{T} . Moreover, we set

$$T^{-1}(\mathcal{T}, *, \eta, \delta) := (\mathcal{T}, T^{-1} \circ *, \delta \cdot \eta, \delta).$$

Remark 2.2.8. For example, if $\delta = 1$, then

$$T^{n}(\mathcal{T}, \ast, \eta, \delta) := (\mathcal{T}, T^{n} \circ \ast, (-1)^{\frac{n(n+1)}{2}} \eta, (-1)^{n} \cdot \delta)$$

is a triangulated category with $(-1)^n \delta$ -duality for $n \in \mathbb{Z}$ by [Bal00, Remark and Definition 2.9].

So, for each $n \in \mathbb{Z}$, we obtain a triangulated category with duality associated to $(\mathcal{T}, *, \eta, \delta)$. In particular, if $(\mathcal{T}, \otimes, \text{hom})$ is a rigid tensor triangulated category, $T^n(\mathcal{T}, D_L, \eta_L)$ is a triangulated category with duality for a fixed \otimes -invertible object L in \mathcal{T} .

This subsection has introduced the notion of rigid tensor triangulated categories which embodies the idea of allowing mainly Hom-dualities when talking about triangulated categories with duality. We have seen examples and further notions as a preparation to talk about filtrations by dimension of support in the next section.

2.3 Dimension Functions and Filtrations

The reason why we introduced rigid tensor triangulated categories above was that we wanted the dualities to behave well with the support, in particular, with the filtration of our tensor triangulated category we are about to construct. In order to show this, we first introduce notions of *dimension* before constructing the filtration components.

Definition 2.3.1. A dimension function on a tensor triangulated category $\mathcal{T} = (\mathcal{T}, \otimes, \mathbb{1})$ is a map

$$\dim: \operatorname{Spc}(\mathcal{T}) \to \mathbb{Z} \cup \{\pm \infty\}$$

satisfying the following properties:

(i) For tensor prime ideals $\mathcal{Q} \subset \mathcal{P}$ of \mathcal{T} , we have $\dim(\mathcal{Q}) \leq \dim(\mathcal{P})$.

(ii) If $\mathcal{Q} \subset \mathcal{P}$ and $\dim(\mathcal{Q}) = \dim(\mathcal{P})$, it follows that $\mathcal{Q} = \mathcal{P}$.

Convention 2.3.2. All categories are essentially small as in Convention 2.0.1. Moreover, let the Balmer spectrum be noetherian for any tensor triangulated category. Tensor triangulated categories will be equipped with a dimension function dim. Given a rigid tensor triangulated category ($\mathcal{T}, \otimes, 1$, hom), we fix a \otimes -invertible object L. When considered a triangulated category with duality, \mathcal{T} will be equipped with the duality $D_L := \hom(-, L)$ as in Example 2.2.6 (ii).

Remark 2.3.3. We will see later that the most important examples for us will be, on the one hand, the opposite of the Krull-codimension $-\operatorname{codim}_{\mathrm{Krull}}$, equipping Example 2.2.6 (iii), i.e. $D^{\mathrm{perf}}(X)$ for X quasi-compact and quasi-separated, with a dimension function, see [Bal07, Example 3.3]

On the other hand, the Krull dimension \dim_{Krull} will be our choice of a dimension function for kG-stab from Example 2.2.6 (iv), see Definition 5.1.24

For further details, the reader is referred to [Bal07, Examples 3.2 and 3.3].

Definition 2.3.4. For a tensor triangulated category $\mathcal{T} = (\mathcal{T}, \otimes, \mathbb{1})$ with a dimension function dim, we define the *dimension* of a subset $V \subset \operatorname{Spc}(\mathcal{T})$ as $\dim(V) := \sup\{\dim(\mathcal{P}) \mid \mathcal{P} \in V\}$. For any $n \in \mathbb{Z} \cup \{\pm\infty\}$, we set

$$\mathcal{T}_{(n)} := \{ A \in \mathcal{T} \mid \dim(\operatorname{supp}(A)) \le n \}$$

yielding a thick tensor ideal in \mathcal{T} by the properties of support ([Bal07, Proposition 1.3]) and dimension function (Definition 2.3.1).

We often call $\mathcal{T}_{(n)}$ the *n*-th filtration component of the tensor triangulated category \mathcal{T} . Moreover, we define $\operatorname{Spc}(\mathcal{T})_n$ as the set of points \mathcal{Q} in $\operatorname{Spc}(\mathcal{T})$ with $\dim(\mathcal{Q}) = n$.

Lemma 2.3.5. Let $(\mathcal{T}, \otimes, \mathbb{1}, \text{hom}, \dim)$ be a rigid tensor triangulated category with dimension function that we consider a triangulated category with the duality $D_L := \text{hom}(-, L)$ for a fixed \otimes -invertible object L in \mathcal{T} following Example 2.2.6 (i)/(ii). Let $n \in \mathbb{Z} \cup \{\pm \infty\}$.

Then, the thick tensor ideal $\mathcal{T}_{(n)}$ is a triangulated category with duality when equipped with the duality induced by D_L .

Proof. Since $\mathcal{T}_{(n)}$ is a tensor prime ideal, it is in particular a full triangulated subcategory. It follows from [Bal07, Proposition 2.7] that for any object a in \mathcal{T} , we have $\operatorname{supp}(a) \cong \operatorname{supp}(\operatorname{hom}(a, 1))$, and thus especially $\dim(\operatorname{supp}(a)) =$ $\dim(\operatorname{supp}(\operatorname{hom}(a, 1)))$, which means that if a is in $\mathcal{T}_{(n)}$, then so is $\operatorname{hom}(a, 1)$. More generally, by [Bal07, Proposition 2.7] and Lemma 2.2.5, we have $\operatorname{supp}(\operatorname{hom}(a, L)) =$ $\operatorname{supp}(a) \cap \operatorname{supp}(L)$ and $\operatorname{supp}(L) = \operatorname{Spc}(\mathcal{T})$ as subspaces of $\operatorname{Spc}(\mathcal{T})$, yielding that for a in $\mathcal{T}_{(n)}$ and $L \otimes$ -invertible in \mathcal{T} ,

$$\dim(\operatorname{supp}(\operatorname{hom}(a, L))) = \dim(\operatorname{supp}(a) \cap \operatorname{supp}(L))$$
$$= \dim(\operatorname{supp}(a) \cap \operatorname{Spc}(\mathcal{T})) = \dim(\operatorname{supp}(a)).$$

That is, if a is in $\mathcal{T}_{(n)}$, so is hom(a, L). The duality functor hom(-, L) and the double dual identification from Example 2.2.6 therefore restrict to $\mathcal{T}_{(n)}$, equipping it with a duality structure.

Remark 2.3.6. The above lemma states that for a rigid tensor triangulated category $(\mathcal{T}, \otimes, \mathbb{1}, \text{hom})$, we obtain triangulated categories with duality $\mathcal{T}_{(n)}$ equipped with the restricted duality of $D_L := \text{hom}(-, L)$, allowing us to define Witt groups and Grothendieck-Witt groups of the filtration components $\mathcal{T}_{(n)}$ later.

Apart from the filtration by dimension of support, we will also need the technique of idempotent completion of a (tensor) triangulated category later since the quotients of the filtration components are not necessarily idempotent complete even if the category \mathcal{T} is in the first place, see [Bal07, Remark 1.14].

We know from [BS01] and [Bal07, Proposition 2.15] that applying the usual idempotent completion $(-)^{\natural}$ of an additive category to a (tensor) triangulated category naturally yields a (tensor) triangulated category again. In our setting, we need the idempotent completion to moreover behave well with dualities.

Lemma 2.3.7. The idempotent completion \mathcal{T}^{\natural} of a triangulated category with duality $\mathcal{T} = (\mathcal{T}, \otimes, \mathbb{1}, *, \eta)$ is again a triangulated category with duality.

Proof. Recall that $(\mathcal{T})^{\natural}$ is naturally already a triangulated category by [BS01]. It inherits a duality structure

$$\overset{\sim}{*}: ((\mathcal{T})^{\natural})^{\mathrm{op}} \to (\mathcal{T})^{\natural}$$

given by

$$(A, a) \mapsto (A^*, a^*), f : [(A, a) \to (B, b)] \mapsto [f^{\widetilde{*}} : (B^*, b^*) \to (A^*, a^*)],$$

where $f^{\tilde{*}} = f^*$. Note that the dual of an idempotent morphism is still idempotent since we have $a^* \circ a^* = (a \circ a)^* = a^*$. Also, $f^{\tilde{*}}$ actually is a morphism in the idempotent completion since we have $b \circ f = f = f \circ a$ by definition of the idempotence of f and thus obtain the desired idempotence of $f^{\tilde{*}}$ by applying the duality functor *.

Moreover, η induces a natural transformation

$$\widetilde{\omega}: \mathbbm{1}_{\mathcal{T}^{\natural}} \to \widetilde{\ast} \circ \widetilde{\ast}$$

defined elementwise as $\widetilde{\omega}_{(A,a)} := \eta_A \times \eta_a$. Hence, \mathcal{T}^{\ddagger} is a triangulated category with duality.

Moreover, we need idempotent completion to behave well with the structure of a rigid tensor triangulated category as well as with Verdier localization.

Lemma 2.3.8. ([Bal07, Proposition 2.15])

(i) The idempotent completion of a rigid tensor triangulated category is again a rigid tensor triangulated category. The idempotent completion functor is a duality-preserving functor when regarding the rigid tensor triangulated category a triangulated category with duality (see Example 2.2.6 (i) and (ii)). (ii) The Verdier localization of a rigid tensor triangulated category at a thick tensor ideal is again a rigid tensor triangulated category. The localization functor is a duality-preserving functor when regarding the rigid tensor triangulated category a triangulated category with duality (see Example 2.2.6 (i) and (ii)).

Recall from Definition 2.3.4 that, for a rigid tensor triangulated category with a dimension function $(\mathcal{T}, \otimes, 1, \text{hom}, \text{dim})$, we defined filtration components

$$\mathcal{T}_{(n)} := \{ A \in \mathcal{T} \mid \dim(\operatorname{supp}(A)) \le n \}$$

for $n \in \mathbb{Z} \cup \{\pm \infty\}$, which yield a filtration by thick tensor ideals

$$\mathcal{T}_{(-\infty)} \subset ... \subset \mathcal{T}_{(n-1)} \subset \mathcal{T}_{(n)} \subset ... \subset \mathcal{T}_{(\infty)} = \mathcal{T}_{(\infty)}$$

It lifts to a filtration of \mathcal{T} when considered a triangulated category with duality with each component being a triangulated category with duality again (cf. Lemma 2.3.5):

$$\mathcal{T}_{(-\infty)} \subset \ldots \subset (\mathcal{T}_{(n-1)}, D_L|_{\mathcal{T}_{(n-1)}}, \eta) \subset (\mathcal{T}_{(n)}, D_L|_{\mathcal{T}_{(n)}}, \eta) \subset \ldots \subset \mathcal{T}_{(\infty)}.$$

The inclusion functors are duality-preserving functors since all the dualities are induced by hom(-, L).

For a tensor prime ideal $\mathcal{P} \in \operatorname{Spc}(\mathcal{T})$, let $\mathcal{T}_{\mathcal{P}} := (\mathcal{T}/\mathcal{P})^{\natural}$ and let $\operatorname{FL}(\mathcal{T}_{\mathcal{P}}) := \{a \in \mathcal{T}_{\mathcal{P}} | \operatorname{supp}(a) \subset *\}$, where * is the unique closed point of $\operatorname{Spc}(\mathcal{T}_{\mathcal{P}})$. See [Bal07, Definition 3.7] for more details.

For future reference, we recall the following important decomposition statement.

Theorem 2.3.9. ([Bal07, Theorem 3.24]) Let $(\mathcal{T}, \otimes, 1, \text{hom}, \dim)$ be an essentially small, rigid tensor triangulated category equipped with a dimension function such that $\text{Spc}(\mathcal{T})$ is noetherian. Then, for all $n \in \mathbb{Z}$, we have a triangulated equivalence

$$(\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)})^{\natural} \to \bigoplus_{\substack{\mathcal{P} \in \operatorname{Spc}(\mathcal{T}) \\ \dim(\mathcal{P})=n}} \operatorname{FL}(\mathcal{T}_{\mathcal{P}}).$$

Corollary 2.3.10. The above equivalence passes to an equivalence of triangulated categories with duality when considering the rigid tensor triangulated category $(\mathcal{T}, \otimes, \mathbb{1}, \hom)$ a triangulated category with duality $(\mathcal{T}, D_L, \eta_L)$ as in Example 2.2.6 (i)/(ii) for L a \otimes -invertible object of \mathcal{T} .

Proof. It follows from Example 2.2.6 (ii) that $(\mathcal{T}, D_L, \eta_L)$ is indeed a triangulated category with duality as well as $(\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)})^{\natural}$ by the above considerations of this section, see e.g. Lemma 2.3.8. The duality \overline{D}_L on the triangulated category on the right-hand side is induced by the localization functors $q_{\mathcal{P}}: \mathcal{T} \to \mathcal{T}/\mathcal{P}$ for each

 $\mathcal{P} \in \operatorname{Spc}(\mathcal{T})$. Applying [Bal07, Proposition 2.15], we see that for all objects a we have

$$\overline{D}_L(a) = \hom(q_{\mathcal{P}}(a), q_{\mathcal{P}}(L)) \cong q_{\mathcal{P}}(\hom(a, L)) = q_P(D_L(a)).$$

The triangulated equivalence from the statement is induced by the functor

$$\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)} \to \bigoplus_{\substack{\mathcal{P}\in \operatorname{Spc}(\mathcal{T})\\\dim(\mathcal{P})=n}} \operatorname{Min}(\mathcal{T}/\mathcal{P}), \quad a \mapsto q_{\mathcal{P}}(a).$$

We can thus deduce from Lemma 2.3.8 that the equivalence is a duality-preserving functor. $\hfill \Box$

In this subsection, the central idea of tensor triangular geometry to filtrate by dimension of support was introduced. We validated that this construction, as well as important tools such as idempotent completion or Verdier localization, behave well with dualities.

The whole section intended to introduce the reader to tensor triangular geometry, starting with the basic definition of its objects of study, the functors between them, and main concepts such as tensor prime ideals and the Balmer spectrum $\operatorname{Spc}(\mathcal{T})$ of a tensor triangulated category. After that, rigid tensor triangulated categories were introduced and portrayed as our central way of introducing a duality structure on tensor triangulated categories inherited by the internal Hom. Examples have been given and shifted dualities have been defined. Eventually, dimension functions embodied the last ingredient to filtrate a given tensor triangulated category by dimension of support. This construction and moreover idempotent completion and Verdier localization have been shown to be compatible with our concept of duality.

The following section will start from a different point of view and introduce Grothendieck–Witt and Witt groups as a generalization of classical algebraic K-theory, which will be a main tool to generalize the construction of tensor triangular Chow groups to tensor triangular Chow–Witt groups in Section 4.2.

Chapter 3

Grothendieck–Witt and Witt Groups

The generalization from Chow groups to Chow–Witt groups in the classical algebrogeometric case involves, in particular, the passage from Milnor K-theory to Milnor– Witt K-theory. In degree 0, this nails down to the passage from the Grothendieck group K_0 to the Grothendieck–Witt group GW.

We assume that the reader is familiar with the theory of Grothendieck–Witt and Witt groups of fields and rings. As an introduction, we recommend [Lam05, MH73, Scha85].

In this chapter, we introduce the notions of Grothendieck–Witt groups and Witt groups of triangulated categories with duality as well as of complicial exact categories with weak equivalences and duality. We introduce basic notions in these settings and compare the definitions. In the last section, we examine the different localization theorems that will later serve as a tool to show that the definition of Chow–Witt groups of tensor triangulated categories indeed generalizes the classical one of schemes.

In the triangulated case, we follow the references [Bal00, Wal03, Sch17], whereas for the exact case [Sch10a, Sch10b, FS09] are our main references.

Convention 3.0.1. As stated above, all categories will be considered essentially small if not stated otherwise. In this chapter, we assume that any triangulated category \mathcal{T} or exact category \mathcal{E} contains $\frac{1}{2}$, i.e., the abelian group Hom(A, B) is uniquely divisible by 2 for all objects A and B in \mathcal{T} (or \mathcal{E} , respectively). We also say that \mathcal{T} is $\mathbb{Z}[\frac{1}{2}]$ -linear. For the example of schemes we will consider later, all base fields k will have a characteristic $\neq 2$ (compare Remark 3.1.4).

3.1 (Grothendieck–)Witt Groups of Triangulated Categories

We start with the theory of (Grothendieck–)Witt groups of triangulated categories with duality. The constructions of this chapter do not rely on any tensor structure.

A rigid tensor triangulated category may be turned into a triangulated category with duality as in Example 2.2.6 (i)/(ii), to which one can apply GW or W. We will proceed like this in Section 4.2. Now, we begin with some conventions before introducing the notions necessary to define (Grothendieck–)Witt groups of triangulated categories with duality.

Witt groups and Grothendieck–Witt groups of fields and rings intend to capture the behavior of symmetric bilinear forms on vector spaces and modules, respectively, which can be described briefly as symmetric isomorphisms to their dual. To generalize this definition for triangulated categories, one needs to define symmetric objects.

Definition 3.1.1. For objects A, B of a triangulated category with duality $\mathcal{T} = (\mathcal{T}, *, \eta)$, the transpose of a morphism $u : A \to B^*$ is defined as $u^t := u^* \circ \eta_B$. A morphism $w : A \to A^*$ is symmetric if $w^t = w$. A symmetric object of \mathcal{T} is a pair (A, w), where $w : A \xrightarrow{\cong} A^*$ is a symmetric isomorphism. Two symmetric objects (A, w) and (B, v) are isometric if there is an isomorphism $r : A \xrightarrow{\cong} B$ such that $w = r^* vr$.

We denote by SymOb(\mathcal{T}) the free abelian group of isometry classes of symmetric objects of \mathcal{T} with the orthogonal sum as group operation, where the orthogonal sum $(A, \alpha) \oplus (B, \beta)$ of symmetric spaces (A, α) and (B, β) is defined as

$$\alpha \oplus \beta := \begin{bmatrix} \alpha & 0 \\ 0 & \beta \end{bmatrix} : A \oplus B \to A^* \oplus B^*.$$

Remark 3.1.2. The symmetric objects together with their isomorphisms form a groupoid. Note that a duality-preserving functor $(F, \mu) : (\mathcal{A}, *) \to (\mathcal{B}, \tilde{*})$ between triangulated categories with duality induces a morphism of groupoids sending $w : A \xrightarrow{\cong} A^*$ to $\mu_A \circ Fw : FA \xrightarrow{\cong} F(A^*) \cong F(A)^{\tilde{*}}$.

Definition-Lemma 3.1.3. [[Bal00], Theorem 2.6] Let $(\mathcal{T}, *, \eta)$ be a triangulated category with duality.

(i) For any morphism $u: T^{-1}(A) \to A^*$ that is symmetric with respect to the (-1)-st shifted duality, there exists an exact triangle of the form

$$T^{-1}(A) \xrightarrow{u} A^* \xrightarrow{\phi^{-1}v^*} B \xrightarrow{v} A.$$

Here, $\phi : B \xrightarrow{\cong} B^*$ is an isomorphism with respect to the 0-th shifted duality and (B, ϕ) is uniquely determined by (A, u) up to isomorphism. We define $\operatorname{cone}(A, u) := (B, \phi).$

- (ii) We call a symmetric object metabolic if it is isomorphic to some cone(A, u). In the above setting, A^* is called a Lagrangian of (B, ϕ) .
- (iii) A hyperbolic object is a symmetric object $\mathbb{H}(A)$ for some object A of the form

$$\mathbb{H}(A) := (A \oplus A^*, \begin{bmatrix} 0 & 1_{A^*} \\ \eta_A & 0 \end{bmatrix}) = \operatorname{cone}(A, 0).$$

Remark 3.1.4. Note that we usually assume that $\operatorname{char}(k) \neq 2$ (or $\frac{1}{2}$, i.e. 2, being invertible). If 2 is invertible, *quadratic* and symmetric forms coincide as well as *split-metabolic* and hyperbolic objects (see [Bal05a, Remark 36]) for the category of finitely generated projective *R*-modules.

However, if 2 is not invertible, this is not always the case, see [Bal05a, Example 38] for an example of a split-metabolic object that is not hyperbolic.

There has been progress for when 2 is not invertible. For example, in [Sch10a] and [Sch10b], assumptions on the characteristic of the base field when considering (Grothendieck–)Witt groups of exact categories are dropped already.

Definition 3.1.5. Let $\mathcal{T} = (\mathcal{T}, *, \eta)$ be a triangulated category with duality.

- (i) The Grothendieck-Witt group $GW(\mathcal{T}) = GW(\mathcal{T}, *, \eta)$ is the quotient of $SymOb(\mathcal{T})$ divided by the following relations:
 - (a) $[(A, \alpha) \perp (B, \beta)] = [A, \alpha] + [B, \beta]$ and
 - (b) $[\operatorname{cone}(Y, f)] = [\mathbb{H}(Y)].$
- (ii) The Witt group $W(\mathcal{T}, *, \eta)$ is the quotient of $GW(\mathcal{T})$ by the subgroup \mathbb{H} generated by the hyperbolic classes, that is $W(\mathcal{T}) := GW(\mathcal{T})/\mathbb{H}$.
- (iii) For $n \in \mathbb{Z}$, the *n*-th shifted Grothendieck-Witt group $\mathrm{GW}^{[n]}(\mathcal{T})$ (and Witt group $\mathrm{W}^{[n]}(\mathcal{T})$) is the usual Grothendieck-Witt group (and Witt group, respectively) equipped with the *n*-th shifted duality $T^n \circ *$ from Definition 2.2.7.
- (iv) Let $* = D_L$ for a \otimes -invertible object in \mathcal{T} , for example in the case of Example 2.2.6 (ii). In this special case, we denote by $\mathrm{GW}(\mathcal{T}, L) := \mathrm{GW}(\mathcal{T}, D_L, \eta_L)$ and $\mathrm{W}(\mathcal{T}, L) := \mathrm{W}(\mathcal{T}, D_L, \eta_L)$ the Grothendieck–Witt group and Witt group twisted by L, respectively.
- Remark 3.1.6. (i) Note that symmetric objects from [Wal03] coincide with symmetric spaces from [Bal00]. In [Wal03, p.6], an equivalent definition of $W(\mathcal{T})$ is given via relations that immediately shows the equivalence of Definition 3.1.5 below to Balmer's definition of triangular Witt groups in [Bal00, Definition 2.13]. In total, (Grothendieck–)Witt groups of triangulated categories of Walter ([Wal03]), Schlichting ([Sch17]), and Balmer ([Bal00]) coincide by [Sch17, Remark 3.14].
 - (ii) Readers familiar with the notion of *higher* Grothendieck–Witt groups, e.g. of exact categories as we will use later, must not confuse these with *shifted* Grothendieck–Witt groups, in this case of triangulated categories, for there are no higher Grothendieck–Witt groups of triangulated categories. More details on this comparison can be found in [Sch10a, Remark 4.14].

Lemma 3.1.7. Given a duality-preserving functor between triangulated categories with duality $(F, \varphi) : (\mathcal{T}, *_{\mathcal{T}}) \to (\mathcal{S}, *_{\mathcal{S}})$ as in Definition 2.1.4, i.e. $\varphi : F \circ *_{\mathcal{T}} \simeq$ $*_{\mathcal{S}} \circ F$ is a triangulated equivalence, we obtain induced group homomorphisms for $n \in \mathbb{Z}$

$$\overline{F}: \mathrm{GW}^{[n]}(\mathcal{T}, *_{\mathcal{T}}) \to \mathrm{GW}(\mathcal{S}, *_{\mathcal{S}})$$

and

 $\overline{F}: W^{[n]}(\mathcal{T}, *_{\mathcal{T}}) \to W(\mathcal{S}, *_{\mathcal{S}}).$

In particular, if $*_{\mathcal{T}} = D_L$ and $*_{\mathcal{S}} = D_{L'}$ for fixed \otimes -invertible objects L and L'of \mathcal{T} and \mathcal{S} , respectively, we obtain group homomorphisms $\overline{F} : \mathrm{GW}^{[n]}(\mathcal{T}, L) \to \mathrm{GW}(\mathcal{S}, L')$ and $\overline{F} : \mathrm{W}^{[n]}(\mathcal{T}, L) \to \mathrm{W}(\mathcal{S}, L')$.

If F is moreover a triangulated equivalence, \overline{F} is an isomorphism.

Proof. For symmetric objects (A, a) of $GW(\mathcal{T}, *_{\mathcal{T}})$, we define

$$\overline{F}(A,a) = (FA,\varphi_a \circ Fa).$$

The mapping preserves metabolic objects and Lagrangians. Hence, it induces a morphism on GW. In particular, it preserves hyperbolic classes (cf. [Sch10a, Section 3.1]) and, thus, induces morphisms on Witt groups. The "moreover" statement follows immediately from the concrete description of \overline{F} . The case of nonzero *n* follows analogously.

Having seen the definition of (Grothendieck–)Witt groups of triangulated categories with duality, we will introduce the same notion for certain types of exact categories in the next section.

3.2 (Grothendieck–)Witt Groups of Exact Categories

We are now familiar with the definition of (Grothendieck–)Witt groups of triangulated categories with duality. One can also define Grothendieck–Witt and Witt groups of exact categories with duality as in [Sch10a]. The two constructions agree in the sense of the forthcoming Proposition 3.2.14. Note that Schlichting uses the word *symmetric space* for what we call a symmetric object here.

In Section 4.2, we will need both notions to prove that our tensor triangular Chow–Witt groups indeed generalize the classical Chow–Witt groups.

For the convenience of the reader, we briefly recall the definition of exact categories with duality.

Definition 3.2.1. We call a triple $(\mathcal{E}, *, \eta)$ an *exact category with duality* if \mathcal{E} is an exact category in the sense of Quillen, $* : \mathcal{E} \to \mathcal{E}^{\text{op}}$ is an exact functor, and $\eta_A : A \xrightarrow{\cong} A^{**}$ is a natural isomorphism fulfilling $\mathrm{id}_{A^*} = \eta_A^* \circ \eta_{A^*}$ for all objects A in \mathcal{E} , called the double dual identification.

As in the case of triangulated categories, we now want to introduce symmetric objects on exact categories with duality.

Definition 3.2.2. Let $(\mathcal{E}, *, \eta)$ be an exact category with duality. A symmetric object is a pair (M, ϕ) , where M is an object of \mathcal{E} and $\phi : M \xrightarrow{\cong} M^*$ a symmetric isomorphism, i.e. an isomorphism satisfying $\phi^* \eta = \phi$. An isometry between two symmetric objects (M, ϕ) and (N, ψ) is an isomorphism $f : M \xrightarrow{\cong} N$ such that $\phi = f^* \psi f$.

We denote the Grothendieck group of the abelian monoid of isometry classes of symmetric objects of \mathcal{E} with the orthogonal sum as group operation by SymOb(\mathcal{E}).

Remark 3.2.3. In Definition 3.1.1 and Definition 3.2.2, we defined symmetric objects as objects A equipped with a symmetric isomorphism to its dual $\varphi : A \xrightarrow{\cong} A^*$. In the example of vector spaces over a field K, there is moreover the notion of a symmetric bilinear form. By definition, this is a map

$$B:V\times V\to K$$

satisfying the following three conditions:

- (i) B(x, y) = B(y, x),
- (ii) B(x+y,z) = B(x,z) + B(y,z),
- (iii) $B(\lambda x, y) = \lambda B(x, y)$ for all $x, y, z \in V$ and $\lambda \in K$.

Given a symmetric object (V, φ) , we obtain a symmetric bilinear form by setting $B(x, y) := \varphi(x)(y)$. Conversely, a symmetric bilinear form B yields a symmetric object (V, φ) with $\varphi(x) := B(x, -)$. Hence, we can switch between symmetric isomorphisms to its dual and symmetric bilinear forms. For more details, see [Scha85, Chapter 1, §2].

Definition 3.2.4. Let $(\mathcal{E}, *, \eta)$ be an exact category with duality.

- (i) A totally isotropic subobject of a symmetric object (M, φ) is an admissible monomorphism $i : L \to M$ such that $0 = i^* \varphi i$, and such that $L \to \ker(i^*\varphi) \subset M$ is also an admissible monomorphism.
- (ii) A Lagrangian of a symmetric object (M, φ) is a totally isotropic subobject $L \xrightarrow{i} M$ such that

$$L \xrightarrow{i} M \xrightarrow{i^* \varphi} L^*$$

is an exact sequence in \mathcal{E} .

- (iii) A symmetric object (M, φ) is called *metabolic* if it has a Lagrangian.
- (iv) A hyperbolic object is a symmetric object $\mathbb{H}(A)$ for some object A of the form

$$\mathbb{H}(A) := (A \oplus A^*, \begin{bmatrix} 0 & 1_{A^*} \\ \eta_A & 0 \end{bmatrix})$$

Definition 3.2.5. Let $(\mathcal{E}, *)$ be an exact category with duality. We define the *Grothendieck–Witt group* $\mathrm{GW}_0(\mathcal{E}, *) := \mathrm{GW}(\mathcal{E}, *)$ as the quotient of $\mathrm{SymOb}(\mathcal{E})$ divided by the relation $[(M, \varphi)] = [\mathbb{H}(L)]$ for metabolic objects $[(M, \varphi)]$ with Lagrangian L.

The Witt group $W_0(\mathcal{E}, *) := W(\mathcal{E}, *)$ is the quotient of $GW_0(\mathcal{E}, *)$ modulo the relation $[\mathbb{H}(A)] = 0$ for any hyperbolic object $\mathbb{H}(A)$.

Example 3.2.6. Given a ring R with involution and a scheme X together with a line bundle L, the category of finitely generated projective right R-modules P(R) and the category of algebraic vector bundles Vect(X) are exact categories with duality via the dualities $Hom_R(-, R)$ and $Hom_{\mathcal{O}_X}(-, L)$, respectively, compare [Sch10a, Examples 2.2 and 2.3].

We now verify the compatibility of the definition of Grothendieck–Witt groups of exact categories with duality and triangulated categories with duality. The argument in particular implies that the relation $[(M, \varphi)] = [\mathbb{H}(L)]$ is always fulfilled in the split-exact category P(k) of finite-dimensional k-vector spaces as stated in [QSS79, Theorem 5.6].

Proposition 3.2.7. [[Wal03, Theorem 6.1], [Bal01, Theorem 4.3]] Let k be a field of characteristic $\neq 2$. Then, we have a group isomorphism of Grothendieck–Witt groups

$$\operatorname{GW}(k) := \operatorname{GW}(\operatorname{P}(k), \operatorname{Hom}_k(-, k)) \cong \operatorname{GW}(\operatorname{D}^{\operatorname{perf}}(\operatorname{Spec}(k)), \operatorname{hom}(-, 1))$$

as well as a group isomorphism

$$W(k) := W(P(k), Hom_k(-, k)) \cong W(D^{perf}(Spec(k)), hom(-, 1))$$

Remark 3.2.8. In the above proposition, the isomorphisms generalizes to isomorphisms

$$\operatorname{GW}(\operatorname{P}(k), \operatorname{Hom}_k(-, L)) \cong \operatorname{GW}(\operatorname{D}^{\operatorname{perf}}(\operatorname{Spec}(k)), \operatorname{hom}(-, L))$$

(and for Witt groups respectively) for any line bundle L on Spec(k) corresponding to an invertible k-vector space L, i.e. a vector space of dimension 1 over k. Note that $GW(k, L) \cong GW(k)$ and $W(k, L) \cong W(k)$ for any invertible k-vector space L, but this isomorphism is non-canonical. In the theory of Chow–Witt groups, twists of this sort can be interpreted as local orientations, see [Deg23, Remark 2.1.15].

We will need additional structure in the form of *weak equivalences*, similar to the idea of Waldhausen categories in algebraic K-theory, in order to formulate localization sequences.

Definition 3.2.9. (i) An exact category with weak equivalences (\mathcal{E}, w) consists of an exact category \mathcal{E} in the sense of [Qui73] (compare [Kel96, Chapter 4]) and a set of morphisms w that contains all identity morphisms, is closed under isomorphisms, composition, retracts, pushouts along admissible monomorphisms, pullbacks along admissible epimorphisms, and satisfies the 2-out-of-3 property. We call morphisms in w weak equivalences.

- (ii) An exact category with weak equivalences and duality is a collection of data $(\mathcal{E}, w, *, \eta)$ such that (\mathcal{E}, w) is an exact category with weak equivalences and $(\mathcal{E}, *, \eta)$ is an exact category with (not necessarily strong) duality such that
 - (i) $*: (\mathcal{E}^{op}, w) \to (\mathcal{E}, w)$ is exact, hence, in particular, $*(w) \subset w$, and
 - (ii) $\eta_X \in w$ for all objects X.

Remark 3.2.10. If $(\mathcal{E}, w, *, \eta)$ is an exact category with weak equivalences and duality, the subcategory $w\mathcal{E}$ of weak equivalences is an exact category with duality.

The example of chain complexes of coherent locally free sheaves on a scheme X will play an important role in showing the agreement of algebro-geometric Chow–Witt groups and tensor triangular Chow–Witt groups. It can be considered an exact category with weak equivalences and duality in the following way.

Example 3.2.11. Let k be a field, X be a k-scheme, and L a line bundle on X. By [Sch11, Example 3.2.4] the exact category with duality of coherent locally free sheaves on a scheme X

$$(P(X), \operatorname{Hom}_{\mathcal{O}_X}(-, L))$$

yields an exact category with weak equivalences and duality (see [Sch10b, Section 6.1])

$$(Ch^{b}(P(X)), quis, Hom_{\mathcal{O}_{X}}(-, L))$$

by fixing the quasi-isomorphisms w := quis, which is moreover *complicial* in the sense of [Sch11, Definition 3.2.2]. We define for $n \in \mathbb{N}$ the subcategory of complexes with homology supported in codimension $\geq n$

$$\operatorname{Ch}^{b}(P(X))^{(n)} := \{P_{\bullet} \in \operatorname{Ch}^{b}(P(X)) \mid \operatorname{codim}(\operatorname{supph}(P_{\bullet})) \ge n\}$$

(see Definition 4.2.5 for the definition of the homological support). By quis^{n'}, we denote the morphisms in $\operatorname{Ch}^{b}(P(X))^{(n)}$ with cone in $\operatorname{Ch}^{b}(P(X))^{(n')}$ for $n' \geq n$. Then, for any $n' \geq n$,

$$(\operatorname{Ch}^{b}(P(X))^{(n)}, \operatorname{quis}^{n'}, \operatorname{Hom}_{\mathcal{O}_{X}}(-, L))$$

is a $(\mathbb{Z}[\frac{1}{2}])$ -complicial exact category with weak equivalences and duality, compare [FS09, Definition 6]. We denote the triangulated category where we invert the quasi-isomorphisms quis^{n'} by $\operatorname{Ch}^{b}(P(X))^{(n)}[(\operatorname{quis}^{n'})^{-1}]$.

Remark 3.2.12. In a preprint version of [Sch17], Schlichting used the language of complicial exact categories with weak equivalences and duality as we do here. The published version is written in the setting of dg-categories, but Schlichting explains in the introduction of *loc. cit.* (p.1733) that, in practice, all complicial exact categories have a *dg-enhancement*. They might have different exact structures, but this does not affect the GW-groups. For this reason, our references to [Sch17] remain valid despite the fact that we work in the complicial setting. For further details, the reader is referred to [Sch10b, Sch17].

Analogously to the triangulated case, one can define for $n \in \mathbb{Z}$ the shifted Grothendieck–Witt groups $\mathrm{GW}_{0}^{[n]}(\mathcal{E}, w, *, \eta) := \mathrm{GW}^{[n]}(\mathcal{E}, w, *, \eta) =: \mathrm{GW}^{[n]}(\mathcal{E})$ and Witt groups $\mathrm{W}_{0}^{[n]}(\mathcal{E}, w, *, \eta) := \mathrm{W}^{[n]}(\mathcal{E}, w, *, \eta) =: \mathrm{W}^{[n]}(\mathcal{E})$ of a complicial exact category with weak equivalences and duality. We will not give the explicit definition but refer the reader to [Sch17].

Now, we discuss the compatibility between the concept of GW of exact categories with duality and exact categories with weak equivalences and duality, yielding an intuitive result well-known for more theories such as algebraic K-theory.

The result is stated in a more general way. Expanding the definitions we just gave, one can define *higher (shifted) Grothendieck–Witt groups* $\mathrm{GW}_m^{[n]}(\mathcal{E}), m \geq 1$, for complicial exact categories with duality (see [Sch10a, Section 4]) as well as for exact categories with weak equivalences and duality (see [Sch10b, Section 2.7]).

Proposition 3.2.13. ([Sch10b, Proposition 6]) Let $(\mathcal{E}, *)$ be an exact category with duality, $m \in \mathbb{N}$. There are isomorphisms of (higher) Grothendieck–Witt groups induced by the inclusion as complexes concentrated in degree 0

$$\operatorname{GW}_m(\mathcal{E},*) \xrightarrow{\cong} \operatorname{GW}_m^{[0]}(\operatorname{Ch}^b(\mathcal{E}),\operatorname{quis},*).$$

Moreover, we have the following agreement between Grothendieck–Witt groups of exact categories with weak equivalences and duality, and of triangulated categories.

Proposition 3.2.14. ([Sch17, Proposition 3.8]) For an exact category with duality $(\mathcal{E}, *), n, n' \in \mathbb{N}, n' \geq n, i \in \mathbb{Z}$, we have isomorphisms of (shifted) Grothendieck–Witt groups

$$\operatorname{GW}_{0}^{[i]}(\operatorname{Ch}^{b}(\mathcal{E})^{(n)}, \operatorname{quis}^{n'}, *) \cong \operatorname{GW}_{0}^{[i]}(\operatorname{Ch}^{b}(\mathcal{E})^{(n)}[(\operatorname{quis}^{n'})^{-1}]).$$

Remark 3.2.15. Let X be a scheme and L a line bundle on X. We follow the idea of Fasel and Srinivas ([FS09]) in Theorem 3.3.4. For $\operatorname{Ch}^{b}(P(X))^{(n)}[(\operatorname{quis}^{n'})^{-1}]$, they use the notation $\operatorname{D}^{b}(P(X))^{n/n'}$. Moreover, they denote by $\operatorname{GW}_{m}(\operatorname{D}^{b}(P(X))^{n/n'})$ the *m*-th higher Grothendieck–Witt group of the complicial exact category with duality $(\operatorname{Ch}^{b}(P(X))^{(n)}, \operatorname{quis}^{n'}, \operatorname{Hom}_{\mathcal{O}_{X}}(-, L))$. In our case, we consider this notation misleading since it suggests the existence of higher Grothendieck–Witt groups of triangulated categories with duality. However, so far we do not have such a definition, hence the change of notation.

Note that, as in the case of rings, negative Grothendieck–Witt groups ([Sch17]) of complicial exact categories with weak equivalences and duality can be defined and recover Witt groups in the following sense.

Proposition 3.2.16. ([Sch17, Proposition 6.3]) Let $\mathcal{A} = (\mathcal{A}, w, *, \eta)$ be a complicial exact category with weak equivalences and duality, m < 0, and $n \in \mathbb{Z}$. Then, there are isomorphisms

$$\operatorname{GW}_{m}^{[n]}(\mathcal{A}, w) \cong \operatorname{GW}_{m-1}^{[n-1]}(\mathcal{A}, w)$$

and

$$W^{[n-m]}(\mathcal{A}, w) \cong GW_m^{[n]}(\mathcal{A}, w).$$

Together with [Bal01, Theorem 4.3] this yields the following comparison.

Corollary 3.2.17. Let $(\mathcal{E}, *)$ be an exact category with duality, $n, n' \in \mathbb{N}$, $n' \ge n$, m < 0, and $i \in \mathbb{Z}$

$$\operatorname{GW}_{m}^{[i]}(\operatorname{Ch}^{b}(\mathcal{E})^{(n)},\operatorname{quis}^{n'},*) \cong \operatorname{W}^{[i-m]}(\operatorname{Ch}^{b}(\mathcal{E})^{(n)},\operatorname{quis}^{n'},*)$$
$$\cong \operatorname{W}^{[i-m]}(\operatorname{Ch}^{b}(\mathcal{E})^{(n)}[(\operatorname{quis}^{n'})^{-1}]).$$

In a nutshell, this section introduced the notion of (Grothendieck–)Witt groups of exact categories with duality as well as of complicial exact categories with weak equivalences and duality. For an exact category with duality (\mathcal{E} , *), we have seen that its (Grothendieck–)Witt groups coincide with the ones for the category with weak equivalences and duality ($\mathrm{Ch}^{b}(\mathcal{E})$, quis, *, η) as well as with the (Grothendieck–)Witt groups of triangulated categories with duality after inverting the quasi-isomorphisms.

These concepts and their connection will be essential in the proof of the central agreement theorem of tensor triangular Chow–Witt groups and classical Chow–Witt groups in Section 4.2. The main ingredients will be the different forms of localization sequences arising from the different GW/W-theories we have seen so far in this chapter.

3.3 Localization

As already mentioned, one central idea of the agreement between triangular and algebro-geometric Chow–Witt groups lies in the agreement of the different localization sequences. To lay the foundations for this, we discuss localization sequences for GW and W in the different settings of triangulated categories and exact categories.

We begin with the triangulated setting in which a lot of work on Witt groups has been done by Balmer, e.g. in [Bal00, Bal01]. Although Walter did not publish the article [Wal03] in which he stated the localization theorem for Grothendieck– Witt and Witt groups of triangulated categories, we still rely on this source since Grothendieck–Witt groups play a more central role here.

Let us first recall the definition of *exact sequences* in the triangulated setting.

Definition 3.3.1. We say that a sequence $\mathcal{J} \to \mathcal{T} \to \mathcal{L}$ of triangulated categories with duality and duality-preserving functors is a *localization* or *exact sequence* if $\mathcal{L} = S^{-1}\mathcal{T}$ for some class S of morphisms and \mathcal{J} is the full subcategory of \mathcal{T} spanned by those objects that become isomorphic to 0 in \mathcal{L} .

Theorem 3.3.2. ([Bal00, Theorem 6.2 and Corollary/Definition 5.16]) If we are given a localization of triangulated categories with duality $\mathcal{J} \xrightarrow{j} \mathcal{T} \xrightarrow{q} S^{-1}\mathcal{T}$ such that $A \oplus B \cong B$ implies $A \cong 0$ in $S^{-1}\mathcal{T}$ (i.e. $S^{-1}\mathcal{T}$ is weakly cancellative), $\frac{1}{2} \in \mathcal{T}$, and \mathcal{T} satisfies the axiom (TR4⁺) in [Bal00, Section 1], we obtain a long exact sequence for $n \in \mathbb{Z}$

$$\dots \to \mathbf{W}^{[n-1]}(S^{-1}\mathcal{T}) \xrightarrow{d^{n-1}} \mathbf{W}^{[n]}(\mathcal{J}) \xrightarrow{\mathbf{W}^{n}(j)} \mathbf{W}^{[n]}(\mathcal{T})$$

$$\xrightarrow{\mathbf{W}^{[n]}(q)} \mathbf{W}^{[n]}(S^{-1}\mathcal{T}) \xrightarrow{d^{n}} \mathbf{W}^{[n+1]}(\mathcal{J}) \to \dots,$$

where d^n is the well-defined homomorphism

 $d^{n}: \mathbf{W}^{[n]}(S^{-1}(\mathcal{T})) \to \mathbf{W}^{[n+1]}(\mathcal{J}), x \mapsto [\operatorname{cone}(A, s)]$

with (A, s) being any S-space (see [Bal00], Definition 5.4) such that x = [q(A), q(s)].

Walter then derives the following localization theorem involving Grothendieck–Witt groups.

Theorem 3.3.3. ([Wal03, Localization Theorem 2.4]) For $n \in \mathbb{N}$, Balmer's long exact sequence from above extends to an exact sequence

$$\operatorname{GW}^{[n]}(\mathcal{J}) \to \operatorname{GW}^{[n]}(\mathcal{T}) \to \operatorname{GW}^{[n]}(S^{-1}\mathcal{T}) \xrightarrow{d^n} \operatorname{W}^{[n+1]}(\mathcal{J}) \to \operatorname{W}^{[n+1]}(\mathcal{T}) \to \dots$$

Now, to establish a localization sequence in the setting of complicial exact categories with weak equivalences and duality, it is not necessary for our purposes to state it in the full generality of [Sch17, Theorem 6.6] but only for $(Ch^b(P(X))^{(n)}, quis)$ and $(Ch^b(P(X))^{(n)}, quis^{n'})$ from Example 3.2.11.

Theorem 3.3.4. ([FS09, Theorem 8], [Sch17, Theorem 6.6]) For a k-scheme X with char(k) $\neq 2$, a line bundle L, and $n, n' \in \mathbb{N}$ such that $n' \geq n$, consider the functors of complicial exact categories with weak equivalences and duality

$$(\mathrm{Ch}^{b}(P(X))^{(n')},\mathrm{quis}) \xrightarrow{I_{\mathrm{Ch}}} (\mathrm{Ch}^{b}(P(X))^{(n)},\mathrm{quis}) \xrightarrow{Q_{\mathrm{Ch}}} (\mathrm{Ch}^{b}(P(X))^{(n)},\mathrm{quis}^{n'}),$$

where all dualities are derived from $\operatorname{Hom}_{\mathcal{O}_X}(-, L)$, see Example 3.2.11. They yield a long exact sequence for all $i, m \in \mathbb{Z}$

$$\dots \to \operatorname{GW}_{m}^{[i]}(\operatorname{Ch}^{b}(P(X))^{(n')}, \operatorname{quis}) \xrightarrow{\substack{i_{\operatorname{Ch}}^{i,m} \\ \bigoplus}} \operatorname{GW}_{m}^{[i]}(\operatorname{Ch}^{b}(P(X))^{(n)}, \operatorname{quis}) \xrightarrow{\substack{q_{\operatorname{Ch}}^{i,m} \\ \longrightarrow}} \operatorname{GW}_{m}^{[i]}(\operatorname{Ch}^{b}(P(X))^{(n)}, \operatorname{quis}^{n'})$$
$$\dots \xrightarrow{d_{\operatorname{Ch}}^{m}} \operatorname{GW}_{m-1}^{[i]}(\operatorname{Ch}^{b}(P(X))^{(n)}, \operatorname{quis}) \xrightarrow{\substack{i_{\operatorname{Ch}}^{i,m-1} \\ \bigoplus}} \operatorname{GW}_{m-1}^{[i]}(\operatorname{Ch}^{b}(P(X))^{(n)}, \operatorname{quis}) \to \dots$$

The sequence starting at $\mathrm{GW}_0^{[i]}(\mathrm{Ch}^b(P(X))^{(n')})$ coincides with the sequence

$$\operatorname{GW}^{[i]}(\operatorname{Ch}^{b}(P(X))^{(n')}[(\operatorname{quis})^{-1}]) \xrightarrow{i^{i}} \operatorname{GW}^{[i]}(\operatorname{Ch}^{b}(P(X))^{(n)}[(\operatorname{quis})^{-1}]) \xrightarrow{q^{i}} \operatorname{GW}^{[i]}(\operatorname{Ch}^{b}(P(X))^{(n)}[(\operatorname{quis})^{-1}]) \xrightarrow{d^{i}} \operatorname{W}^{[i+1]}(\operatorname{Ch}^{b}(P(X))^{(n')}[(\operatorname{quis})^{-1}]) \xrightarrow{\dots} \dots$$

of triangulated categories with duality from Theorem 3.3.3 for the localization

$$\operatorname{Ch}^{b}(P(X))^{(n')}[(\operatorname{quis})^{-1}] \to \operatorname{Ch}^{b}(P(X))^{(n)}[(\operatorname{quis})^{-1}] \to \operatorname{Ch}^{b}(P(X))^{(n)}[(\operatorname{quis}^{n'})^{-1}].$$
In this chapter, we have seen localization sequences for Grothendieck–Witt and Witt groups in different settings. On the one hand, Balmer and Walter give a localization sequence for triangulated categories with duality. On the other hand, Fasel, Srinivas, and Schlichting show localization for complicial exact categories with weak equivalences and duality. These sequences coincide by construction, as Fasel and Srinivas state.

Having introduced (Grothendieck–)Witt groups and localization for triangulated and exact categories in this chapter and having shown their agreement, we now have enough foundation to generalize the definition of Chow–Witt groups of schemes to Chow–Witt groups of rigid tensor triangulated categories in the next chapter and prove the agreement of these two approaches.

Chapter 4

Tensor Triangular Chow–Witt Groups

In this work's central chapter, we construct *tensor triangular Chow–Witt groups* and justify their name by showing the equivalence to Chow–Witt groups of a regular scheme X for the derived category of perfect complexes $D^{perf}(X)$.

In order to do so, we refer to two underlying generalization concepts. On the one hand, we let the construction be inspired by the passage from classical Chow groups to classical Chow–Witt groups. On the other hand, Balmer's definition and Klein's work on tensor triangular Chow groups ([Bal13, Kl16a, Kl16b]) will lead the way to lift the algebro-geometric definition to the tensor triangular setting.

Consequently, this chapter needs to begin with recalling these underlying ideas. In Section 4.1, Balmer's idea of tensor triangular Chow groups and Klein's work on the agreement with the algebro-geometric definition are reconstructed. After that, Section 4.2 will give an overview of Chow–Witt groups of a scheme, before we can sketch the idea and give the concrete definition of tensor triangular Chow–Witt groups in Section 4.3, but with the restriction of certain quotients being idempotent complete. Section 4.4 is then dedicated to the proof of the agreement theorem Theorem 4.4.5, making use of the foundations we presented in Chapter 2 and Chapter 3. Section 4.5 treats the functoriality of the construction before Section 4.6 focuses on the case when the quotients mentioned above are not idempotent complete, which will lead to the introduction of *models*.

4.1 Tensor Triangular Chow Groups

A standard reference for classical Chow groups of schemes is [Ful98]. Let us first recall the definition of tensor triangular Chow groups that was first presented by Paul Balmer in [Bal13] and elaborated by Sebastian Klein in [Kl16a] and [Kl16b]. We closely follow [Kl16a] here.

The rough idea is to use a definition of classical Chow groups based on a long exact sequence in K-theory alternative to the original one of cycles modulo rational equivalence (compare [Ful98, Section 1.3]). This sequence comes from

a filtration of the abelian category of coherent sheaves $\operatorname{Coh}(X)$ on a separated, non-singular scheme X of finite type over a base field k by codimension of support. This filtration can be transferred to the tensor triangular setting as introduced in Section 2.3. This way, the definition of Chow groups as the cohomology of the *Rost-Schmid complex* at a certain point can be generalized.

Convention 4.1.1. Let \mathcal{T} be an essentially small rigid tensor triangulated category equipped with a dimension function dim and such that $\operatorname{Spc}(\mathcal{T})$ is noetherian. Moreover, let X be a regular, separated scheme of finite type over a field k.

Klein begins with considering a filtration of the abelian category of coherent sheaves $\operatorname{Coh}(X)$

$$\dots \subset M^n \subset M^{n-1} \subset \dots \subset M^0 = \operatorname{Coh}(X),$$

where M^n is the abelian subcategory of $\operatorname{Coh}(X)$ with $\operatorname{codim}(\operatorname{supp}(A)) \geq n$ for A in $\operatorname{Coh}(X)$. Since the components of this filtration are moreover *Serre subcate*gories of the respective next component, one obtains quotient abelian categories M^n/M^{n+1} and, thus, short exact sequences

$$M^{n+1} \to M^n \to M^n / M^{n+1}.$$

These yield a long exact sequence in K-theory

$$\dots \to K_m(M^{n+1}) \xrightarrow{i_m^n} K_m(M^n) \xrightarrow{q_m^n} K_m(M^n/M^{n+1})$$
$$\xrightarrow{b_m^n} K_{m-1}(M^{n+1}) \xrightarrow{i_{m-1}^n} K_{m-1}(M^n) \xrightarrow{q_{m-1}^n} K_{m-1}(M^n/M^{n+1}) \to \dots$$

One considers the composition

$$d_1: K_1(M^{n-1}/M^n) \xrightarrow{b_1^{n-1}} K_0(M^n) \xrightarrow{q_0^n} K_0(M^n/M^{n+1})$$

and uses Quillen's theorem [Qui73, Chapter 7, Proposition 5.14] to prove that the image of d_1 equals the subgroup of codimension-*n*-cocycles rationally equivalent to 0. That means, one has

$$Z^n(X) / \operatorname{Im}(d_1) \cong \operatorname{CH}^n(X),$$

where $Z^n(X) = K_0(M^n/M^{n+1})$, compare [Kl16a, Remark 3.1.3]. Moreover, this proves that, by using the exactness of the above long exact sequence in K-theory of abelian categories, we have

$$\operatorname{Im}(d_1) = \operatorname{Im}(q_0^n \circ b_1^{n-1}) = q_0^n(\operatorname{Im}(b_1^{n-1})) = q_0^n(\ker(i_0^{n-1})).$$

Thus, the codimension-*n*-cocycles rationally equivalent to 0 are, moreover, equivalent to $q_0^n(\ker(i_0^{n-1}))$, which gives us an inspiration for the tensor triangular definition.

The maps i_0^n and q_0^n are straightforward to translate into the tensor triangular setting. For a tensor triangulated category \mathcal{T} with dimension function, we obtain a filtration by dimension of support as mentioned in Section 2.3

$$\mathcal{T}_{(-\infty)} \subset ... \subset \mathcal{T}_{(n-1)} \subset \mathcal{T}_{(n)} \subset ... \subset \mathcal{T}_{(\infty)} = \mathcal{T}.$$

At first sight, this mimics the filtration of the category $\operatorname{Coh}(X)$ of coherent sheaves, but keep in mind that $\operatorname{Coh}(X)$ is *not* a tensor triangulated category since it is not even triangulated. Moreover, one has to be careful about the indices since the sign will change when passing to the tensor triangular setting later.

For the agreement with the algebro-geometric case, we will make use of the passage to its bounded derived category $D^b(Coh(X))$, but first, let us consider the following diagram of triangulated categories and functors coming from the above filtration

$$\begin{array}{c} \mathcal{T}_{(n)} \xrightarrow{I_{\mathcal{T}}^{n+1}} \mathcal{T}_{(n+1)} \\ \downarrow^{Q_{\mathcal{T}}^{n}} \\ \mathcal{T}_{(n)}/\mathcal{T}_{(n-1)} \\ \downarrow^{J_{\mathcal{T}}^{n}} \\ (\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)})^{\natural} \end{array}$$

for each $n \in \mathbb{Z}$.

Here, $I_{\mathcal{T}}^{n+1}$ is the inclusion functor, $Q_{\mathcal{T}}^n$ the localization functor, and $J_{\mathcal{T}}^n$ the idempotent completion functor.

We will later explain why we need to include the idempotent completion $J^n_{\mathcal{T}}$ in this process, but first, recall that the idempotent completion of a triangulated category naturally inherits the structure of a triangulated category ([BS01]). Now, one can apply K_0 to this diagram to obtain

$$\begin{array}{cccc}
K_0(\mathcal{T}_{(n)}) & \xrightarrow{i_{\mathcal{T}}^{n+1}} & K_0(\mathcal{T}_{(n+1)}) \\ \downarrow^{q_{\mathcal{T}}^n} \\ K_0(\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)}) \\ & & \int_{\mathcal{T}_{\mathcal{T}}}^{j_{\mathcal{T}}^n} \\ K_0((\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)})^{\natural}), \end{array} \tag{4.1}$$

which we call the *defining diagram* of tensor triangular Chow groups. Inspired by the alternative definition of cocycles rationally equivalent to 0 as $q_0^n(\ker(i_0^{n-1}))$, triangular algebraic cocycles and Chow groups are defined as follows.

Definition 4.1.2. For an essentially small rigid tensor triangulated category $(\mathcal{T}, \otimes, \mathbb{1})$ equipped with a dimension function dim such that $\operatorname{Spc}(\mathcal{T})$ is noetherian, its *n*-th tensor triangular Chow group, $n \in \mathbb{Z}$, is defined as

$$\operatorname{CH}^{n}_{\Delta}(\mathcal{T}) := \overset{Z^{n}_{\Delta}(\mathcal{T})}{j_{\mathcal{T}}^{n} \circ q_{\mathcal{T}}^{n}(\ker(i_{\mathcal{T}}^{n+1}))},$$

where the n-th tensor triangular cocycles are given as follows

$$Z^n_{\Delta}(\mathcal{T}) := K_0((\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)})^{\natural})$$

and $j_{\tau}^{n}, q_{\tau}^{n}$ and i_{τ}^{n+1} are the morphisms from Diagram 4.1.

To see the agreement of this definition to the original Chow Witt groups of a regular scheme X for the tensor triangulated category $D^{perf}(X)$, see Definition 2.1.18, there are a few things left to show.

First, Klein proves that tensor triangular cocycles and algebraic cocycles are equivalent for that special case. He makes use of theorems on the algebraic geometric side as well as on the tensor triangular side as follows.

Quillen's dévissage theorem yields equivalences for all $n \ge 0$

$$K_0(M^n/M^{n+1}) \cong \bigoplus_{x \in X^{(n)}} K_0(k(x)) \cong \bigoplus_{x \in X^{(n)}} \mathbb{Z} \cong Z^n(X),$$

showing that the Grothendieck group of the quotient M^n/M^{n+1} obtained by the filtration of $\operatorname{Coh}(X)$ by codimension of support equals the usual cocycle groups $Z^n(X)$ of a regular scheme X. This is an elegant way to overcome the concrete description of algebraic cocycles and instead use an alternative definition that we can translate more directly into the tensor triangular settings.

In the tensor triangular case, using Balmer's Theorem 2.3.9, we obtain an equivalence after applying K_0

$$Z_{\Delta}^{n}(\mathcal{T}) = K_{0}((\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)})^{\natural}) \cong \bigoplus_{\substack{\mathcal{P} \in \operatorname{Spc}(\mathcal{T}) \\ \dim(\mathcal{P}) = n}} K_{0}(\operatorname{Min}(\mathcal{T}_{P})).$$

The necessity of the idempotent completion for the equivalence in Theorem 2.3.9 now motivates us to include the idempotent completion functors $J^n_{\mathcal{T}}$ and the maps $j^n_{\mathcal{T}}$, which will become the identity in the algebro-geometric case for regular X (see below).

Eventually, Klein proves the following.

Lemma 4.1.3. ([Kl16a, Corollary 3.1.9]) For a noetherian, regular scheme X and $\mathcal{T} := D^{\text{perf}}(X)$ equipped with the dimension function $- \operatorname{codim}_{Krull}$, there is an isomorphism for all $n \geq 0$

$$Z^{-n}_{\Delta}(\mathcal{T}) \xrightarrow{\cong} Z^n(X).$$

After having generalized the definition of algebraic cocycles and seen their agreement, the second problem in the construction is to compare the defining diagram in the algebro-geometric setting to the tensor triangulated case. We at least have homomorphisms (the diagonal maps in Diagram 4.2) between the diagrams given by the formula

$$[C^{\bullet}] \mapsto \sum_{i} (-1)^{i} [H^{i}(C^{\bullet})],$$

yielding the following diagram

$$K_{0}(\mathbf{D}^{b}(X)_{(n)}) \xrightarrow{i_{\mathcal{T}}^{n+1}} K_{0}(\mathbf{D}^{b}(X)_{(n+1)})$$

$$\downarrow^{q_{\mathcal{T}}^{n}} K_{0}(\mathbf{D}^{b}(X)_{(n)}/\mathbf{D}^{b}(X)_{(n-1)}) \xrightarrow{K_{0}(M^{-n})} K_{0}(M^{-n-1})$$

$$\downarrow^{q_{0}^{-n}}$$

$$K_{0}((\mathbf{D}^{b}(X)_{(n)}/\mathbf{D}^{b}(X)_{(n-1)})^{\natural}) \xrightarrow{K_{0}(M^{-n}/M^{-n+1})}.$$

$$(4.2)$$

For the justification of the construction of tensor triangular Chow groups, it is now left to show that the diagonal arrows are isomorphisms and the map $j_{\mathcal{T}}^n$ is the identity. For this, Klein applies the following theorem.

Theorem 4.1.4. ([Kel99, Section 1.15]) Let \mathcal{A} be a Serre subcategory of an abelian category \mathcal{B} and let the following criterion be satisfied:

If $0 \to A \to B \to C \to 0$ is an exact sequence in \mathcal{B} with A in \mathcal{A} , then there is a commutative diagram with exact rows



such that A', A'' are in \mathcal{A} .

Then, the inclusion induces a triangulated equivalence of triangulated categories $D^b(\mathcal{A}) \xrightarrow{\simeq} D^b_{\mathcal{A}}(\mathcal{B})$, where $D^b_{\mathcal{A}}(\mathcal{B}) \subset D^b(\mathcal{B})$ is the full subcategory of complexes with homology in \mathcal{A} .

Moreover, the sequence

$$\mathrm{D}^{b}(\mathcal{A}) \to \mathrm{D}^{b}(\mathcal{B}) \to \mathrm{D}^{b}(\mathcal{B}/\mathcal{A})$$

is exact, i.e. $D^b(\mathcal{B}/\mathcal{A}) \simeq D^b(\mathcal{B})/D^b(\mathcal{A})$.

The conditions are satisfied by [Kl16a, Lemma 3.2.5] if X is regular, so, one obtains an equivalence

$$\mathbf{D}^{b}(M^{-n}) \simeq \mathbf{D}^{b}_{M^{-n}}(\mathrm{Coh}(X)) = \mathbf{D}^{b}(X)_{(n)}$$

for all n, where $D^b_{M^{-n}}(Coh(X))$ consists of the complexes in $D^b(Coh(X))$ with homology in M^{-n} . Moreover, there is an equivalence of triangulated categories

$$D^{b}(M^{-n}/M^{-n+1}) \simeq D^{b}(M^{-n})/D^{b}(M^{-n+1}).$$

This again establishes equivalences

$$D^{b}(X)_{(n)}/D^{b}(X)_{(n-1)} \simeq D^{b}(M^{-n}/M^{-n+1}).$$

Moreover, we know that there are isomorphisms $K_0(D^b(\mathcal{A})) \cong K_0(\mathcal{A})$ for abelian categories \mathcal{A} (see e.g. [Wei13, Chapter II, Theorem 9.2.2] which is essentially [SGA6, I.6.4] for Waldhausen categories or [Nee05, Theorem 3] for triangulated categories). In particular, this holds for all components of the filtration of Coh(X)and its quotients since Serre quotients of abelian categories are abelian. This proves that the diagonal maps in Diagram 4.2) are isomorphisms.

Moreover, the derived category of an abelian category is always idempotent complete, hence so is $D^b(X)_{(n)}/D^b(X)_{(n-1)}$ for all n by the above equivalences. As a consequence, $J^n_{\mathcal{T}}$ and thus also $j^n_{\mathcal{T}}$ are the identity.

In [Kl16a, Theorem 3.2.6], the results are summarized and the agreement of tensor triangular and classical Chow groups is shown for the category $D^{perf}(X)$ for a separated, regular scheme of finite type.

We have seen in this section the construction of tensor triangular Chow groups and the agreement with the classical definition for schemes when considering the tensor triangulated category $D^{\text{perf}}(X)$. The main idea was to use an alternative definition of Chow groups as the cohomology of a chain complex arising from the filtration of the category Coh(X) of coherent sheaves by codimension of support. This filtration can be translated to the tensor triangular setting, giving rise to the definition of tensor triangular cocycles and coboundaries, and, eventually, of tensor triangular Chow groups.

In the next section, basics on Chow–Witt groups of a scheme will be recalled serving as the second inspiration of how Chow–Witt groups of a tensor triangulated category could be constructed.

4.2 Chow–Witt Groups of a Scheme

Chow-Witt groups were originally constructed by Barge and Morel in [BM00] to answer the question of when a projective A-module is isomorphic to the direct sum $Q \oplus A$ for some projective A-module Q. To approach this problem, they constructed the *Euler class* $\tilde{c}_n(E)$ of a vector bundle E as a refinement of the top Chern class $c_n(E)$. One solution to the problem now breaks down to determining whether the Euler class (or Chern class) is zero, see [BM00] for more details. More information on Chow-Witt groups of schemes can be found in e.g. [Fas07, Fas08]. We will mainly follow the latter here.

Convention 4.2.1. Let X be a separated, regular scheme of finite type over a perfect field k of characteristic $\neq 2$.

The Chow–Witt groups of X can be defined as the cohomology of a complex obtained as the fiber product



whose components we will define below. If twists by a graded line bundle L are included, the square changes as follows

With this fiber product at hand, one defines:

Definition 4.2.2. Let X be a separated, regular k-scheme of finite type. We define the n-th Chow-Witt group of X as

$$\widetilde{\operatorname{CH}}^n(X) := H^n(C(X, G^n)) = \widetilde{Z}^n(X) / \widetilde{B}^n(X)$$

and the Chow-Witt groups twisted by L for a fixed line bundle L

$$\widetilde{\operatorname{CH}}^{n}(X,L) := H^{n}(C(X,G^{n},L)) = \widetilde{Z}^{n}(X,L)/\widetilde{B}^{n}(X,L).$$

We denote the differentials of $C(X, G^n)$ and $C(X, G^n, L)$ by ∂^n and will talk about these later.

Here, $\widetilde{Z}^n(X)$ (and $\widetilde{Z}^n(X, L)$) are called the *(twisted) oriented cocycles* and defined as the kernel of the complex at n. Similarly, $\widetilde{B}^n(X)$ (and $\widetilde{B}^n(X, L)$) denote the *(twisted) oriented coboundaries* and define the image of the complex at n.

To understand the complex $C(X, K_n^{\mathrm{M}})$, we recall that Milnor K-theory $K_*^{\mathrm{M}}(F)$ of a field F is given as the graded tensor algebra

$$T(F^{\times}) = \mathbb{Z} \oplus F^{\times} \oplus (F^{\times} \otimes F^{\times}) \oplus \dots$$

divided by the equivalence relation $a \otimes (1-a) = 0$. The Milnor K-groups $K_n^{\mathrm{M}}(F)$ are then given by the *n*-th homogene components, see [Mil70].

The Milnor-K-theory complex is then given as follows. Note that the differentials are slightly complicated and can be found in detail, for example, in [Fas08, Chapter 2].

Definition 4.2.3. ([Fas08, Théorème 2.2.13]) For a noetherian scheme X over a field k and any $n \in \mathbb{Z}$, we define the *Milnor-K-theory complex* $C(X, K_n^M)$ as

$$0 \longrightarrow \bigoplus_{x \in X^{(0)}} K_n^{\mathcal{M}}(k(x)) \stackrel{d}{\longrightarrow} \bigoplus_{x \in X^{(1)}} K_{n-1}^{\mathcal{M}}(k(x)) \longrightarrow \dots$$
$$\dots \longrightarrow \bigoplus_{x \in X^{(m)}} K_{n-m}^{\mathcal{M}}(k(x)) \stackrel{d}{\longrightarrow} \bigoplus_{x \in X^{(m+1)}} K_{n-m-1}^{\mathcal{M}}(k(x)) \longrightarrow \dots$$

Remark 4.2.4. Grading this complex by the codimension of points in X, one obtains isomorphism for all n

$$H^n(C(X, K_n^{\mathcal{M}})) \cong CH^n(X),$$

where the right-hand side denotes the Chow groups defined as cocycles modulo rational equivalence as in [Ful98, Section 1.3].

The other complexes involved in the definition of $C(X, G^n, L)$ come from the *Gersten-Witt complex* C(X, W, L) by restriction to powers of the *fundamental ideals* or quotients of these. First, let us take a look at the construction of the Gersten-Witt complex.

Let P(X) be the exact category of locally free \mathcal{O}_X -modules of finite type over X. Then, by [Bal99, Proposition 2.3], $(D^b(P(X)), R \operatorname{Hom}_{\mathcal{O}_X}(-, L))$ is a triangulated category with duality, and since X is regular, there is an equivalence $D^b(P(X)) \simeq D^b(\operatorname{Coh}(X))$ of triangulated categories with duality when considering the duality $R \operatorname{Hom}(-, L)$ on $D^b(\operatorname{Coh}(X))$.

Now, we can define the following subcategories analogously to Example 3.2.11.

Definition 4.2.5. The homological support supph(P_{\bullet}) of $P_{\bullet} \in D^b(P(X))$ is defined as

$$\operatorname{supph}(P_{\bullet}) := \bigcup_{i} \operatorname{supp}(H_i(P_{\bullet})).$$

Definition 4.2.6. We define for each $n \in \mathbb{N}$

$$D^{b}(P(X))^{(n)} = \{P_{\bullet} \in D^{b}(P(X)) \mid \operatorname{codim}(\operatorname{supph}(P_{\bullet})) \ge n\}.$$

This way, we obtain a filtration for $\dim(X) = d$

$$0 = D^{b}(P(X))^{(d+1)} \subset D^{b}(P(X))^{(d)} \subset ... \subset D^{b}(P(X))^{(0)} = D^{b}(P(X)).$$

As in Section 4.1, the indices of the filtration decrease with each step. Note that the filtrations of triangulated categories as in Section 2.3 have *increasing* indices, which has as a consequence that the signs swap when comparing tensor triangular Chow(–Witt) groups and Chow(–Witt) groups of a scheme. More details will be given in the proof of Theorem 4.4.5.

Since $D^b(P(X))^{(n)}$ is a thick subcategory of $D^b(P(X))$ for all $n \in \mathbb{N}$, we also have quotients of the filtration components and exact sequences of triangulated categories with duality

$$\mathrm{D}^{b}(P(X))^{(n+1)} \xrightarrow{I^{n}} \mathrm{D}^{b}(P(X))^{(n)} \xrightarrow{Q^{n}} \mathrm{D}^{b}(P(X))^{(n)} / \mathrm{D}^{b}(P(X))^{(n+1)},$$

where all the functors are duality-preserving when considering the dualities induced by $R \operatorname{Hom}_{\mathcal{O}_X}(-, L)$ for a fixed line bundle L.

In particular, I^n and Q^n induce maps on Witt groups and Grothendieck–Witt groups for $i\in\mathbb{Z}$

$$\overset{W^{[i]}(\mathcal{D}^{b}(P(X))^{(n+1)})}{\longrightarrow} \overset{i^{n,W}}{\longrightarrow} W^{[i]}(\mathcal{D}^{b}(P(X))^{(n)})$$
$$\overset{q^{n,W}}{\longrightarrow} W^{[i]}(\mathcal{D}^{b}(P(X))^{(n)} / \mathcal{D}^{b}(P(X))^{(n+1)})$$

and

$$\operatorname{GW}^{[i]}(\mathrm{D}^{b}(P(X))^{(n+1)}) \xrightarrow{i^{n}} \operatorname{GW}^{[i]}(\mathrm{D}^{b}(P(X))^{(n)}) \xrightarrow{q^{n}} \operatorname{GW}^{[i]}(\mathrm{D}^{b}(P(X))^{(n)} / \operatorname{D}^{b}(P(X))^{(n+1)}),$$

respectively. One also obtains a boundary morphism by [Bal00, Theorem 6.2/Corollary and Definition 5.16] which is exactly the one from Theorem 3.3.3,

$$d_n^i: W^{[i]}(D^b(P(X))^{(n)}/D^b(P(X))^{(n+1)}) \to W^{[i+1]}(D^b(P(X))^{(n+1)})$$

and, hence, a homomorphism

$$\widetilde{\partial}_n^i = q^{i+1,W} \circ d_n^i : W^{[i]}(D^b(P(X))^{(n)} / D^b(P(X))^{(n+1)}) \to W^{[i+1]}(D^b(P(X))^{(n+1)} / D^b(P(X))^{(n+2)}).$$

If i = n, we write $d_n^i = d^n$ and $\widetilde{\partial}_n^i = \widetilde{\partial}^n$.

We can already define a complex with these components and differentials, but to get to the original definition of the Gersten–Witt complex, we consider the following.

Definition 4.2.7. (i) Let A be a regular local ring of dimension d and A- mod_{fl} be the category of A-modules of finite length. Then, the derived Hom functor

$$\operatorname{Ext}_{A}^{d}(-,A): (A\operatorname{-mod}_{fl})^{\operatorname{op}} \to A\operatorname{-mod}_{fl}$$

yields a duality on A-mod_{fl}, turning (A-mod_{fl}, Ext^d_A(-, A)) into an exact category with duality, see [Fas08, Section 3.3].

(ii) We define the Witt group of finite length modules as

$$W^{fl}(A) := W(A \operatorname{-mod}_{fl}, \operatorname{Ext}^d_A(-, A)).$$

(iii) The category of A-modules of finite type shall be denoted by $A - \text{mod}_{ft}$.

Theorem 4.2.8. ([Bal01, Theorem 4.3], [BW02, Lemma 6.4]) We have an isomorphism

$$W^{fl}(A) \xrightarrow{\cong} W^{[0]}(D^b(A \operatorname{-mod}_{fl}), \operatorname{Ext}^d_A(-, A))$$

and duality-preserving categorical equivalences

$$D^b_{fl}(P(A)) \xrightarrow{\simeq} D^b_{fl}(A\operatorname{-mod}_{ft}) \xleftarrow{\simeq} D^b(A\operatorname{-mod}_{fl}),$$

where $\mathrm{D}^b_{fl}(-)$ is the subcategory of complexes with finite length homology yielding an isomorphism

$$W^{[d]}(D^b_{fl}(P(A))) \xrightarrow{\cong} W^{[0]}(D^b(A\operatorname{-mod}_{fl},\operatorname{Ext}^d_A(-,A))).$$

Here, the duality of the bounded derived category of complexes with finite length homology is induced by the duality on the respective original bounded derived category. We will now use a concrete form of Theorem 2.3.9/Corollary 2.3.10, namely [BW02, Proposition 7.1], to obtain isomorphisms for all n and i after applying W

$$W^{[i]}(D^{b}(P(X))^{(n)}/D^{b}(P(X))^{(n+1)}) \xrightarrow{\cong} \bigoplus_{x \in X^{(n)}} W^{[i]}(D^{b}_{fl}(P(\mathcal{O}_{X,x}))).$$

Now, we can finally define the Gersten–Witt complex, using the above considerations.

Definition 4.2.9. For a regular scheme X of finite dimension, we define the *Gersten-Witt complex* as

$$\dots \longrightarrow \bigoplus_{x \in X^{(n)}} W^{fl}(\mathcal{O}_{X,x}) \stackrel{d}{\longrightarrow} \bigoplus_{y \in X^{(n+1)}} W^{fl}(\mathcal{O}_{X,y}) \longrightarrow \dots$$

and denote it by C(X, W).

Remark 4.2.10. One can instead equip $D^b(X)$ with the duality derived from Hom(-, L) for an invertible \mathcal{O}_X -module L and obtains analogously the twisted versions $W^{fl}(A, L)$ and C(X, W, L). Note that the components $C^n(X, W)$ and $C^n(X, W, L)$ are isomorphic (but not necessarily canonically).

We will briefly recall the pullback square of the defining complex of Chow–Witt groups.

$$\begin{array}{c} C^{i}(X,G^{j},L) \longrightarrow C^{i}(X,I^{j},L) \\ \downarrow \\ \downarrow \\ C^{i}(X,K^{\mathrm{M}}_{j}) \longrightarrow C^{i}(X,I^{j}/I^{j+1}) \end{array}$$

We are now familiar with the Milnor-K-theory complex and the Gerste–Witt complex. To understand the complexes $C^i(X, I^j, L)$ and $C^i(X, I^j/I^{j+1})$, we will need the following lemma.

Lemma 4.2.11. ([MH73, Chapter III, §3, Lemma 3.3]) Let F be a field. There exists a unique ideal I in W(F) such that W(F)/ $I \cong \mathbb{F}_2$. This ideal (consisting of all Witt classes of even dimension) is called the fundamental ideal of W(F).

We want to restrict the Gersten–Witt complex to a generalized version of this fundamental ideal.

Definition 4.2.12. Let *B* be a regular local ring of dimension *n* with residue field *F*. For any generator β of $\operatorname{Ext}_{B}^{n}(F, B)$, we define the *m*-th fundamental ideal $I^{m}(B) = I_{m}^{fl}(B)$ as

$$I_m^{fl}(B) := \phi_\beta(I^m(F)),$$

where $\phi_{\beta} : W(F) \to W^{fl}(B)$ is an isomorphism induced by β and $\phi_{\beta}(I^m(F))$ is independent of the choice of the generator β , compare [Fas08, Section 9.2]

To restrict the Gersten–Witt complex to the fundamental ideals, we need to check if the differentials behave well with these ideals, which is answered by the following theorem. **Proposition 4.2.13.** ([Fas08, Theorem 9.2.4]) The differentials of the Gersten– Witt complex preserve the fundamental ideal.

Definition 4.2.14. We denote by $C(X, I^j)$ the complex

$$\dots \to \bigoplus_{x \in X^{(n)}} I^{j-n}(\mathcal{O}_{X,x}) \to \bigoplus_{x \in X^{(n+1)}} I^{j-(n+1)}(\mathcal{O}_{X,x}) \to \dots$$

graded by codimension of support. Analogously, for a fixed invertible \mathcal{O}_X -module L we denote by $C(X, I^j, L)$ the following complex

$$\dots \to \bigoplus_{x \in X^{(n)}} I^{j-n}(\mathcal{O}_{X,x}, L_x) \to \bigoplus_{x \in X^{(n+1)}} I^{j-(n+1)}(\mathcal{O}_{X,x}, L_x) \to \dots$$

The maps induced by the inclusion of powers of the fundamental ideals

$$C(X, I^{j+1}) \to C(X, I^j)$$
 and $C(X, I^{j+1}, L) \to C(X, I^j, L)$

have canonically isomorphic cokernels (see [Fas08, Corollary E.1.3]). Thus, we can define:

Definition 4.2.15. We denote by $C(X, I^j/I^{j+1})$ the cokernel of the above morphisms graded by codimension of support.

We now have defined the components of the pullback square

The arrow on the right is induced by the canonical projection. For the arrow on the bottom of the diagram, the reader is referred to [Fas08, Section 10.2]; the differentials of $C^i(X, G^j, L)$ are induced by the differentials on $C^i(X, K_j^M)$ and $C^i(X, I^j, L)$, see [Fas08, Lemma 10.2.12].

Remark 4.2.16. The complex $C^i(X, G^j, L)$ is called the Rost-Schmid complex and can also be defined directly as follows (in the for us important case n := j = i). Here, $\det(\Omega_{k(x)/k})$ is the determinant bundle of the differential sheaf, see [Fas20, Section 1.4], and L is a fixed invertible \mathcal{O}_X -module.

$$\bigoplus_{x \in X^{(n-1)}} K_1^{\mathrm{MW}}(k(x), \det(\Omega_{k(x)/k}) \otimes L)$$

$$\xrightarrow{\partial_1} \bigoplus_{x \in X^{(n)}} K_0^{\mathrm{MW}}(k(x), \det(\Omega_{k(x)/k}) \otimes L)$$

$$\xrightarrow{\partial_0} \bigoplus_{x \in X^{(n+1)}} K_{-1}^{\mathrm{MW}}(k(x), \det(\Omega_{k(x)/k}) \otimes L)$$

Note that the differentials are complicated, see for example [Fas20] for more details, and in their explicit form not necessary for our purposes.

The definition of Milnor–Witt K-theory of fields F is due to Hopkins and Morel and can be found in [Mor12, Chapter 3], where it is defined as a quotient of the \mathbb{Z} -graded ring A(F), generated in degree 1 by [a] for all $a \in F^{\times}$ and by one additional generator η in degree -1, defined by concrete relations. The basic idea we will make use of here is that we have isomorphisms $\mathrm{GW}(F) \cong K_0^{\mathrm{MW}}(F)$ and $\mathrm{W}(F) \cong K_{-i}^{\mathrm{MW}}(F)$ for all i > 0, see *loc.cit*.

Remark 4.2.17. It will be important for us to know that, by using [Fas07, Remark 3.23], there are isomorphisms

$$C^{n}(X, G^{n}, L) \cong \mathrm{GW}^{[n]}(\mathrm{D}^{b}(P(X))^{(n)} / \mathrm{D}^{b}(P(X))^{(n+1)}, R \operatorname{Hom}_{\mathcal{O}_{X}}(-, L)).$$

More details can be found in [Fas08, Remark 10.2.9].

Hence, we can rewrite the complex from Definition 4.2.2 as follows

$$\dots \to C^{n-1}(X, G^n, L) \xrightarrow{\partial^{n-1}} \mathrm{GW}^{[n]}(\mathrm{D}^b(P(X))^{(n)} / \mathrm{D}^b(P(X))^{(n+1)}, R \operatorname{Hom}_{\mathcal{O}_X}(-, L))$$
$$\xrightarrow{\tilde{\partial}^n} \mathrm{W}^{[n+1]}(\mathrm{D}^b(P(X))^{(n+1)} / \mathrm{D}^b(P(X))^{(n+2)}, R \operatorname{Hom}_{\mathcal{O}_X}(-, L)) \to \dots.$$

Here, $\tilde{\partial}^n$ is defined as $q^{n+1,W} \circ d^n$, where d^n is the boundary map from Theorem 3.3.3.

Remark 4.2.18. Let us recall some results about the connection between Chow groups and Chow–Witt groups.

- (i) Let X be a scheme of dimension d over an algebraically closed field k, then there is a natural isomorphism between $\widetilde{\operatorname{CH}}^d(X)$ and $\operatorname{CH}^d(X)$, see [Fas08, Remarque 10.2.16].
- (ii) If we replace K^{MW} in the Rost–Schmid complex as in the description before Remark 4.2.17 by K^{M} , we basically divide out the ideal generated by η . Since η commutes with the differentials, we obtain differentials on the level of Milnor K-theory. The cohomology at i = -n now yields Chow groups instead of Chow–Witt groups, hence we get a homomorphism

$$\widetilde{\operatorname{CH}}^n(X,L) \to \operatorname{CH}^n(X),$$

which is, in general, neither injective nor surjective. For more details, compare [Fas13, HW19].

This section has served to recall the theory of Chow–Witt groups of a scheme. Together with the construction of tensor triangular Chow groups from Section 4.1, we can now glue these concepts together to sketch the idea, give the concrete construction, and prove the agreement theorem in the following two sections.

4.3 Tensor Triangular Chow–Witt Groups

We now come to the central section of the first part of this work, where we construct Chow–Witt groups of tensor triangulated categories. Before coming to the definition in Subsection 4.3.2, we give an overview of the idea in Subsection 4.3.1. This way, we prepare the main concepts of Section 4.4, where we prove the agreement of our definition to the classical definition of Chow–Witt groups of a scheme for the derived category of perfect complexes $D^{perf}(X)$.

Convention 4.3.1. Let \mathcal{T} be an essentially small rigid tensor triangulated category such that $\operatorname{Spc}(\mathcal{T})$ is noetherian and $\frac{1}{2} \in \mathcal{T}$. We fix a \otimes -invertible object L and a dimension function dim. When considered a triangulated category with duality, \mathcal{T} will be equipped with the duality $D_L := \hom(-, L)$ as in Example 2.2.6 (ii). If not mentioned explicitly otherwise, restrictions of \mathcal{T} will always carry the restriction of D_L .

Moreover, we will assume that all quotients of the form $\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)}$ are idempotent complete. This is true for a very wide range of examples, for example for $T = D^{\text{perf}}(X)$ when X is a regular scheme. We will comment on the case that the quotient is not idempotent complete for some n later in Section 4.6, see also [Kl16a, Remark 3.2.7].

4.3.1 Idea

Let X be a regular, separated scheme of finite type over a base field k of characteristic $\neq 2$. The natural idea to generalize Chow–Witt groups of a scheme is to generalize the oriented cocycles $\widetilde{Z}^n(X)$ and coboundaries $\widetilde{B}^n(X)$, hence, the kernel of the differential ∂^n and the image of the differential ∂^{n-1} from Definition 4.2.2.

The problem of this approach is that the differentials of the Rost–Schmid complex constructed in Section 4.2 are quite concrete and do not suggest a categorical generalization at first sight. However, we did see in Remark 4.2.17 that there is a way to rewrite at least a part of it in terms of triangulated categories with duality, namely by viewing Chow–Witt groups as the cohomology of the following complex:

$$\dots \to C^{n-1}(X, G^n) \xrightarrow{\partial^{n-1}} \mathrm{GW}^{[n]}(\mathrm{D}^b(P(X))^{(n)} / \mathrm{D}^b(P(X))^{(n+1)}, R \operatorname{Hom}_{\mathcal{O}_X}(-, L))$$
$$\xrightarrow{\widetilde{\partial}^n} \mathrm{W}^{[n+1]}(\mathrm{D}^b(P(X))^{(n+1)} / \mathrm{D}^b(P(X))^{(n+2)}, R \operatorname{Hom}_{\mathcal{O}_X}(-, L)) \to \dots.$$

Now, the map $\tilde{\partial}^n$ can be generalized directly to the tensor triangular setting since it already uses Grothendieck–Witt groups of triangulated categories with duality. Hence, we may construct a triangular form of $\tilde{\partial}^n$ when replacing $D^b(P(X))^{(n)}$ by filtration components $\mathcal{T}_{(n')}$ of suitable tensor triangulated categories.

We deliberately chose different indices here, since it is not clear at first sight whether n' = n or else and, in fact, we will see later that for $\mathcal{T} = D^{\text{perf}}(X)$, we have n' = -n. We briefly recall that, for an essentially small strongly tensor triangulated category equipped with a dimension function and a fixed \otimes -invertible object L, we have a filtration of triangulated categories with duality

$$\mathcal{T}_{(-\infty)} \subset \ldots \subset (\mathcal{T}_{(n-1)}, D_L|_{\mathcal{T}_{(n-1)}}) \subset (\mathcal{T}_{(n)}, D_L|_{\mathcal{T}_{(n)}}) \subset \ldots \subset \mathcal{T}_{(\infty)} = (\mathcal{T}, D_L).$$

We will omit the duality functor from the notation for better readability if we consider the duality D_L or its restrictions.

We obtain exact sequences of triangulated categories with duality and dualitypreserving functors for each $n \in \mathbb{Z} \cup \{\pm \infty\}$

$$\mathcal{T}_{(n-1)} \xrightarrow{I_{\mathcal{T}}^n} \mathcal{T}_{(n)} \xrightarrow{Q_{\mathcal{T}}^n} \mathcal{T}_{(n)} / \mathcal{T}_{(n-1)}$$

which induce exact sequences by Theorem 3.3.3

 $\operatorname{GW}^{[n]}(\mathcal{T}_{(n-1)}) \xrightarrow{i_{\mathcal{T}}^{n}} \operatorname{GW}^{[n]}(\mathcal{T}_{(n)}) \xrightarrow{q_{\mathcal{T}}^{n}} \operatorname{GW}^{[n]}(\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)})$

and by Theorem 3.3.2

$$W^{[n]}(\mathcal{T}_{(n-1)}) \xrightarrow{i^{n,W}_{\mathcal{T}}} W^{[n]}(\mathcal{T}_{(n)}) \xrightarrow{q^{n,W}_{\mathcal{T}}} W^{[n]}(\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)}).$$

The idea of generalizing the cocycles, i.e. the kernel of ∂^n , is now as follows: We know that $\ker(\partial^n) = \ker(\tilde{\partial}^n)$, and we can write $\tilde{\partial}^n$ in terms of maps coming from functors of the underlying triangulated categories with duality such as q_T^n or i_T^n . Of course, we need to take care of (lower and upper!) indices, but, for the details, the reader is referred to Subsection 4.3.2.

The coboundaries, i.e. the image of ∂^{n-1} , however, are not that easily generalized since the term $C^{n-1}(X, G^n)$ above of the Rost—Schmid complex is still in its concrete algebro-geometric form. We prove in Theorem 4.4.3, that, even though we might not be able to write ∂^{n-1} in terms of maps coming from functors of triangulated categories with duality, we can still do so for its image. Concretely, we show that the image of ∂^{n-1} can be written only in terms of maps q_T^n and i_T^n coming from the inclusion and quotient functor. The passage to complicial exact categories with weak equivalences and duality is essential here since there are no higher Grothendieck–Witt groups of triangulated categories with duality.

Then again, we can generalize the coboundaries and, hence, the whole concept of Chow–Witt groups.

Remark 4.3.2. The attentive reader might observe that the idea works mainly with the underlying triangulated categories with duality and the tensor structure does not appear too often in the sketch of the idea. The tensor structure and the additional information of a dimension function only serve as a basis to filtrate the category by dimension of support.

However, one can easily generalize the definition for essentially small triangulated categories \mathcal{C} with duality containing $\frac{1}{2}$ that allow a filtration $\mathcal{C}_{(-\infty)}$... \subset $\mathcal{C}_{(n-1)} \subset \mathcal{C}_{(n)} \subset \ldots \subset \mathcal{C}_{(\infty)} = \mathcal{C}$ of \mathcal{C} , where all the filtration components are triangulated categories with duality and the inclusion functors are duality-preserving.

Since the leading concept of this work is tensor triangular geometry, we specialize on filtrations coming from rigid tensor triangulated categories introcuced in Section 2.3.

4.3.2 Construction and Definition

Finally, we come to the construction of tensor triangular Chow–Witt groups. We still follow Convention 4.3.1.

Recall the following, for us central localization sequence from Theorem 3.3.3

$$\operatorname{GW}^{[n]}(\mathcal{T}_{(n-1)}) \xrightarrow{i_{\mathcal{T}}^{n}} \operatorname{GW}^{[n]}(\mathcal{T}_{(n)}) \xrightarrow{q_{\mathcal{T}}^{n}} \operatorname{GW}^{[n]}(\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)}) \xrightarrow{d_{\mathcal{T}}^{n}} \operatorname{W}^{[n+1]}(\mathcal{T}_{(n-1)}) \xrightarrow{i_{\mathcal{T}}^{n,\mathrm{W}}} \operatorname{W}^{[n+1]}(\mathcal{T}_{(n)}) \xrightarrow{q_{\mathcal{T}}^{n,\mathrm{W}}} \operatorname{W}^{[n+1]}(\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)}) \to \dots$$

To define tensor triangular Chow–Witt groups in terms of these maps, we consider the following defining diagram.

$$W^{[n+1]}(\mathcal{T}_{(n-2)}) \xrightarrow{i_{\mathcal{T}}^{n-1,W}} W^{[n+1]}(\mathcal{T}_{(n-1)}) \qquad GW^{[n]}(\mathcal{T}_{(n)}) \xrightarrow{i_{\mathcal{T}}^{n+1}} GW^{[n]}(\mathcal{T}_{(n+1)}) \xrightarrow{d_{\mathcal{T}}^{n}} \qquad \downarrow^{q_{\mathcal{T}}^{n}} \\ GW^{[n]}(\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)}) \qquad (4.3)$$

Then, we can give the definition of *tensor triangular oriented cocycles* and *coboundaries*.

Definition 4.3.3. For \mathcal{T} as in Convention 4.3.1, we define its *n*-th tensor triangular oriented cocycles (twisted by L) for $n \in \mathbb{Z}$ as

$$\widetilde{Z}^n_{\Delta}(\mathcal{T},L) := (d^n_{\mathcal{T}})^{-1}(\operatorname{Im}(i^{n-1,W}_{\mathcal{T}}))$$

and its *n*-th tensor triangular oriented coboundaries (twisted by L) as

$$\widetilde{B}^n_{\Delta}(\mathcal{T},L) := q^n_{\mathcal{T}}(\ker(i^{n+1}_{\mathcal{T}})),$$

where the maps are the ones from the defining diagram Diagram 4.3.

Proposition 4.3.4. In the setting of Definition 4.3.3, $\widetilde{B}^n_{\Delta}(\mathcal{T})$ is a subgroup of $\widetilde{Z}^n_{\Delta}(\mathcal{T})$ and $\widetilde{Z}^n_{\Delta}(\mathcal{T})$ is a subgroup of $\mathrm{GW}^{[n]}(\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)})$.

Proof. What we need to check is whether $q_{\mathcal{T}}^n(\ker(i_{\mathcal{T}}^{n+1}))$ is a subgroup of $\widetilde{Z}^n_{\Delta}(\mathcal{T}) := (d^n)_{\mathcal{T}}^{-1}(\operatorname{Im}(i_{\mathcal{T}}^{n-1,W}))$ and whether the latter is itself a subgroup of $\operatorname{GW}^{[n]}(\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)})$. We will omit the subscript \mathcal{T} from the notation here for better readability. However, be careful not to confuse it with the maps from the algebro-geometric setting in Definition 4.2.6.

The last statement follows from the fact that $(d^n)^{-1}(\operatorname{Im}(i^{n-1,W})) = \ker(q^{n-1,W} \circ d^n)$ by the exactness of the sequence from Theorem 3.3.3. Moreover, by construction, $q^n(\ker(i^{n+1}))$ is a subgroup of $\operatorname{GW}^{[n]}(\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)})$. It is left to check that

$$q^{n}(\ker(i^{n+1})) \subset (d^{n})^{-1}(\operatorname{Im}(i^{n-1,W})).$$

Since $\text{Im}(q^n) = \text{ker}(d^n)$ by exactness of the localization sequence (Theorem 3.3.3)

$$\operatorname{GW}^{[n]}(\mathcal{T}_{(n-1)}) \xrightarrow{i^{n}} \operatorname{GW}^{[n]}(\mathcal{T}_{(n)}) \xrightarrow{q^{n}} \operatorname{GW}^{[n]}(\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)})$$
$$\xrightarrow{d^{n}} \operatorname{W}^{[n+1]}(\mathcal{T}_{(n-1)}) \xrightarrow{i^{n,W}} \operatorname{W}^{[n+1]}(\mathcal{T}_{(n)}) \xrightarrow{q^{n,W}} \operatorname{W}^{[n+1]}(\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)}) \longrightarrow \dots,$$

we have, in particular, $\operatorname{Im}(q^n) \subset \ker(d^n) = (d^n)^{-1}(\{0\})$. Since $0 \in \operatorname{Im}(i^{n-1,W})$, we obtain $\operatorname{Im}(q^n) \subset (d^n)^{-1}(\{0\}) \subset (d^n)^{-1}(\operatorname{Im}(i^{n-1,W}))$, hence, $q^n(\ker(i^{n+1})) \subset (d^n)^{-1}(\operatorname{Im}(i^{n-1,W}))$ as stated.

 \square

This guarantees that the following, for this work central new notion of *tensor* triangular Chow–Witt groups is well-defined.

Definition 4.3.5. For \mathcal{T} as in Convention 4.3.1, we define its *n*-th tensor triangular Chow-Witt groups twisted by L for $n \in \mathbb{Z}$ as

$$\widetilde{\operatorname{CH}}^n_{\Delta}(\mathcal{T},L) := \widetilde{Z}^n_{\Delta}(\mathcal{T},L) / \widetilde{B}^n_{\Delta}(\mathcal{T},L),$$

where $\widetilde{Z}^n_{\Delta}(\mathcal{T}, L)$ and $\widetilde{B}^n_{\Delta}(\mathcal{T}, L)$ are defined in Definition 4.3.3.

Remark 4.3.6. If we are given a dimension function dim, we can always *shift* it by some $m \in \mathbb{Z}$ to obtain a dimension function dim'. Then, we have isomorphisms $\widetilde{CH}^{n}_{\Delta}(\mathcal{T}, \dim) \cong \widetilde{CH}^{n+m}_{\Delta}(\mathcal{T}, \dim')$ and $CH^{n}_{\Delta}(\mathcal{T}, \dim) \cong CH^{n+m}_{\Delta}(\mathcal{T}, \dim')$ for all n. However, since the constructions depend on the choice of the dimension function a priori, it is not clear if the change of tensor triangular Chow(–Witt) groups can be controlled when changing the dimension function in arbitrary ways.

Remark 4.3.7. If we replace GW with K_0 in the construction, it coincides with Balmer's tensor triangular Chow groups from [Bal13]. The assumption that all quotients of filtration components are idempotent complete is essential in this comparison.

We have given a construction of tensor triangular Chow–Witt groups in this section. However, the definition comes at the price of an agreement theorem that justifies the name, i.e. it shows that the definition agrees with the classical algebro-geometric case for regular X.

4.4 Agreement with Algebraic Geometry

Convention 4.4.1. Let X be a regular, separated scheme of finite type over a field k with $char(k) \neq 2$.

This section will justify the name of Definition 4.3.5. We will show that, for regular X, the definition for the tensor triangulated category $\mathcal{T} = D^{\text{perf}}(X)$ recovers the usual Definition 4.2.2 for X. Note that, in this case, \mathcal{T} satisfies all the conditions from Convention 4.3.1.

In Subsection 4.4.1, we reformulate the classical definition of oriented cocycles to a form that we can refine to the tensor triangular setting using the filtration by dimension of support. In Subsection 4.4.2, we do the same for classical oriented coboundaries. Finally, we merge these alternative definitions in Subsection 4.4.3 to show the agreements of tensor triangular and classical Chow–Witt groups and calculate an easy example by hand.

4.4.1 Reformulation of Classical Oriented Cocycles

As for Chow groups, we first aim to reformulate the definition of classical oriented cocycles in terms of maps coming from localization that we can directly generalize to the triangular setting.

Recall that P(X) is the exact category of locally free \mathcal{O}_X -modules of finite type over a noetherian scheme X of dimension d. Keep in mind that, if in addition X is regular, the bounded derived category of P(X) is equivalent to $D^b(X) :=$ $D^b(Coh(X))$ as a rigid tensor triangulated category.

The filtration by codimension of homological support from Definition 4.2.6

$$0 = D^{b}(P(X))^{(d+1)} \subset D^{b}(P(X))^{(d)} \subset ... \subset D^{b}(P(X))^{(0)} = D^{b}(P(X))$$

yields short exact sequences of triangulated categories with duality for $0 \le n \le d$

$$\mathbf{D}^{b}(P(X))^{(n+1)} \xrightarrow{I^{n}} \mathbf{D}^{b}(P(X))^{(n)} \xrightarrow{Q^{n}} \mathbf{D}^{b}(P(X))^{(n)} / \mathbf{D}^{b}(P(X))^{(n+1)}$$

We will need Walter's localization sequence from Theorem 3.3.3 for triangulated categories with duality

$$\operatorname{GW}^{[n]}(\mathrm{D}^{b}(P(X))^{(n+1)}) \xrightarrow{i^{n}} \operatorname{GW}^{[n]}(\mathrm{D}^{b}(P(X))^{(n)}) \xrightarrow{q^{n}} \operatorname{GW}^{[n]}(\mathrm{D}^{b}(P(X))^{(n)} / \operatorname{D}^{b}(P(X))^{(n+1)}) \xrightarrow{d^{n}} \operatorname{W}^{[n+1]}(\mathrm{D}^{b}(P(X))^{(n+1)}) \xrightarrow{i^{n,\mathrm{W}}} \operatorname{W}^{[n+1]}(\mathrm{D}^{b}(P(X))^{(n)}) \xrightarrow{q^{n,\mathrm{W}}} \operatorname{W}^{[n+1]}(\mathrm{D}^{b}(P(X))^{(n)} / \operatorname{D}^{b}(P(X))^{(n+1)}) \to \dots$$

to reformulate the classical oriented cocycles for a scheme X.

Theorem 4.4.2. Let X be a separated, regular scheme of finite type over a field k of characteristic $\neq 2$. With the above notation, there are isomorphisms for all n

$$\widetilde{Z}^{-n}(X,L) \cong (d^{-n})^{-1}(\operatorname{Im}(i^{-n+1,W}))_{:}$$

where we are in the algebro-geometric setting as in Definition 4.2.2.

Proof. By Remark 4.2.17, we know that
$$\widetilde{Z}^{-n}(X, L) := \ker(\partial^{-n}) \cong \ker(\widetilde{\partial}^{-n})$$
, where
 $\widetilde{\partial}^{-n} := q^{-n+1,W} \circ d^{-n} : \operatorname{GW}^{[-n]}(\operatorname{D}^{b}(P(X))^{(-n)}/\operatorname{D}^{b}(P(X))^{(-n+1)}, \operatorname{Hom}_{\mathcal{O}_{X}}(-, L))$
 $\to \operatorname{W}^{[-n+1]}(\operatorname{D}^{b}(P(X))^{(-n+1)}, \operatorname{Hom}_{\mathcal{O}_{X}}(-, L))$
 $\to \operatorname{W}^{[-n+1]}(\operatorname{D}^{b}(P(X))^{(-n+1)}/\operatorname{D}^{b}(P(X))^{(-n+2)}, \operatorname{Hom}_{\mathcal{O}_{X}}(-, L)).$

Since d^{-n} is exactly the boundary map from the localization sequence Theorem 3.3.3 for (Grothendieck–)Witt groups of triangulated categories with duality, we can use the exactness of that sequence to write $\widetilde{Z}^{-n}(X, L)$ as

$$\widetilde{Z}^{-n}(X,L) := \ker(\partial^{-n}) \cong \ker(\widetilde{\partial}^{-n}) = \ker(q^{-n+1,W} \circ d^{-n})$$
$$= (d^{-n})^{-1}(\ker(q^{-n+1,W})) = (d^{-n})^{-1}(\operatorname{Im}(i^{-n+1,W})).$$

After having found an alternative description of algebraic oriented cocycles in terms of maps coming from the localization sequence for (Grothendieck–)Witt groups, we do the same for oriented coboundaries in the next subsection.

4.4.2 Reformulation of Classical Oriented Coboundaries

Reformulating classical oriented coboundaries in terms of maps coming from localization is trickier, as mentioned in Subsection 4.3.1. We recall from Example 3.2.11 that

$$(Ch^{b}(P(X)), quis, Hom_{\mathcal{O}_{X}}(-, L))$$

is a complicial exact category with weak equivalences and duality by fixing the quasi-isomorphisms quis. We have subcategories

$$\mathrm{Ch}^{b}(P(X))^{(n)} := \{P_{\bullet} \in \mathrm{Ch}^{b}(P(X)) \mid \operatorname{codim}(\mathrm{supph}(P_{\bullet})) \ge n\}$$

and denote by quis^{n'} the morphisms in $\operatorname{Ch}^{b}(P(X))^{(n)}$ with cone in $\operatorname{Ch}^{b}(P(X))^{(n')}$ for $n' \geq n$, which again yield complicial exact categories with weak equivalences and duality

$$(\operatorname{Ch}^{b}(P(X))^{(n)}, \operatorname{quis}^{n'}, \operatorname{Hom}_{\mathcal{O}_{X}}(-, L)).$$

Recall that we denote the triangulated categories with duality obtained by inverting the quasi-isomorphisms $quis^{n'}$ or quis by $Ch^b(P(X))^{(n)}[(quis^{n'})]^{-1}$ and $Ch^b(P(X))^{(n)}[(quis)]^{-1}$, respectively.

By Theorem 3.3.4, we obtain for $n \in \mathbb{Z}$ a localization sequence for Grothendieck– Witt groups of complicial exact categories with weak equivalences and duality

$$\dots \to \operatorname{GW}_{m}^{[n]}(\operatorname{Ch}^{b}(P(X))^{(n+1)}, \operatorname{quis}) \xrightarrow{i_{\operatorname{Ch}}^{n,m}} \operatorname{GW}_{m}^{[n]}(\operatorname{Ch}^{b}(P(X))^{(n)}, \operatorname{quis}) \xrightarrow{q_{\operatorname{Ch}}^{n,m}} \operatorname{GW}_{m}^{[n]}(\operatorname{Ch}^{b}(P(X))^{(n)}, \operatorname{quis}^{n+1}) \xrightarrow{d_{\operatorname{Ch}}^{n,m}} \operatorname{GW}_{m-1}^{[n]}(\operatorname{Ch}^{b}(P(X))^{(n)}, \operatorname{quis}) \xrightarrow{i_{\operatorname{Ch}}^{n,m-1}} \operatorname{GW}_{m-1}^{[n]}(\operatorname{Ch}^{b}(P(X))^{(n)}, \operatorname{quis}) \to \dots$$

With these tools, we can prove a hermitian analog of [Qui73, Theorem 5.14], which has already been proven in [Hor08, Theorem 1.7] and [FS09, Theorem 33]. An important part of its proof will be the comparison of the localization sequences

in the different settings introduced in Section 3.3, and the comparison between localization sequence of the triangulated categories $\operatorname{Ch}^{b}(P(X))^{(n)}[(\operatorname{quis}^{n+1})^{-1}]$ and $\operatorname{D}^{b}(P(X))^{(n)}/\operatorname{D}^{b}(P(X))^{(n+1)}$. Recall that the notation for the latter is

$$\operatorname{GW}^{[n]}(\operatorname{D}^{b}(P(X))^{(n+1)}) \xrightarrow{i^{n}} \operatorname{GW}^{[n]}(\operatorname{D}^{b}(P(X))^{(n)}) \xrightarrow{q^{n}} \operatorname{GW}^{[n]}(\operatorname{D}^{b}(P(X))^{(n)} / \operatorname{D}^{b}(P(X))^{(n+1)}) \xrightarrow{d^{n}} \operatorname{W}^{[n+1]}(\operatorname{D}^{b}(P(X))^{(n+1)}) \to \dots$$

Now, we can formulate:

Theorem 4.4.3. Let X be a separated, regular scheme of finite type over a field k of characteristic $\neq 2$. With the notation from above, there are isomorphisms for all n

$$\widetilde{B}^n(X,L) \cong q^n(\ker(i^{n-1})),$$

i.e. the image of

$$\partial^{n-1} : C^{n-1}(X, G^n, L) \to C^n(X, G^n, L)$$
$$\xrightarrow{\cong} \operatorname{GW}^{[n]}(\operatorname{D}^b(P(X))^{(n)} / \operatorname{D}^b(P(X))^{(n+1)}, R \operatorname{Hom}_{\mathcal{O}_X}(-, L))$$

of Definition 4.2.2 equals $q^n(\ker(i^{n-1}))$.

Here, the isomorphism to $\widetilde{\mathrm{GW}}^{[n]}(\widetilde{\mathrm{D}^{b}}(P(X))^{(n)}/\mathrm{D}^{b}(P(X))^{(n+1)}, R \operatorname{Hom}_{\mathcal{O}_{X}}(-,L))$ is the one from Remark 4.2.17.

Proof of Theorem 4.4.3. We do the proof in three steps. First, we reformulate the oriented coboundaries in terms of maps coming from the localization sequence of exact categories with weak equivalences and duality from Theorem 3.3.4. Second, we move this description to the triangular setting after inverting the respective quasi-isomorphisms of the considered exact categories with weak equivalences and duality. At last, we compare the localization sequences for triangulated categories with duality as in Theorem 3.3.3 for the categories $\operatorname{Ch}^{b}(P(X))^{(n)}[(\operatorname{quis}^{n+1})^{-1}]$ and $\operatorname{D}^{b}(P(X))^{(n)}/\operatorname{D}^{b}(P(X))^{(n+1)}$, i.e. for the case of first inverting the quasi-isomorphisms and then filtrating the category, and vice versa.

By [FS09, Theorem 33], the *n*-th Chow–Witt group can also be defined as the cohomology of the following sequence of (Grothendieck–)Witt groups of complicial exact categories with weak equivalences and duality, all dualities again being derived from $\operatorname{Hom}_{\mathcal{O}_X}(-, L)$,

$$\begin{array}{c} \operatorname{GW}_{1}^{[n]}(\operatorname{Ch}^{b}(P(X))^{(n-1)},\operatorname{quis}^{n}) \\ \xrightarrow{\widetilde{\partial}_{\operatorname{Ch}}^{1,n}} & \operatorname{GW}^{[n]}(\operatorname{Ch}^{b}(P(X))^{(n)},\operatorname{quis}^{n+1}) \\ \xrightarrow{\widetilde{\partial}_{\operatorname{Ch}}^{0,n}} & \operatorname{W}^{[n+1]}(\operatorname{Ch}^{b}(P(X))^{n+1},\operatorname{quis}^{n+2}). \end{array}$$

Hence, we have an isomorphism

$$H^{n}(C(X, G^{n}, L)) = \widetilde{Z}^{n}(X, L) / \widetilde{B}^{n}(X, L) \cong \ker(\widetilde{\partial}_{\mathrm{Ch}}^{0, n}) / \operatorname{Im}(\widetilde{\partial}_{\mathrm{Ch}}^{1, n}),$$

where $C(X, G^n, L)$ is the complex from Definition 4.2.2. The sequence arises from the Gersten-Grothendieck-Witt spectral sequence from [FS09, Definition 26] and, thus, it follows by construction that, in particular, $\operatorname{Im}(\widetilde{\partial}_{\mathrm{Ch}}^{1,n}) \cong \widetilde{B}^n(X,L)$. Now, Theorem 3.3.4 yields the long exact localization sequence for Grothendieck–

Witt groups of exact categories with weak equivalences and duality

$$\dots \to \operatorname{GW}_{m}^{[n]}(\operatorname{Ch}^{b}(P(X))^{(n+1)}, \operatorname{quis}) \xrightarrow{i_{\operatorname{Ch}}^{n,m}} \operatorname{GW}_{m}^{[n]}(\operatorname{Ch}^{b}(P(X))^{(n)}, \operatorname{quis}) \\ \xrightarrow{q_{\operatorname{Ch}}^{n,m}} \to \operatorname{GW}_{m}^{[n]}(\operatorname{Ch}^{b}(P(X))^{(n)}, \operatorname{quis}^{n+1}) \\ \dots \xrightarrow{d_{\operatorname{Ch}}^{n,m}} \operatorname{GW}_{m-1}^{[n]}(\operatorname{Ch}^{b}(P(X))^{(n+1)}, \operatorname{quis}) \xrightarrow{i_{\operatorname{Ch}}^{n,m-1}} \operatorname{GW}_{m-1}^{[n]}(\operatorname{Ch}^{b}(P(X))^{(n)}, \operatorname{quis}) \to \dots,$$

and we know that $\widetilde{\partial}_{\mathrm{Ch}}^{1,n}$ is the composition

$$\begin{split} \widetilde{\partial}_{\mathrm{Ch}}^{1,n} &:= q_{\mathrm{Ch}}^{n,0} \circ d_{\mathrm{Ch}}^{n-1,1} : \mathrm{GW}_1^{[n]}(\mathrm{Ch}^b(P(X))^{(n-1)}, \mathrm{quis}^n) \\ &\to \mathrm{GW}_0^{[n]}(\mathrm{Ch}^b(P(X))^{(n)}, \mathrm{quis}) \\ &\to \mathrm{GW}_0^{[n]}(\mathrm{Ch}^b(P(X))^{(n)}, \mathrm{quis}^{n+1}). \end{split}$$

We can use the exactness of the localization sequence from Theorem 3.3.4 to conclude

$$\operatorname{Im}(\widetilde{\partial}_{\mathrm{Ch}}^{1,n}) = \operatorname{Im}(q_{\mathrm{Ch}}^{n,0} \circ d_{\mathrm{Ch}}^{n-1,1}) = q_{\mathrm{Ch}}^{n,0}(\operatorname{Im}(d_{\mathrm{Ch}}^{n-1,1})) = q_{\mathrm{Ch}}^{n,0}(\ker(i_{\mathrm{Ch}}^{n-1,0})).$$

After having rewritten the algebro-geometric coboundaries in terms of maps in the localization sequence of exact categories with weak equivalences and duality, we will now pass to the triangular setting.

For this, we show that the horizontal arrows in the diagram

$$\begin{split} \mathrm{GW}_{0}^{[n]}(\mathrm{Ch}^{b}(P(X))^{(n-1)},\mathrm{quis}) & \longrightarrow \mathrm{GW}^{[n]}(\mathrm{Ch}^{b}(P(X))^{(n-1)}[(\mathrm{quis})^{-1}]) \\ & \uparrow^{i_{\mathrm{Ch}}^{n-1,0}} & \uparrow \\ \mathrm{GW}_{0}^{[n]}(\mathrm{Ch}^{b}(P(X))^{(n)},\mathrm{quis}) & \longrightarrow \mathrm{GW}^{[n]}(\mathrm{Ch}^{b}(P(X))^{(n)}[(\mathrm{quis})^{-1}]) \\ & \downarrow^{q_{\mathrm{Ch}}^{n,0}} & \downarrow \\ \mathrm{GW}_{0}^{[n]}(\mathrm{Ch}^{b}(P(X)^{(n)}),\mathrm{quis}^{n+1}) & \longrightarrow \mathrm{GW}^{[n]}(\mathrm{Ch}^{b}(P(X))^{(n)}[(\mathrm{quis}^{n+1})^{-1}]) \end{split}$$

are isomorphisms. The left column treats Grothendieck-Witt groups of exact categories with weak equivalences and duality, whereas the right column contains Grothendieck–Witt groups of triangulated categories with duality.

The isomorphisms follow from Proposition 3.2.14 and the fact that, by [FS09, Theorem 8], the part of the localization sequence from above (Theorem 3.3.4) for complicial exact categories with weak equivalences and duality starting in GW_0 coincides with the tensor triangular localization sequence Theorem 3.3.2 for the short exact sequence

$$\operatorname{Ch}^{b}(P(X))^{(n-1)}[(\operatorname{quis})^{-1}] \to \operatorname{Ch}^{b}(P(X))^{(n)}[(\operatorname{quis})^{-1}] \to \operatorname{Ch}^{b}(P(X))^{(n)}[(\operatorname{quis}^{n+1})^{-1}].$$

Lastly, we want to show that the horizontal morphisms in the diagram

are isomorphism.

Both columns involve Grothendieck–Witt groups of triangulated categories with duality. In the left column, we first filtrate the exact category with weak equivalences and duality $\operatorname{Ch}^{b}(P(X))$ by codimension of homological support and then invert the given quasi-isomorphisms; in the column on the right, we first invert all quasi-isomorphisms in $\operatorname{Ch}^{b}(P(X))$ before filtrating.

The underlying functors of f and g are categorical equivalences since inverting quasi-isomorphisms commutes with restricting objectwise to the subcategory $(-)^{(k)}$ for $k \in \{n-1, n, n+1\}$. Moreover, we note that the underlying triangulated categories in the domain and codomain of h are equivalent by universality of the Verdier quotient functor (compare e.g. [Nee05, Proposition 2.1.24] or [Wei95, Exercise 10.3.2] for abelian categories). The isomorphisms on the level of GW now follow from Proposition 3.2.14.

Hence, we obtain a commutative diagram with the horizontal maps being isomorphisms

$$\begin{array}{c} \operatorname{GW}_{0}^{[n]}(\operatorname{Ch}^{b}(P(X))^{(n-1)}, \operatorname{quis}) & \xrightarrow{\cong} & \operatorname{GW}^{[n]}(\operatorname{D}^{b}(P(X))^{(n-1)}) \\ & \uparrow_{i_{\operatorname{Ch}}}^{i_{n-1}, 0} & i^{n-1} \\ & \operatorname{GW}_{0}^{[n]}(\operatorname{Ch}^{b}(P(X))^{(n)}, \operatorname{quis}) & \xrightarrow{\cong} & \operatorname{GW}^{[n]}(\operatorname{D}^{b}(P(X))^{(n)}) \\ & \downarrow_{q_{\operatorname{Ch}}}^{n, 0} & q^{n} \\ & \operatorname{GW}_{0}^{[n]}(\operatorname{Ch}^{b}(P(X)^{(n)}), \operatorname{quis}^{n+1}) & \xrightarrow{\cong} & \operatorname{GW}^{[n]}(\operatorname{D}^{b}(P(X)^{(n)}) / \operatorname{D}^{b}(P(X)^{(n+1)})) \end{array}$$

Eventually, this yields $\widetilde{B}^n(X,L) \cong q_{\mathrm{Ch}}^{n,0}(\ker(i_{\mathrm{Ch}}^{n-1,0})) \cong q^n(\ker(i^{n-1}))$ as claimed.

Remark 4.4.4. Note that, until now, we have only considered filtrations by *codimension of homological support* as in Definition 4.2.6, which is why there has not appeared a change of sign yet as announced before. We have not yet considered tensor triangular oriented cocycles/coboundaries; their definition requires the filtration of tensor triangulated categories by *codimension of support* as in Definition 2.3.4. Passing between these filtrations will involve change of signs. In this and the last section, we have rewritten the algebro-geometric oriented cocycles $\widetilde{Z}^n(X, L)$ and coboundaries $\widetilde{B}^n(X, L)$ in terms of maps coming from the localization sequence

$$\operatorname{GW}^{[n]}(\operatorname{D}^{b}(P(X))^{(n+1)}) \xrightarrow{i^{n}} \operatorname{GW}^{[n]}(\operatorname{D}^{b}(P(X))^{(n)})$$
$$\xrightarrow{q^{n}} \operatorname{GW}^{[n]}(\operatorname{D}^{b}(P(X))^{(n)}/\operatorname{D}^{b}(P(X))^{(n+1)}) \xrightarrow{d^{n}} \operatorname{W}^{[n+1]}(\operatorname{D}^{b}(P(X))^{(n+1)}) \xrightarrow{i^{n,W}} \dots$$

for triangulated categories with duality. Hence, we have lifted the definition of Chow–Witt groups already to a triangular setting. However, to show the agreement between the definition of tensor triangular Chow–Witt groups and the classical Chow–Witt groups, we need to check whether these rewritten forms coincide with Definition 4.3.5 in the tensor triangular setting. We will do so in the following section.

4.4.3 Agreement of Chow–Witt Groups

Above, we have brought oriented cocycles and coboundaries to a form that we can generalize directly. In this sense, Theorem 4.4.3 and Theorem 4.4.2 pave the path for the following central agreement theorem this section aims to prove. It is a refinement of [Kl16a, Theorem 3.2.6].

Theorem 4.4.5. Let X be a separated, regular scheme of finite type over a field k with char(k) $\neq 2$, L a line bundle over X, and $D^{\text{perf}}(X)$ be equipped with the dimension function $-\operatorname{codim}_{Krull}$. Then, there are isomorphisms for all $n \in \mathbb{Z}$

$$\widetilde{Z}^n_{\Delta}(\mathcal{D}^{\mathrm{perf}}(X), \hom(-, L)) \cong \widetilde{Z}^{-n}(X, L),$$

$$\widetilde{B}^n_{\Delta}(\mathcal{D}^{\mathrm{perf}}(X), \hom(-, L)) \cong \widetilde{B}^{-n}(X, L)$$

and

$$\widetilde{\operatorname{CH}}^{n}_{\Delta}(\operatorname{D}^{\operatorname{perf}}(X), \operatorname{hom}(-, L)) \cong \widetilde{\operatorname{CH}}^{-n}(X, L).$$

Proof. We start in the algebro-geometric setting as in Definition 4.2.2. Then, Theorem 4.4.2 and Theorem 4.4.3 tell us that we have isomorphisms

$$\widetilde{CH}^{-n}(X) \cong (d^{-n})^{-1}(\mathrm{Im}(i^{-n+1,\mathrm{W}}))/q^{-n}(\ker(i^{-n-1})),$$

where the maps are the ones from the localization sequence of triangulated categories with duality

$$\operatorname{GW}^{[n]}(\operatorname{D}^{b}(P(X))^{(n+1)}) \xrightarrow{i^{n}} \operatorname{GW}^{[n]}(\operatorname{D}^{b}(P(X))^{(n)})$$

$$\xrightarrow{q^{n}} \operatorname{GW}^{[n]}(\operatorname{D}^{b}(P(X))^{(n)} / \operatorname{D}^{b}(P(X))^{(n+1)}) \xrightarrow{d^{n}} \operatorname{W}^{[n+1]}(\operatorname{D}^{b}(P(X))^{(n+1)}) \xrightarrow{i^{n,W}} \dots$$

Now, let $\mathcal{T} := D^{\text{perf}}(X)$ be the rigid tensor triangulated category from Example 2.2.6 (iii) equipped with $- \operatorname{codim}_{\mathrm{Krull}}$, and L a fixed \otimes -invertible object; note

that here hom := R Hom. We know that, by our assumptions, there is a categorical equivalence $D^{perf}(X) \simeq D^b(P(X))$.

We recall the definition of the filtration components in the tensor-triangular setting as in Section 2.3

$$D^{\text{perf}}(X)_{(n)} := \{ P \in D^{\text{perf}}(X) \mid - \text{codim}_{\text{Krull}}(\text{supp}(P)) \le n \}$$

with

$$\operatorname{supp}(P) := \{I \in \operatorname{Spc}(\operatorname{D^{perf}}(X)) \mid P \notin I\} \subset \operatorname{Spc}(\operatorname{D^{perf}}(X))$$

and the filtration components in the algebro-geometric setting as in Definition 4.2.6

$$D^{b}(P(X))^{(-n)} := \{ P \in D^{b}(P(X)) \mid \operatorname{codim}(\operatorname{supph}(P)) \ge -n \}$$
$$= \{ P \in D^{b}(P(X)) \mid -\operatorname{codim}(\operatorname{supph}(P)) \le n \}$$

with

$$\operatorname{supph}(P) := \bigcup_{i} \operatorname{supp}(H_i(P)) \subset X.$$

We know by [Bal05b, Corollary 5.6] that, under the isomorphism $\operatorname{Spec}(D^{\operatorname{perf}}(X)) \cong X$ from Example 2.1.20, $\operatorname{supph}(A) \subset X$ and $\operatorname{supp}(A) \subset \operatorname{Spc}(D^{\operatorname{perf}}(X))$ coincide for $A \in D^{\operatorname{perf}}(X)$. Thus, using $D^{\operatorname{perf}}(X) \simeq D^b(P(X))$, we obtain equivalences of triangulated categories with duality

$$(\mathbf{D}^{\mathrm{perf}}(X)_{(n)}, \hom(-, L)) \simeq (\mathbf{D}^{b}(P(X))^{(-n)}, \operatorname{Hom}(-, L))$$

and

$$(\mathrm{D}^{\mathrm{perf}}(X)_{(n)} / \mathrm{D}^{\mathrm{perf}}(X)_{(n-1)}, \mathrm{hom}(-, L))$$

 $\simeq (\mathrm{D}^{b}(P(X))^{(-n)} / \mathrm{D}^{b}(P(X))^{(-n+1)}, \mathrm{Hom}(-, L)).$

It follows that the horizontal arrows in the commutative diagrams (where the dualities are omitted from the notation)

$$\begin{array}{c} \operatorname{GW}^{[n]}(\mathcal{T}_{(n+1)}) & \longrightarrow \operatorname{GW}^{[n]}(\operatorname{D}^{b}(P(X))^{(-n-1)}) \\ & i_{\mathcal{T}}^{n+1} \uparrow & i^{-n-1} \uparrow \\ & \operatorname{GW}^{[n]}(\mathcal{T}_{(n)}) & \longrightarrow \operatorname{GW}^{[n]}(\operatorname{D}^{b}(P(X))^{(-n)}) \\ & \downarrow^{q_{\mathcal{T}}^{n}} & \downarrow^{q^{-n}} \\ & \operatorname{GW}^{[n]}(\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)}) & \longrightarrow \operatorname{GW}^{[n]}(\operatorname{D}^{b}(P(X))^{(-n)}/\operatorname{D}^{b}(P(X))^{(-n+1)}) \end{array}$$

and

$$\begin{split} \operatorname{GW}^{[n]}(\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)}) & \longrightarrow \operatorname{GW}^{[n]}(\operatorname{D}^{b}(P(X))^{(-n)}/\operatorname{D}^{b}(P(X))^{(-n+1)}) \\ & \downarrow^{d^{n}} & \downarrow^{d^{-n}} \\ \operatorname{W}^{[n+1]}(\mathcal{T}_{(n-1)}) & \longrightarrow \operatorname{W}^{[n+1]}(\operatorname{D}^{b}(P(X))^{(-n+1)}) \\ & \uparrow^{i^{n-1,W}}_{\mathcal{T}} & \uparrow^{i^{-n+1,W}} \\ \operatorname{W}^{[n+1]}(\mathcal{T}_{(n-2)}) & \longrightarrow \operatorname{W}^{[n+1]}(\operatorname{D}^{b}(P(X))^{(-n+2)}) \end{split}$$

coming from the above equivalences are isomorphisms. As a consequence, we have

$$\widetilde{B}^{-n}(X,L) := \operatorname{Im}(\partial^{-n+1}) \cong q^{-n}(\ker(i^{-n-1}))$$
$$\cong q_{\mathcal{T}}^n(\ker(i_{\mathcal{T}}^{n+1})) =: \widetilde{B}^n_{\Delta}(\operatorname{D}^{\operatorname{perf}}(X), \hom(-, L))$$

and, moreover,

$$\widetilde{Z}^{-n}(X,L) := \ker(\partial^{-n}) \cong \ker(\widetilde{\partial}^{-n}) \cong (d^{-n})^{-1}(\operatorname{Im}(i^{-n+1,W}))$$
$$\cong (d^{n}_{\mathcal{T}})^{-1}(\operatorname{Im}(i^{n-1,W}_{\mathcal{T}})) =: \widetilde{Z}^{n}_{\Delta}(\operatorname{D}^{\operatorname{perf}}(X), \operatorname{hom}(-,L)).$$

Remark 4.4.6. Fasel and Srinivas present an alternative definition of Chow–Witt groups of a scheme in terms of Grothendieck–Witt groups of triangulated categories with duality. One could also have started the attempt of generalizing Chow–Witt groups of a scheme at this point.

Consider the exact sequences of triangulated categories with duality

$$D^{b}(P(X))^{(n+1)}[(quis^{n+2})^{-1}] \to D^{b}(P(X))^{(n)}[(quis^{n+2})^{-1}] \\ \to D^{b}(P(X))^{(n)}[(quis^{n+1})^{-1}]$$

and

$$\begin{split} \mathbf{D}^{b}(P(X))^{(n)}[(\mathbf{quis}^{n+1})^{-1}] &\to \mathbf{D}^{b}(P(X))^{(n-1)}[(\mathbf{quis}^{n+1})^{-1}] \\ &\to \mathbf{D}^{b}(P(X))^{(n-1)}[(\mathbf{quis}^{n})^{-1}] \end{split}$$

They yield exact sequences by localization

$$\operatorname{GW}^{[n]}(\mathrm{D}^{b}(P(X))^{(n)}[(\operatorname{quis}^{n+2})^{-1}]) \xrightarrow{\alpha} \operatorname{GW}^{[n]}(\mathrm{D}^{b}(P(X))^{(n)}[(\operatorname{quis}^{n+1})^{-1}]) \xrightarrow{q^{n-1,\mathrm{W}} \circ d^{n}} \operatorname{W}^{[n+1]}(\mathrm{D}^{b}(P(X))^{(n+1)}[(\operatorname{quis}^{n+2})^{-1}])$$

and

$$GW_{1}^{[n]}(D^{b}(P(X))^{(n-1)}[(quis^{n})^{-1}]) \xrightarrow{q^{n} \circ d^{n-1}} GW^{[n]}(D^{b}(P(X))^{(n)}[(quis^{n+1})^{-1}])$$
$$\xrightarrow{\beta} GW^{[n]}(D^{b}(P(X))^{(n-1)}[(quis^{n+1})^{-1}]),$$

where $\operatorname{GW}_{1}^{[n]}(\operatorname{D}^{b}(P(X))^{(n-1)}[(\operatorname{quis}^{n})^{-1}])$ is defined as the higher Grothendieck– Witt group $\operatorname{GW}_{1}^{[n]}(\operatorname{Ch}^{b}(P(X))^{(n-1)},\operatorname{quis}^{n})$ of the underlying complicial exact category with weak equivalences and duality, compare Remark 3.2.15. The localization sequences of triangulated categories with duality from Theorem 3.3.3 and of complicial exact categories with weak equivalences and duality from Theorem 3.3.4 are combined here, using that they coincide from Grothendieck–Witt groups of degree ≤ 0 on by Theorem 3.3.4.

In [FS09, Theorem 34], Fasel and Srinivas prove that

$$\widetilde{\operatorname{CH}}^{n}(X,L) = \operatorname{Im}(\alpha) / \ker(\beta)$$

for a regular $\mathbb{Z}[\frac{1}{2}]$ -linear scheme X and a line bundle L. Using the tools of this proof, one can show Theorem 4.4.2 and Theorem 4.4.3 in the following way.

By exactness and [FS09, Theorem 34], it follows that

$$\widetilde{Z}^{n}(X,L) = \operatorname{Im}(\alpha) = \ker(q^{n-1,W} \circ d^{n}) = (d^{n})^{-1}(\ker(q^{n-1,W}))$$
$$= (d^{n})^{-1}(\operatorname{Im}(i^{n-1,W})).$$

and

$$\widetilde{B}^n(X,L) = \ker(\beta) = \operatorname{Im}(q^n \circ d^{n-1}) = q(\operatorname{Im}(d^{n-1})) = q^n(\ker(i^{n-1})).$$

One of the main reasons why we did not go down the route of [FS09, Theorem 34] was that we wanted to distinguish more clearly the localization sequences in the setting of triangulated categories with duality and of complicial exact categories with weak equivalences and duality. On the other hand, we aimed to stay close to the definition of tensor triangular Chow groups introduced in [Bal13] and the corresponding agreement theorem [Kl16a, Theorem 3.2.6].

Remark 4.4.7. In Theorem 4.4.5, we assume X to be separated, regular, and of finite type over a field k with characteristic $\neq 2$. These assumptions originate in the definition of Chow–Witt groups we use, compare [Fas20, Section 2.1]. If one uses a more general definition of Chow–Witt groups for regular $\mathbb{Z}[\frac{1}{2}]$ -linear schemes as for example in [FS09, Definition 32], the proof of Theorem 4.4.5 also holds true for this slightly more general case (so do the proofs from Remark 4.4.6).

To get an intuitive understanding of why Theorem 4.4.5 is true, we compare easy Chow–Witt groups "by hand".

Example 4.4.8. Let k be a field of characteristic $\neq 2$. We know that

$$\widetilde{\operatorname{CH}}^{n}(\operatorname{Spec}(k)) = \begin{cases} \operatorname{GW}(k) & \text{if } n = 0\\ 0 & \text{else.} \end{cases}$$

We want to show the same for $\widetilde{CH}^n_{\Delta}(D^{\text{perf}}(\text{Spec}(k)))$.

Consider $\mathcal{T} := D^{\text{perf}}(\text{Spec}(k))$ equipped with the dimension function dim := $- \operatorname{codim}_{\text{Krull}}$, i.e. the category of perfect complexes of $\mathcal{O}_{\text{Spec}(k)}$ -modules. An $\mathcal{O}_{\text{Spec}(k)}$ -module is a sheaf that sends open sets in $\operatorname{Spec}(k)$ to $\mathcal{O}_{\text{Spec}(k)} = k$ -modules. Since

Spec(k) contains only one point, $\mathcal{O}_{\text{Spec}(k)}$ -modules are completely determined by the k-vector space the point is sent to, i.e. we can identify $\mathcal{O}_{\text{Spec}(k)}$ -modules with k-vector spaces. It follows that $D^{\text{perf}}(\text{Spec}(k))$ is given by perfect complexes of k-vector spaces.

We now want to understand what $\mathcal{T}_{(n)}$ looks like for $n \in \mathbb{Z}$. For this, we check that for all objects $A \neq 0$ in \mathcal{T} we have $\operatorname{supp}(A) = \{0\}$, hence $\operatorname{dim}(\operatorname{supp}(A)) = 0$, i.e. $-1 < \operatorname{dim}(\operatorname{supp}(A)) \leq 0$. For $0 \in \mathcal{T}$, we have $\operatorname{supp}(0) = \emptyset$ by [Bal07, Proposition 1.3 (i)]. Hence, we obtain

$$T_{(n)} = \begin{cases} \mathcal{T} & \text{for } n \ge 0\\ 0 & \text{for } n \le -1 \end{cases}$$

and for the quotients

$$\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)} = \begin{cases} 0 & \text{for } n \ge 1\\ \mathcal{T} & \text{for } n = 0\\ 0 & \text{for } n \le -1. \end{cases}$$

For n = 0, the defining diagram Diagram 4.3 becomes

and thus

$$W^{[1]}(0) \xrightarrow{0_{i}} W^{[1]}(0) \xrightarrow{GW^{[0]}(\mathcal{T}) \xrightarrow{id}} GW^{[1]}(\mathcal{T}).$$

$$(\mathcal{T}) \xrightarrow{0_{d}} \downarrow^{id}$$

$$(\mathcal{T})$$

It follows that

$$\widetilde{\operatorname{CH}}^{0}_{\Delta}(\operatorname{D}^{\operatorname{perf}}(\operatorname{Spec}(k))) = {}^{0_{d}^{-1}(\operatorname{Im}(0_{i}))}_{d}(\operatorname{ker}(\operatorname{id})) = \operatorname{GW}(\mathcal{T})_{0} = \operatorname{GW}(\mathcal{T}).$$

Using Proposition 3.2.7, we obtain

$$\widetilde{\operatorname{CH}}^{0}_{\Delta}(\operatorname{D}^{\operatorname{perf}}(\operatorname{Spec}(k))) = \operatorname{GW}(\mathcal{T}) \cong \operatorname{GW}(k)$$

The cases $n \neq 0$ can be seen directly.

This subsection has proven that the construction of tensor triangular Chow–Witt groups we gave in Chapter 4 agrees with the classical definition of Chow–Witt groups of a scheme X for the derived category of perfect complexes $D^{perf}(X)$, X being regular. Moreover, we have seen an easy example by hand. The following section discusses functorial properties of $\widetilde{CH}^n_{\Delta}$.

4.5 Functoriality

We are in the situation of Convention 4.3.1. After having seen the agreement theorem between classical and tensor triangular Chow–Witt groups, we now show the functoriality of the construction from Chapter 4. We will see that we obtain, in particular, flat pullbacks and proper pushforwards.

First, we need to specify what kind of functors we allow.

Definition 4.5.1. Let \mathcal{T}, \mathcal{L} be tensor triangulated categories equipped with dimension functions, and let $F : \mathcal{T} \to \mathcal{L}$ be a triangulated functor. We say that F is of relative dimension m if there exists a smallest $m \in \mathbb{N}$ such that $F(\mathcal{T}_{(n)}) \subset \mathcal{L}_{(n+m)}$ for all n.

Remark 4.5.2. We do not require F to be a tensor functor following the idea of [Kl16a, Remark 4.1.2]. There are various examples of functors of relative dimension that are neither tensor functors nor duality-preserving, e.g. $a \otimes -$ when $\dim(\operatorname{supp}(a)) \neq \pm \infty$, see [Kl16a, Proposition 4.1.6].

Remark 4.5.3. Suppose F is a functor of relative dimension m. By shifting the dimension function, we can always obtain m = 0. Recall that shifting the dimension function on a rigid tensor triangulated category \mathcal{T} by $m' \in \mathbb{Z}$ yields isomorphisms $\widetilde{\operatorname{CH}}^n_{\Delta}(\mathcal{T}) \cong \widetilde{\operatorname{CH}}^{n+m'}_{\Delta}(\mathcal{T})$ for all n, compare Remark 4.3.6.

We can generalize [Kl16a, Theorem 4.1.3] in the following way.

Theorem 4.5.4. Let \mathcal{T}, \mathcal{L} be rigid tensor triangulated categories equipped with dimension functions and let K and L be \otimes -invertible objects in \mathcal{T} and \mathcal{L} , respectively. Moreover, let $F : \mathcal{T} \to \mathcal{L}$ be a functor of relative dimension 0 that is duality-preserving when considering \mathcal{T} and \mathcal{L} triangulated categories with duality $(\mathcal{T}, \hom(-, K))$ and $(\mathcal{L}, \hom(-, L))$ as in Example 2.2.6. Then, F induces group homomorphisms for all $n \in \mathbb{Z}$

$$\widetilde{z}^n : \widetilde{C}^n_{\Delta}(\mathcal{T}, K) \to \widetilde{Z}^n_{\Delta}(\mathcal{L}, L) \text{ and}$$
 $\widetilde{c}^n : \widetilde{CH}^n_{\Delta}(\mathcal{T}, K) \to \widetilde{CH}^n_{\Delta}(\mathcal{L}, L).$

Proof. We will omit the dualities from the notation for better readability. The functor F restricts to functors on the filtration components $F_i : \mathcal{T}_{(i)} \to \mathcal{L}_{(i)}$ for all i. Consider the following commutative diagram, which is the underlying right part of the defining Diagram 4.3,



Here, \overline{F}_n exists since $F(\mathcal{T}_{(n-1)}) \subset \mathcal{L}_{(n-1)}$ by definition, hence, F induces a morphism on quotients.

Since \overline{F} is duality-preserving, so are F_i and \overline{F}_i for all i. Consequently, we obtain the commutative diagram



by applying $GW^{[n]}$. For the left part of the defining Diagram 4.3, note that the diagram

commutes, and, as above, we can apply $W^{[n+1]}$ and obtain a commutative diagram

It is left to show that the diagram

also commutes, but this is Lemma 4.5.5 below, and, thus, we can glue together Diagram 4.4 and Diagram 4.5. Hence, we have a homomorphism

$$\bar{f}_n : \mathrm{GW}^{[n]}(\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)}) \to \mathrm{GW}^{[n]}(\mathcal{L}_{(n)}/\mathcal{L}_{(n-1)})$$

which restricts to a homomorphism

$$\bar{f}_n: \widetilde{Z}^n_\Delta(\mathcal{T}) \to \mathrm{GW}^{[n]}(\mathcal{L}_{(n)}/\mathcal{L}_{(n-1)})$$

since $\widetilde{Z}^n_{\Delta}(\mathcal{T})$ is a subgroup of $\mathrm{GW}^{[n]}(\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)})$ by Proposition 4.3.4. Now, the glued diagram yields

$$\bar{f}_n(\widetilde{Z}^n_\Delta(\mathcal{T})) \subset \widetilde{Z}^n_\Delta(\mathcal{L}),$$

so we set $\bar{f}_n =: \tilde{z}^n$.

To show that \tilde{z}_n induces a homomorphism on Chow–Witt groups, we observe directly from Diagram 4.4 and Diagram 4.5 that

$$\tilde{z}^n(\widetilde{B}^n_{\Delta}(\mathcal{T})) \subset \widetilde{B}^n_{\Delta}(\mathcal{L}).$$

Lemma 4.5.5. The diagram

from above commutes.

Proof. To see that this diagram commutes, we recall that it comes from the localization sequence Theorem 3.3.3. We have exact sequences (the horizontal lines) of triangulated categories with duality and (duality-preserving) maps between these as above such that the following diagram commutes.

$$\begin{array}{c} \mathcal{T}_{n-1} \xrightarrow{I_{\mathcal{T}}^{n}} \mathcal{T}_{(n)} \xrightarrow{Q_{\mathcal{T}}^{n}} \mathcal{T}_{(n)} / \mathcal{T}_{(n-1)} \\ \downarrow^{F_{n-1}} \downarrow^{F_{n}} \downarrow^{F_{n}} \downarrow^{F_{n}} \\ \mathcal{L}_{n-1} \xrightarrow{I_{\mathcal{L}}^{n}} \mathcal{L}_{(n)} \xrightarrow{Q_{\mathcal{L}}^{n}} \mathcal{L}_{(n)} / \mathcal{L}_{(n-1)} \end{array}$$

The horizontal sequences now give rise to long exact localization sequences (by Theorem 3.3.3) with induced maps between them. The square from the statement can now be found as a part of the lower diagram and, hence, commutes by construction of the differential in the localization sequences.

As an example, let us consider flat pullbacks and proper pushforwards.

Example 4.5.6. (i) Let $f : X \to Y$ be a faithfully flat morphism between separated schemes of finite type over a field and consider the rigid tensor triangulated categories $D^{perf}(X)$ and $D^{perf}(Y)$ equipped with the dimension function $-\operatorname{codim}_{Krull}$. Then, by [Kl16a, Lemma 4.3.2], the functor $Lf^* : D^{perf}(Y) \to D^{perf}(X)$ has relative dimension 0. Hence, it induces homomorphisms on the respective tensor triangular Chow groups that coincide with the flat pullbacks of the usual Chow groups (see [Kl16a, Section 4.3]).

For tensor triangular Chow–Witt groups, let K be a line bundle on Y. Assume that X and Y are regular. The functor

$$Lf^* : (D^{\operatorname{perf}}(Y), \operatorname{hom}(-, K)) \to (D^{\operatorname{perf}}(X), \operatorname{hom}(-, Lf^*(K)))$$

is duality-preserving by [CH11, Theorem 4.1] and, hence, it induces a morphism for all n between the respective Chow–Witt groups

$$Lf^*: \widetilde{\operatorname{CH}}^n_\Delta(\operatorname{D^{\operatorname{perf}}}(Y),K) \to \widetilde{\operatorname{CH}}^n_\Delta(\operatorname{D^{\operatorname{perf}}}(X),Lf^*(K)).$$

The author assumes that one can show that it coincides with the flat pullback for usual Chow–Witt groups. For a more general case of pullbacks, see [CH09, Theorem 4.1.2 and Corollary 4.1.3].

(ii) Let $f: X \to Y$ be a proper morphism between integral, regular, separated schemes of finite type over an algebraically closed field and consider the rigid tensor triangulated categories $D^{perf}(X)$ and $D^{perf}(Y)$ equipped with the dimension function \dim_{Krull} . Then, by [Kl16a, Lemma 4.4.1], the functor $Rf_*: D^{perf}(X) \to D^{perf}(Y)$ has relative dimension 0. Hence, it induces homomorphisms on the respective tensor triangular Chow groups that coincide with the proper pushforwards of the usual Chow groups (see [Kl16a, Section 4.4]).

For tensor triangular Chow–Witt groups, let K be a line bundle on Y. The functor

$$Rf_* : (D^{\operatorname{perf}}(X), \operatorname{hom}(-, f^!(K))) \to (D^{\operatorname{perf}}(Y), \operatorname{hom}(-, K))$$

is duality-preserving by [CH11, Theorem 4.4] and, hence, it induces a morphism for all n between the respective Chow–Witt groups

$$Rf_*: \widetilde{\operatorname{CH}}^n_{\Delta}(\mathrm{D}^{\mathrm{perf}}(X), f^!(K)) \to \widetilde{\operatorname{CH}}^n_{\Delta}(\mathrm{D}^{\mathrm{perf}}(X), K).$$

Here, $f^!$ is the right adjoint of Rf_* from [CH11, Theorem 3.8]. The author assumes that one can show that Rf_* coincides with the flat pullback for usual Chow–Witt groups. For a more general case of pushforwards, see [CH09, Corollary 4.3.3].

Summing up, this chapter has given a concrete definition of tensor triangular Chow–Witt groups in Section 4.3 after recalling tensor triangular Chow groups in Section 4.1 and classical Chow–Witt groups in Section 4.2. After this, we have seen the agreement with the algebro-geometric case in Section 4.4 and the functoriality above.

4.6 Chow(–Witt) Groups for Tensor Triangular Categories with Models

Convention 4.6.1. Let \mathcal{T} be as in Convention 4.3.1, but now we allow the existence of quotients of filtration components $\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)}$ that are not idempotent complete. We also assume that \mathcal{T} is idempotent complete. This implies that all filtration components $\mathcal{T}_{(n)}$ are also idempotent complete. We will fix L = 1 and, hence, omit L from the notation.

In this section, ideas are merely sketched and can be generalized to arbitrary \otimes -invertible *L*. We investigate the case of tensor triangular Chow groups at first, although their Definition 4.1.2 does not require quotients of filtration components to be idempotent complete, whereas we do require it for the construction of tensor triangular Chow–Witt groups as in Definition 4.3.5.

We will mainly use the notation from [Sch06].

We want to redefine Chow groups of tensor triangulated categories as introduced in Definition 4.1.2, in particular, when quotients of filtration components are not necessarily idempotent complete. Later, we want to generalize tensor triangular Chow–Witt groups Definition 4.3.5 to the above case.

The underlying motivation is the following: By [Sch06, Remark 1], we know that if $\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)}$ is not idempotent complete, $\mathbb{K}_{-1}(\mathcal{T}_{(n-1)})$ is nonzero. Consider the case of a *nice enough* singular scheme $X, \mathcal{T} = D^{\text{perf}}(X)$, still satisfying Bloch's formula but such that $\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)}$ is not idempotent complete. If Bloch's formula holds, we would like to use this cohomological description of its Chow groups to translate the definition to the tensor triangular setting as seen in Section 4.1 for regular schemes.

If X is regular and negative K-groups vanish, the tensor triangular *n*-cocycles are everything, i.e. $K_0((\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)})^{\natural})$; if not, negative K-groups appear and hence the tensor triangular *n*-cocycles may be a proper subgroup of $K_0((\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)})^{\natural})$. We thus want to redefine tensor triangular *n*-cocycles.

What changes in the non-singular case is that we do not have an equivalence of tensor triangulated categories $D^{perf}(X) \simeq D^b(Coh(X))$ and, hence, the quotients of the filtration components do not have a good description, but the idempotent completion can be described as

$$(\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)})^{\natural} \simeq \prod_{x \in X_{(n)}} \mathcal{D}_{\{x\}}^{\mathrm{perf}}(\mathcal{O}_{X,x}),$$

compare [Bal09, Proof of Theorem 2]. To adapt the construction of tensor triangular cocycles to this case, we thus need a localization sequence of the form

$$K_{0}(\mathcal{T}_{(n-1)}) \xrightarrow{i_{\mathcal{T}}^{n}} K_{0}(\mathcal{T}_{(n)}) \xrightarrow{q_{n}^{\prime}} K_{0}((\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)})^{\natural})$$

$$\xrightarrow{d_{n}^{\prime}} \mathbb{K}_{-1}(\mathcal{T}_{(n-1)}) \xrightarrow{i_{\mathcal{T}}^{n}} \mathbb{K}_{-1}(\mathcal{T}_{(n)}) \xrightarrow{q_{n}^{\prime}} \mathbb{K}_{-1}((\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)})^{\natural})$$

$$\xrightarrow{d_{n}^{\prime}} \mathbb{K}_{-2}(\mathcal{T}_{(n-1)}) \rightarrow \dots,$$

which we obtain from [Sch06, Theorem 1].

Since $\mathcal{T}_{(n)}$ is idempotent complete for all n, the maps $i_{\mathcal{T}}^n$ coincide in degree 0 with the maps $i_{\mathcal{T}}^n$ from Definition 4.1.2 and the maps q'_n in degree 0 are the composition of the maps $j_{\mathcal{T}}^n : K_0(\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)}) \to K_0((\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)})^{\natural})$, induced by idempotent completion, and $q_{\mathcal{T}}^n$ from Definition 4.1.2.

However, the price one has to pay for this is the introduction of *models* in order to obtain localization sequences, which means that we need our tensor triangulated category \mathcal{T} to *admit a model*. The above localization sequence is defined on the level of models, hence, there is a slight abuse of notation here. For more details on models in general, the reader is referred to [Sch06, Section 2.2]. The models Schlichting uses are *Frobenius models*, compare [Sch06, Sections 4-6], which are the same kind of models Klein uses in [Kl16b] to introduce a ring structure on tensor triangular Chow groups, but other models categories are possible such as the category of small exact categories.

In particular, our most prominent examples $D^{\text{perf}}(X)$ for a noetherian k-scheme X and the stable module category kG- stab for G a finite group and k a field with characteristic dividing the order of G are such tensor triangulated categories which admit for instance Frobenius models; more examples of Frobenius models can be found in [Sch06, Chapter 6]. To see why admitting models might not be too strong a restriction the reader is referred to the introduction of [Sch06].

Hence, we can introduce the following variant of Definition 4.1.2, where the idea is to extend the Gersten complex to negative degrees.

Definition 4.6.2. Let \mathcal{T} be an idempotent complete, essentially small tensor triangulated category equipped with a dimension function dim that admits a model and such that $\operatorname{Spc}(\mathcal{T})$ is noetherian. When considering \mathcal{T} a triangulated category with duality, we fix the duality hom(-, 1) and restrictions of it on subcategories, quotients and idempotent completions (by Lemma 2.3.7, Lemma 2.3.8). Then, we define

$$Z^n_{\Delta'}(\mathcal{T}) := \ker(q'_{n-1} \circ d'_n) \subset K_0((\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)})^{\natural})$$

which, by exactness of the above localization sequence, equals

$$(d'_n)^{-1}(\operatorname{Im}(i_{\mathcal{T}}^{n-1})),$$

and

$$CH^{n}_{\Delta'}(\mathcal{T}) := Z^{n}_{\Delta'}(\mathcal{T})/q'_{n}(\ker(i^{n+1}_{\mathcal{T}}))$$
$$= Z^{n}_{\Delta'}(\mathcal{T})/j^{n}_{\mathcal{T}} \circ q^{n}_{\mathcal{T}}(\ker(i^{n+1}_{\mathcal{T}})).$$

The algebraic *n*-cocycles from Definition 4.6.2 and Balmer's Definition 4.1.2 differ a priori if the quotient $\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)}$ is not idempotent complete, i.e. if $j_{\mathcal{T}}^n$ is not the identity, but the coboundaries coincide.

It follows from the construction that for $\mathcal{T} = D^{\text{perf}}(X)$, X non-singular (so, in particular, for $j_{\mathcal{T}}^n$ the identity), $\operatorname{CH}^n_{\Delta'}(\mathcal{T})$ and $\operatorname{CH}^n_{\Delta}(\mathcal{T})$, and thus by Lemma 4.1.3 also $\operatorname{CH}^{-n}(X)$, coincide.

Remark 4.6.3. For noetherian schemes which may be singular, the idea of adapting Balmer's definition of tensor triangular cocycles for the derived category of perfect complexes already appears in the unpublished work [Ya16, Definition 3.6 and Section 3.2]. We sketched the generalized idea for general idempotent complete, essentially small tensor triangulated categories with a dimension function that admit models, hence, in particular, for $\mathcal{T} = kG$ -stab, where G is a finite group and k a field such that char(k) divided |G|.

Let X be a connected quasi-projective variety of dimension $d \geq 2$ whose singular locus $\operatorname{Sing}(X)$ is contained in a finite closed set Y. If X is *nice enough*, i.e. it in particular satisfy some kind of Bloch's formula (see [Ya16, Question 3.19]), Yang's definition for $\mathcal{T} = D^{\operatorname{perf}}(X)$ coincides with the relative Chow groups $\operatorname{CH}(X, Y)$ of Pedrini and Weibel from [PW86] by [Ya16, Corollary 3.21] after tensoring with \mathbb{Q} , hence so they do with Definition 4.6.2 after tensoring with \mathbb{Q} .

Balmer's tensor triangular *n*-cocycles from Definition 4.1.2 differ a priori from $Z^n_{\Delta'}(\mathcal{T})$ and for this reason in the special case mentioned above also from the relative Chow groups from Pedrini and Weibel.

In the case of tensor triangular Chow–Witt groups, we observe something similar. As usual, we now assume \mathcal{T} to be rigid, $\mathbb{Z}[\frac{1}{2}]$ -linear, and dualities as in Convention 4.3.1 when considering \mathcal{T} a triangulated category with duality.

The Witt group $W^{[n+1]}(\mathcal{T}_{(n-1)})$ may be nonzero even when $\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)}$ is idempotent complete. As a consequence, the oriented *n*-cocycles may be a proper subgroup of $\mathrm{GW}^{[n]}((\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)})^{\natural})$, which follows from Definition 4.3.3. If the quotient is idempotent complete, the localization sequence from Theorem 3.3.3 still suffices to define tensor triangular Chow–Witt groups as we did in Definition 4.3.5.

If the quotient is not idempotent complete, as above we need a sequence of the form

$$\operatorname{GW}^{[n]}(\mathcal{T}_{(n-1)}) \xrightarrow{\tilde{i}_{\mathcal{T}}^{n}} \operatorname{GW}^{[n]}(\mathcal{T}_{(n)}) \xrightarrow{\tilde{q}'_{n}} \operatorname{GW}^{[n]}((\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)})^{\natural})$$

$$\xrightarrow{\tilde{d}'_{n}} \operatorname{W}^{[n+1]}(\mathcal{T}_{(n-1)}) \xrightarrow{\tilde{i}_{\mathcal{T}}^{n,\mathrm{W}}} \operatorname{W}^{[n+1]}(\mathcal{T}_{(n)}) \xrightarrow{\tilde{q}'_{n,\mathrm{W}}} \operatorname{W}^{[n+1]}((\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)})^{\natural})$$

$$\xrightarrow{\tilde{d}'_{n}} \operatorname{W}^{[n+2]}(\mathcal{T}_{(n-1)}) \to \dots,$$

which we have after tensoring with $\mathbb{Z}[\frac{1}{2}]$ essentially by [Bal09, Remark 3]. Note that here, models have to carry some notion of *duality* such as complicial BiWald-hausen categories, compare [Bal09] and [Sch06, Section 6.5].

Again, the maps $\tilde{i}_{\mathcal{T}}^{n(,W)}$ coincide with the maps $i_{\mathcal{T}}^{n(,W)}$ from Definition 4.3.5 and the maps $\tilde{q}'_{n(,W)}$ are the composition of the map $\tilde{j}_{\mathcal{T}}^{n}$: GW^[n] $(\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)}) \rightarrow$ GW^[n] $((\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)})^{\natural})$ and $\tilde{j}_{\mathcal{T}}^{n,W}$: W^[n] $(\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)}) \rightarrow$ W^[n] $((\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)})^{\natural})$, respectively, induced by idempotent completion, and $q_{\mathcal{T}}^{n(,W)}$ from Definition 4.3.5.

We thus have a variant of Definition 4.3.5 for tensor triangulated categories admitting models:

Definition 4.6.4. Let \mathcal{T} be a $\mathbb{Z}[\frac{1}{2}]$ -linear, essentially small, idempotent complete, rigid tensor triangulated category equipped with a dimension function admitting

a model and such that $\operatorname{Spc}(\mathcal{T})$ is noetherian. When considering \mathcal{T} a triangulated category with duality, we fix the duality hom(-, 1) and restrictions of it on subcategories. Moreover, we consider all groups being tensored by $\mathbb{Z}[\frac{1}{2}]$ here. We define

$$\widetilde{Z}^n_{\Delta'}(\mathcal{T}) := (\widetilde{d}'_n)^{-1}(\operatorname{Im}(\widetilde{i}^{n-1,W}_{\mathcal{T}})) \\ = \ker(\widetilde{q}'_{n-1,W} \circ \widetilde{d}'_n)$$

and

$$\widetilde{\operatorname{CH}}^n_{\Delta'}(\mathcal{T}) := \widetilde{Z}^n_{\Delta'}(\mathcal{T}) / \widetilde{q}'_n(\ker(\widetilde{i}^{n+1}_{\mathcal{T}})) = \widetilde{Z}^n_{\Delta'}(\mathcal{T}) / \widetilde{j}^n_{\mathcal{T}} \circ q^n_{\mathcal{T}}(\ker(i^{n+1}_{\mathcal{T}})).$$

When $\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)}$ is idempotent complete for all n, for example for $\mathcal{T} = D^{\text{perf}}(X)$ and X smooth, Definition 4.3.5 and Definition 4.6.4 coincide since $\tilde{j}_{\mathcal{T}}^n$ is the identity in this case. Moreover, when replacing GW by K_0 in the construction, Definition 4.6.4 agrees with Definition 4.6.2 and thus with Definition 4.1.2 when all quotients $\mathcal{T}_{(n)}/\mathcal{T}_{(n-1)}$ are idempotent complete.

In this section, we have redefined our definition of tensor triangular Chow–Witt groups for cases when quotients of the filtration components $\mathcal{T}_{(n)}$ are not idempotent complete. For this, we sketched the idea of introducing models for (rigid) tensor triangulated categories.

The following chapter is dedicated to concrete calculations of tensor triangular Chow–Witt groups in the setting of modular representation theory.
Chapter 5

Computations in Modular Representation Theory

In this chapter, we want to compute examples of tensor triangular Chow–Witt groups. Since, for the derived category of perfect complexes over a regular scheme, the definition coincides with the algebro-geometric one for regular schemes (Theorem 4.4.5), the field of algebraic geometry already provides us with many known examples. It is therefore interesting to consider Chow–Witt groups in new areas Definition 4.3.5 and Definition 4.6.4 have opened, for example in modular representation theory.

To do so, we will recall some basic concepts on group rings and the stable module category in Section 5.1. In Section 5.2, we will collect helpful theorems and propositions that will serve as tools for the computation of the Chow–Witt group of the stable module category kG-stab – first for $G = \mathbb{Z}/p^n\mathbb{Z}$ and then we will give partial results for $G = \mathbb{Z}/p\mathbb{Z} \times \mathbb{Z}/p\mathbb{Z}$ with $p \neq 2$.

5.1 The Stable Module Category

Our input candidate for tensor triangular Chow–Witt groups in a non-algebrogeometric setting will be the tensor triangulated category kG-stab, known as the stable module category, for a field k whose characteristic divides the group order. We will focus on two examples, namely $G = \mathbb{Z}/p\mathbb{Z}$ and $G = \mathbb{Z}/p\mathbb{Z} \times \mathbb{Z}/p\mathbb{Z}$ for $p \neq 2$. Already for the latter, obstacles will occur so that we will only give partial results here.

In this section, we first collect basic notions and results on group rings and then on the stable module category for the reader's convenience. As an introductory lecture, we recommend [Car96, BIK12, Ben98a]; we mainly follow these references in this section.

5.1.1 Recollection on Group Rings

We recall some essential notations and results concerning group rings.

Convention 5.1.1. Let G be a finite group and k a field of positive characteristic p.

Definition 5.1.2. We define the group ring kG as the set of all formal k-linear combinations $\sum_{g \in G} \alpha_g g$, where $\alpha_g \in k$ and $g \in G$. It becomes a ring and even a k-algebra via the compositions

$$\sum_{g \in G} \alpha_g g + \sum_{g \in G} \beta_g g := \sum_{g \in G} (\alpha_g + \beta_g) g$$

and

$$\left(\sum_{g\in G}\alpha_g g\right)\cdot\left(\sum_{h\in G}\beta_h h\right):=\sum_{g,h\in G}(\alpha_g\beta_h)gh.$$

As a k-module, kG is free with basis G.

Example 5.1.3. Let $G = \mathbb{Z}/p^n\mathbb{Z} = \langle g \mid g^{p^n} = 1 \rangle$, with g generating G, and char k = p. Then, the group ring is given by $kG \cong k[y]/(y^{p^n} - 1)$ which is isomorphic to $k[x]/(x^{p^n})$ via $x \mapsto y - 1$.

For later calculations, it will be important for us to have conditions whether a group ring is local and to then determine its unique maximal ideal as well as its residue field.

Lemma 5.1.4. ([Car96, Corollary 1.4]) If G is a finite p-group, the group ring kG is local and its unique maximal ideal is the augmentation ideal (see [Car96, Chapter 1]).

Lemma 5.1.5. ([Car96, Proposition 7.4]) For groups A and B, we have $k(A \times B) \cong kA \otimes_k kB$.

- **Example 5.1.6.** (i) For $G = \mathbb{Z}/p^n\mathbb{Z}$, the unique maximal ideal of the local ring $kG \cong k[x]/(x^{p^n})$ is given by (x), hence, its residue field kG/(x) simply is k.
 - (ii) For $G = \mathbb{Z}/p\mathbb{Z} \times \mathbb{Z}/p\mathbb{Z}$, by Lemma 5.1.5 and (i), we have that

$$kG \cong k[x]/(x^p) \otimes_k k[y]/(y^p) \cong k[x,y]/(x^p,y^p)$$

with maximal ideal (x, y), hence the residue field kG/(x, y) again is k. More generally, the residue field of kE for any finite elementary abelian p-group E is k by the same argument.

(iii) The group ring $k\mathbb{Z}$ equals the ring of Laurent polynomials and is not local.

Remark 5.1.7. Note that since k is a field (more general a (left) Artinian ring), kG is left Artinian by [BIK12, Remark 1.24]. It follows that, for modules over kG, being finitely generated and having finite length is equivalent. This will be important for us in the computation of the Grothendieck–Witt group of kG-stab, the first step in the computation of its Chow–Witt group.

The following results on the semisimplicity of kG are well-known.

Theorem 5.1.8. (Maschke) If the characteristic of k does not divide the order of G, then kG is a semisimple k-algebra, i.e. every nonzero kG-module is semisimple (if moreover kG is finite-dimensional).

Theorem 5.1.9. (Converse of Maschke's theorem) If kG is a semisimple kalgebra, then the characteristic of k does not divide the order of G.

For the notion of a *Frobenius algebra* we follow [Hap88].

Definition 5.1.10. A k-algebra R is called *Frobenius algebra* if indecomposable projective R-modules and indecomposable injective R-modules coincide and if moreover it is *locally bounded* (see [Hap88, p.25]).

Example 5.1.11. Let G be a finite group and k a field, then kG is a finitedimensional Frobenius k-algebra by [Ben98a, Proposition 3.1.2].

For our purposes, this is all we need to know about group rings. In the following, we recall the construction and basic results concerning the stable module category kG-stab.

5.1.2 Construction of kG-stab

Convention 5.1.12. If not stated otherwise, modules in this chapter we consider will always be finitely generated left modules. For the rest of this chapter, let G be a finite group and k a field of characteristic p dividing the order of G.

We denote by kG-mod the abelian category of finitely generated left kG-modules. Recall that since k is a field, being finitely generated and having finite length is equivalent due to Remark 5.1.7. The category kG-mod has the following structural properties.

Remark 5.1.13. (i) The tensor product $M \otimes_k N$ of finitely generated left kG-modules is again an object of kG-mod when considering the linear extension of the diagonal action

$$g(m \otimes n) := gm \otimes gn$$

for $g \in G$, $m \in M$, and $n \in N$. Moreover, the set $\operatorname{Hom}_k(M, N)$ of k-linear maps from M to N is an object of kG- mod via linear extension of the action

$$(gf)(m) := f(g^{-1}m)$$

for $g \in G$, $m \in M$ and $f \in \operatorname{Hom}_k(M, N)$.

(ii) The category kG-mod becomes an abelian category with duality when equipping it with the duality $\operatorname{Hom}_k(-,k)$, where k carries the trivial kGmodule structure. The category theoretic analogon of Definition 5.1.10 is the following.

Definition 5.1.14. A Frobenius category is an exact category \mathcal{E} which has enough injectives, enough projectives, and projective and injective objects coincide. A functor of Frobenius categories is a functor preserving projective-injective objects.

Example 5.1.15. For G a finite group and k a field, the abelian (hence, exact) category kG-mod is a Frobenius category. This follows from Example 5.1.11 and [Ben98a, Proposition 1.6.2].

Now, we can define the stable module category via the category kG-mod in the following way.

Definition 5.1.16. Let G be a finite group and k a field whose characteristic divides the group order. For the abelian Frobenius category kG-mod, we denote by kG-stab its associated *stable category* with the same objects as kG-mod. The morphism group between two kG-modules M and N is given as the quotient

$$\operatorname{Hom}_k(M, N)/J,$$

where J is the subgroup of morphisms factoring through projective-injective objects of kG-mod. This construction is called the *stable module category*.

More generally, we can define the stable category of any Frobenius category in an analog manner.

Remark 5.1.17. A construction like in Definition 5.1.16 is called a *quotient category*. Note that the construction of a *localization of a category* is different in general since the amount of morphisms may increase by adding formal inverses to create isomorphisms, whereas, in a quotient category, the number of morphisms decreases. In both cases though, objects can become isomorphic that have not been isomorphic before. The construction of a *Serre quotient* has parallels to the construction of a quotient category, but behaves like a localization of a category in many cases.

Together with the definition of the stable module category come Hellers inverse loop space functors Ω and Ω^{-1} that encode the information of kernels of projective covers and of cokernels of injective hulls, respectively. Keep in mind that in our case, kG-mod is a Frobenius category, hence, injective and projective modules coincide.

Definition-Lemma 5.1.18. ([He60]) Let $\pi : P \to M$ be a projective cover of a kG-module M. Then, we define $\Omega(M) := \ker(\pi)$. Moreover, for an injective hull $i: M \to I$ of M, we define $\Omega^{-1}(M) := \operatorname{coker}(i)$.

Two projective covers (or injective hulls) are isomorphic to each other in kG-stab. We obtain well-defined functors $\Omega, \Omega^{-1} : kG$ -stab $\rightarrow kG$ -stab which are inverse to each other, yielding an autoequivalence of categories.

For our purposes, we of course are interested in the triangulated structure of kG-stab, which turns out to be even a rigid tensor triangulated category.

Proposition 5.1.19. ([Car96, Theorem 5.6]) The category kG-stab is a triangulated category with shift functor $T := \Omega^{-1}$ and exact triangles as follows. For $f \in \operatorname{Hom}_{kG-\operatorname{stab}}(X,Y), X \xrightarrow{f} Y \xrightarrow{g} Z \xrightarrow{h} T(X)$ is a standard triangle in kG-stab if Z is the pushout in kG-mod of f and $i_X : X \to I(X)$, where I(X) is an injective hull, and if g and h appear in the following commutative diagram

$$\begin{array}{cccc} 0 & \longrightarrow X \xrightarrow{i_X} I(X) & \longrightarrow T(X) & \longrightarrow 0 \\ & & & \downarrow^f & & \downarrow \\ 0 & \longrightarrow Y \xrightarrow{g} Z \xrightarrow{h} T(X) & \longrightarrow 0. \end{array}$$

The exact triangles are the ones isomorphic to such a standard triangle.

Proposition 5.1.20. ([Bal07, Proposition 4.2]) Let G be a finite group and k a field of characteristic dividing the group order. The stable module category kG-stab is a rigid tensor triangulated category with tensor product \otimes_k and internal Hom hom := Hom_k.

Remark 5.1.21. (i) Combining Proposition 5.1.20 and Example 2.2.6 (i), we can consider kG- stab a triangulated category with duality $\operatorname{Hom}_k(-, k)$.

We can see by hand that the duality is well-defined: Let A be in kG-mod. For any projective P, we have

$$\operatorname{Hom}_k(A \oplus P, k) \cong \operatorname{Hom}_k(A, k) \oplus \operatorname{Hom}_k(P, k)$$

and since P is projective and kG-mod Frobenius, $P^* = \operatorname{Hom}_k(P, k)$ is also projective. It follows that $\operatorname{Hom}(A \oplus P, k) \cong \operatorname{Hom}(A, k)$ in kG-stab by Theorem 5.2.2. If $f \sim g : A \to B$ in kG-stab, then $b \circ f \sim b \circ g$ for all $b \in \operatorname{Hom}_k(B, k) = B^*$, hence, $f^* \sim g^*$.

(ii) As already mentioned in [Kl16a, Remark 6.2.3], we deliberately consider \otimes_k and consequently $\operatorname{Hom}_k(-,k)$ as opposed to \otimes_{kG} . This is because the tensor product $M \otimes_{kG} N$ of two left kG-modules M, N does not have a natural left module structure.

We can view kG-stab as an Verdier localization of the derived category of kG-modules $D^b(kG - mod)$ as the following result by Rickard shows.

Theorem 5.1.22. ([Ric89, Theorem 2.1]) Let G be a finite group and k a field dividing the group order. The natural functor $kG \operatorname{-mod} \to D^b(kG \operatorname{-mod})$ induces an equivalence of tensor triangulated categories

$$F: kG\operatorname{-stab} \xrightarrow{\simeq} \mathrm{D}^{b}(kG\operatorname{-mod})/\mathrm{D}^{b}(kG\operatorname{-proj}),$$

where kG-proj denotes the subcategory of kG-mod generated by projective kG-modules.

In our case, we need to make sure that the above equivalence preserves duality. The rigid tensor triangulated category kG-stab becomes a triangulated category with duality when fixing the duality $\operatorname{Hom}_k(-, k)$, as Remark 5.1.21 shows.

Lemma 5.1.23. The above equivalence F is duality-preserving.

Proof. The inclusion $kG \operatorname{-} \operatorname{mod} \to \operatorname{D}^{b}(kG \operatorname{-} \operatorname{mod})$ sending a module M to the complex $\ldots \to 0 \to M \to 0 \to \ldots$ concentrated in degree 0 is duality-preserving and, thus, also $kG \operatorname{-} \operatorname{mod} \to \operatorname{D}^{b}(kG \operatorname{-} \operatorname{mod})/\operatorname{D}^{b}(kG \operatorname{-} \operatorname{proj})$. Since $kG \operatorname{-} \operatorname{mod}$ and $kG \operatorname{-} \operatorname{stab}$ have the same objects, the equivalence is duality-preserving on the level of objects. On the level of morphisms, note that kG-stab and $\operatorname{D}^{b}(kG \operatorname{-} \operatorname{mod})/\operatorname{D}^{b}(kG \operatorname{-} \operatorname{stab})$ are well-defined with respect to taking duals. The statement now again follows from the fact that $kG \operatorname{-} \operatorname{mod} \to \operatorname{D}^{b}(kG \operatorname{-} \operatorname{mod})/\operatorname{D}^{b}(kG \operatorname{-} \operatorname{proj})$ is duality-preserving and duality in $\operatorname{D}^{b}(kG \operatorname{-} \operatorname{mod})$ is defined degreewise: Let $f: M \to N$ be a morphism in kG- mod and \overline{f} its equivalence class in kG-stab. Then,

$$F(\overline{f^*}) = F(\overline{f^*}) = [f^*] = [f]^*,$$

where [-] denotes the class of a morphism in $D^{b}(kG - mod)/D^{b}(kG - proj)$.

Apart from the rigid tensor triangular structure, we were moreover interested in dimension functions Section 2.3.

Definition 5.1.24. Let $\mathcal{P} \in \text{Spc}(kG\text{-stab})$. We define its *Krull dimension* $\dim_{\text{Krull}}(\mathcal{P})$ as the maximal length n of a chain of irreducible closed subsets

$$\emptyset \subsetneq C_0 \subsetneq C_1 \subsetneq \dots \subsetneq C_n = \overline{\{\mathcal{P}\}}.$$

It gives rise to a dimension function in the sense of Definition 2.3.1 by [Bal07, Example 3.2].

We will now see that the *cohomology ring* of a group algebra coming with its *projective support variety* already encodes all the information on the Balmer spectrum of kG-stab.

Definition 5.1.25. We define the cohomology ring of kG as the graded ring

$$H^{\bullet}(G,k) := \bigoplus_{i \ge 0} \operatorname{Ext}^{i}_{kG}(k,k)$$

if p = 2, and only as its even part if the characteristic p is odd.

Definition 5.1.26. The projective support variety of kG is given by

$$\mathcal{V}_G(k) := \operatorname{Proj}(H^{\bullet}(G,k)).$$

The projective support variety $\mathcal{V}_G(M)$ of a kG-module M is defined as the subvariety of $\mathcal{V}_G(k)$ associated to the annihilator ideal J(M) of $\operatorname{Ext}_k^*(M, M)$ in $H^{\bullet}(G, k)$.

Proposition 5.1.27. ([Bal05b, Corollary 5.10]) Let G be a finite group and k a field with characteristic dividing the group order. There is a homeomorphism

$$\mathcal{V}_G(k) \xrightarrow{\cong} \operatorname{Spc}(kG\operatorname{-stab})$$
$$x \mapsto \{M \in kG\operatorname{-stab} \mid x \notin \mathcal{V}_G(M)\}$$

Under this homeomorphism, $\mathcal{V}_G(M)$ corresponds to $\operatorname{supp}(M) \subset \operatorname{Spc}(kG\operatorname{-stab})$ for M an object of $kG\operatorname{-stab}$.

Moreover, we can compute the dimension of the support variety of a module via its *complexity*, an invariant depending on the growth of the dimension of components in projective resolutions.

Theorem 5.1.28. ([Car96, Theorem 4.3]) Any object M of kG-mod has a minimal projective resolution, that is, a projective resolution $P_{\bullet} \to M$ such that for any other projective resolution $Q_{\bullet} \to M$ there exists an injective chain map $(P_{\bullet} \to M) \to (Q_{\bullet} \to M)$ and a surjective chain map $(Q_{\bullet} \to M) \to (P_{\bullet} \to M)$ that lift the identity on M.

Definition 5.1.29. For a minimal projective resolution $P_{\bullet} \to M$ of an object M in kG-mod its *complexity* $c_G(M)$ is defined as the smallest $s \in \mathbb{Z}$ such that there is a constant $\kappa > 0$ with

$$\dim_k(P_n) \le \kappa \cdot n^{s-1} \text{ for } n > 0.$$

We obtain the following correspondence between the dimension of $\mathcal{V}_G(M)$ and the complexity of M.

Proposition 5.1.30. ([Ben98b, Proposition 5.7.2]) Let G be a finite group and k a field of characteristic dividing the group order. For an object M of kG-mod, we have

$$\dim(\mathcal{V}_G(M)) = c_G(M) - 1.$$

Remark 5.1.31. The original statement of [Ben98b, Proposition 5.7.2] is that the complexity of a kG-module M equals the dimension of $\mathcal{V}_G(M)$. However, in [Ben98b, Section 5.1], $\mathcal{V}_G(M)$ is defined as the affine variant of the projective support variety we defined in Definition 5.1.26 and carries the name cohomological variety. Their dimensions differ by 1, hence, so do the statements of [Ben98b, Proposition 5.7.2] and Proposition 5.1.30.

In this section, we have recalled foundations on the group ring kG and the stable module category kG-stab. Next, we discuss in more detail the relation between the Frobenius category kG- mod and its stable category kG-stab. The (Grothendieck–)Witt group of the latter is harder to calculate, so we give a statement on how to compute it in terms of the (Grothendieck–)Witt group of kG- mod.

5.2 The (Grothendieck–)Witt Group of kG-stab

Convention 5.2.1. Let G be a finite group and k a field of characteristic $\neq 2$ dividing the group order. When considering kG-stab a triangulated category with duality, we fix the duality $\text{Hom}_k(-, k)$.

As mentioned before, this section intends to simplify the computations of the groups $\mathrm{GW}^{[n]}(kG\operatorname{-stab})$ and $\mathrm{W}^{[n]}(kG\operatorname{-stab})$, $n \in \mathbb{Z}$, by writing it in terms of $\mathrm{GW}^{[n]}(kG\operatorname{-mod})$ and $\mathrm{W}^{[n]}(kG\operatorname{-mod})$, respectively. Calculations like these will

turn out to be the first step in the determination of tensor triangular Chow-Witt groups of kG-stab in Section 5.3 and Section 5.5. Since the (Grothendieck-)Witt groups (and, for that matter, also the K_0 -groups) of kG-stab are harder to calculate by hand than the respective groups of kG-mod, we will use a variation of [TaWa91, Proposition 1] to break it down to a computation involving merely the Frobenius category kG-mod.

For this, we first need to understand how equivalences translate between kG-stab and kG-mod. The category kG-mod is a category with cancellation, compare [He58, p.486/487], which is why we can apply [He60, Theorem 2.2] to our case to obtain the following result.

Theorem 5.2.2. ([He60, Theorem 2.2]) Let $f : M \to N$ be a morphism in kG-mod. Then, the following are equivalent:

- (i) [f] is an isomorphism in kG-stab
- (ii) There are projective modules M', N' and an isomorphism $f' : M \oplus M' \to N \oplus N'$ in kG-mod making the following diagram in kG-mod commute



With these tools at hand, we can prove the following theorem. It will be central in the computation of tensor triangular Chow–Witt groups of kG-stab since the first step is usually to determine its (shifted) Grothendieck–Witt group. The theorem is a generalization of a similar statement for K_0 , namely [TaWa91, Proposition 1], that Klein uses to calculate tensor triangular Chow groups of kG-stab in [Kl16a].

Theorem 5.2.3. Let G be a finite group and k a field of characteristic $\neq 2$ dividing the order of G; all dualities here are given by $\operatorname{Hom}_k(-,k)$.

(i) The quotient map $kG \operatorname{-mod} \to kG$ -stab from Definition 5.1.16 induces isomorphisms for even n

$$\operatorname{GW}^{[n]}(kG\operatorname{-stab}) \cong \operatorname{GW}^{[n]}(\operatorname{D}^{b}(kG\operatorname{-mod})) / \operatorname{Im}(\alpha),$$

where $\alpha : \mathrm{GW}^{[n]}(\mathrm{D}^{b}(kG\operatorname{-}\operatorname{proj})) \to \mathrm{GW}^{[n]}(\mathrm{D}^{b}(kG\operatorname{-}\operatorname{mod}))$ is induced by the inclusion $kG\operatorname{-}\operatorname{proj} \to kG\operatorname{-}\operatorname{mod}$.

In particular,

$$\operatorname{GW}(kG\operatorname{-stab}) \cong \operatorname{GW}(kG\operatorname{-mod})/\operatorname{Im}(\alpha')$$

for $\alpha' : \mathrm{GW}(kG\operatorname{-}\operatorname{proj}) \to \mathrm{GW}(kG\operatorname{-}\operatorname{mod})$ induced by the inclusion $kG\operatorname{-}\operatorname{proj} \to kG\operatorname{-}\operatorname{mod}$.

(ii) The quotient map $kG \operatorname{-mod} \to kG$ -stab from Definition 5.1.16 induces isomorphisms for all n

. .

$$\mathbf{W}^{[n]}(kG\operatorname{-stab}) \cong \mathbf{W}^{[n]}(\mathbf{D}^{b}(kG\operatorname{-mod})) / \mathbf{W}^{[n]}(\mathbf{D}^{b}(kG\operatorname{-proj})).$$

In particular,

$$W(kG\operatorname{-stab}) \cong W(kG\operatorname{-mod})/W(kG\operatorname{-proj})$$

and $W^{[n]}(kG\operatorname{-stab}) = 0$ for n odd.

Proof. We prove (ii) first. Let m be odd. Then, by [BW02, Proposition 5.2], $W^{[m]}(D^b(kG \operatorname{-mod})) = 0 = W^{[m]}(D^b(kG \operatorname{-proj})).$ Moreover, by Theorem 5.1.22 and Theorem 4.1.4, which is applicable by Lemma 5.2.4 below, we have

$$W^{[m]}(kG\operatorname{-stab}) \cong W^{[m]}(D^{b}(kG\operatorname{-mod})/D^{b}(kG\operatorname{-proj}))$$
$$\cong W^{[m]}(D^{b}(kG\operatorname{-mod}/kG\operatorname{-proj})) = 0,$$

again using [BW02, Proposition 5.2]. Now, let n be even. We have Balmer's localization sequence Theorem 3.3.2

$$\dots \to \mathcal{W}^{[n-1]}(\mathcal{D}^{b}(kG\operatorname{-mod})/\mathcal{D}^{b}(kG\operatorname{-proj})) \to \mathcal{W}^{[n]}(\mathcal{D}^{b}(kG\operatorname{-proj})) \to \mathcal{W}^{[n]}(\mathcal{D}^{b}(kG\operatorname{-mod}))$$

$$\to \mathcal{W}^{[n]}(\mathcal{D}^{b}(kG\operatorname{-mod})/\mathcal{D}^{b}(kG\operatorname{-proj})) \to \mathcal{W}^{[n+1]}(\mathcal{D}^{b}(kG\operatorname{-proj})) \to \dots .$$

By the above observations, we have that $W^{[n-1]}(D^b(kG - mod)/D^b(kG - proj)) =$ $0 = W^{[n+1]}(D^b(kG\operatorname{-proj}))$. Consequently, the statement follows from the isomorphism theorem of abelian groups.

Using Proposition 3.2.13, Proposition 3.2.14, and Proposition 3.2.16, we observe that, for an abelian category \mathcal{A} , the Grothendieck–Witt group of \mathcal{A} coincides with the Grothendieck–Witt group of $D^b(\mathcal{A})$. It follows directly that $W(kG\operatorname{-stab}) \cong W(kG\operatorname{-mod})/W(kG\operatorname{-proj}).$

For (i), consider Walter's localization sequence Theorem 3.3.3

$$\operatorname{GW}^{[n]}(\mathrm{D}^{b}(kG\operatorname{-\operatorname{proj}})) \xrightarrow{\alpha} \operatorname{GW}^{[n]}(\mathrm{D}^{b}(kG\operatorname{-\operatorname{mod}})) \to \operatorname{GW}^{[n]}(\mathrm{D}^{b}(kG\operatorname{-\operatorname{mod}})/\operatorname{D}^{b}(kG\operatorname{-\operatorname{proj}})) \to \operatorname{W}^{[n+1]}(\mathrm{D}^{b}(kG\operatorname{-\operatorname{proj}})) \to \dots .$$

For even n, $W^{[n+1]}(D^b(kG\operatorname{-proj})) = 0$ by [BW02, Proposition 5.2]. It follows that the morphism $\mathrm{GW}^{[n]}(\mathrm{D}^{b}(kG\operatorname{-mod})) \to \mathrm{GW}^{[n]}(\mathrm{D}^{b}(kG\operatorname{-mod})/\mathrm{D}^{b}(kG\operatorname{-proj})) \cong \mathrm{GW}^{[n]}(kG\operatorname{-stab})$ is an epimorphism, where the latter isomorphism again follows from Theorem 5.1.22 and Theorem 4.1.4. The first statement now follows from the exactness of the sequence and the isomorphism theorem of abelian groups.

The second statement follows as in the case of Witt groups using that $\mathrm{GW}(\mathcal{A}) \cong$ $\mathrm{GW}(D^b(\mathcal{A}))$ for an abelian category \mathcal{A} .

Lemma 5.2.4. The condition for Theorem 4.1.4 is satisfied for the inclusion kG-proj $\rightarrow kG$ -mod. Moreover, the equivalence

$$D^{b}(kG \operatorname{-mod} / kG \operatorname{-proj}) \simeq D^{b}(kG \operatorname{-mod}) / D^{b}(kG \operatorname{-proj})$$

is duality-preserving.

Proof. Let $0 \to P \to B \to C \to 0$ be a short exact sequence in kG-mod with P in kG-proj. The category kG-mod is Frobenius, hence, we have an injective hull $i_C : C \to I(C)$ with I(C) in kG-proj. Note that I(C) and P are both injective and projective. Consequently, the given sequence splits via an isomorphism $f : B \xrightarrow{\cong} P \oplus C$. Then, the following diagram commutes and the rows are exact:



The second statement follows from the fact that the equivalence is induced by the duality-preserving short exact sequence

$$D^{b}(kG\operatorname{-}\operatorname{proj}) \to D^{b}(kG\operatorname{-}\operatorname{mod}) \to D^{b}(kG\operatorname{-}\operatorname{mod}/kG\operatorname{-}\operatorname{proj}).$$

Remark 5.2.5. In general, $\alpha : \operatorname{GW}^{[n]}(kG\operatorname{-}\operatorname{proj}) \to \operatorname{GW}^{[n]}(kG\operatorname{-}\operatorname{mod})$ from Theorem 5.2.3 above might not be injective and, hence, $\operatorname{Im}(\alpha)$ might not be isomorphic to $\operatorname{GW}^{[n]}(kG\operatorname{-}\operatorname{proj})$, similar to the K-theoretic case in [TaWa91, Proposition 1]. For odd n, the map $\operatorname{GW}^{[n]}(kG\operatorname{-}\operatorname{mod}) \to \operatorname{GW}^{[n]}(\operatorname{D}^{b}(kG\operatorname{-}\operatorname{mod})/\operatorname{D}^{b}(kG\operatorname{-}\operatorname{proj}))$ is not even surjective in general, so other techniques are required to investigate this case. We will not pursue this issue further here.

In order to do further calculations, we will moreover need the following version of the dévissage theorem, for GW and W, respectively. For readers not familiar with the theory of (Grothendieck–)Witt groups of fields, [Scha85, Lam05, MH73] are recommended as an introductory reference.

Proposition 5.2.6. For k a field of characteristic $p \neq 2$ and G a finite p-group, the forgetful functor $U : kG \mod k \mod induces$ isomorphisms

$$W(kG \operatorname{-mod}, \operatorname{Hom}_k(-, k)) \cong W(k)$$

and

$$\operatorname{GW}(kG\operatorname{-}\operatorname{mod},\operatorname{Hom}_k(-,k)) \cong \operatorname{GW}(k),$$

where the left-hand side denotes the shifted (Grothendieck–)Witt group of an exact category with duality as in Definition 3.2.5.

Proof. We know from Lemma 5.1.4 that kG is local with residue field k, hence, all simple modules are isomorphic (as kG-modules) to k, endowed with trivial G-action. It follows that the semisimple modules are isomorphic to k^n for some

 $n \in \mathbb{N}$ since all modules are of finite length in our setting, i.e. they correspond to finite-dimensional k-modules.

Since the subcategory of kG-mod of semisimple modules is equivalent to k-mod, the statement follows directly from [QSS79, Theorem 6.10] (and [QSS79, Theorem 6.9] for Witt groups).

After having seen how to compute the (Grothendieck–)Witt group of kG- stab in terms of the (Grothendieck–)Witt group of kG- mod, we have enough tools at hand to determine the first non-algebro-geometric example of a Chow–Witt group, namely of the tensor triangulated category kG- stab for $G = \mathbb{Z}/p^n\mathbb{Z}$.

5.3 Tensor Triangular Chow–Witt Groups for $G = \mathbb{Z}/p^n\mathbb{Z}$

In order to compute the Chow–Witt group of the rigid tensor triangulated category kG-stab, consider the following setup.

Convention 5.3.1. In this section, we fix $n \in \mathbb{N}$ and $p \neq 2$ prime. Let k be a field of characteristic p and $G = \mathbb{Z}/p^n\mathbb{Z}$. When considering the tensor triangulated category kG – stab a triangulated category with duality, we fix the duality $\operatorname{Hom}_k(-,k)$ and restrictions of it on subcategories. Moreover, we equip kG – stab with the dimension function $\dim_{\operatorname{Krull}}$ (Definition 5.1.24).

We begin with determining the filtration components kG-stab_(i) of kG-stab introduced in Definition 2.3.4. Recall that Proposition 5.1.27 tells us that it suffices to look at support varieties.

By [Car96, Theorem 7.3] we know that $\mathcal{V}_G(k)$ is a point, hence, so is $\mathcal{V}_G(M)$ for any non-projective kG-module M, i.e. $\dim_{\mathrm{Krull}}(\mathcal{V}_G(M)) = 0$ for all $M \neq 0$ in kGstab. Under the isomorphism $\mathcal{V}_G(k) \cong \mathrm{Spc}(kG$ -stab) from Proposition 5.1.27, the support of an object in kG-stab corresponds to its projective support variety. In particular, they have equal dimension. It follows that

$$kG\operatorname{-stab}_{(i)} = \begin{cases} kG\operatorname{-stab} & i \ge 0, \\ 0 & i < 0, \end{cases}$$

and thus

$$kG\operatorname{-stab}_{(i)}/kG\operatorname{-stab}_{(n-1)} = \begin{cases} kG\operatorname{-stab} & i = 0, \\ 0 & \text{else}, \end{cases}$$

compare also [Kl16a, Section 6.3]. In particular, all quotients of filtration components are idempotent complete, which is why we can apply Definition 4.3.5. The only non-trivial degree of $\widetilde{CH}_i^{\Delta}(kG\text{-stab})$ is i = 0. For this case, we obtain the following defining diagram for $\mathcal{T} := kG$ -stab.

which, by the above considerations, equals

$$\begin{array}{ccc} 0 & \stackrel{0}{\longrightarrow} 0 & & \operatorname{GW}(kG\operatorname{-\operatorname{stab}}) & \stackrel{\operatorname{id}}{\longrightarrow} \operatorname{GW}(kG\operatorname{-\operatorname{stab}}) \\ & & & & & \\ & & & & \\ & & & & \\ & & & & & & \\ & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & &$$

So, we obtain

$$\widetilde{\mathrm{CH}}_{\Delta}^{0}(kG\operatorname{-\operatorname{stab}}) = (0_{d})^{-1}(\mathrm{Im}(0))/\operatorname{id}(\ker(\operatorname{id}))$$
$$= \mathrm{GW}(kG\operatorname{-\operatorname{stab}})/0$$
$$= \mathrm{GW}(kG\operatorname{-\operatorname{stab}}).$$

As a consequence, the computation of $\widetilde{\operatorname{CH}}^0_\Delta(kG\operatorname{-stab})$ amounts to the computation of $\operatorname{GW}(kG\operatorname{-stab})$. For this, we observe that kG is local with residue field k by Example 5.1.6. So, we use Theorem 5.2.3 and Proposition 5.2.6 to obtain

$$GW(kG\text{-stab}) \cong GW(kG\text{-mod}) / \operatorname{Im}(\alpha)$$
$$\cong GW(k) / U(\operatorname{Im}(\alpha))$$

where $U : \operatorname{GW}(kG \operatorname{-mod}) \to \operatorname{GW}(k \operatorname{-mod})$ is the isomorphism from Proposition 5.2.6 and $\alpha : \operatorname{GW}(kG \operatorname{-proj}) \to \operatorname{GW}(kG \operatorname{-mod})$ is induced by the inclusion. Abusing notation, we also write U for the composition $(\dim, \det) \circ U$.

We have proven the following.

Proposition 5.3.2. Let $G = \mathbb{Z}/p^n\mathbb{Z}$, k a field of prime characteristic $p \neq 2$ and $n \in \mathbb{N}$ and U, α as above. Then, the tensor triangular Chow-Witt groups of kG-stab are given as

$$\widetilde{\mathrm{CH}}^{i}_{\Delta}(kG\operatorname{-stab}) \cong \begin{cases} \mathrm{GW}(k)/U(\mathrm{Im}(\alpha)) & i = 0, \\ 0 & i \neq 0. \end{cases}$$

We can even be more concrete than this general result. If k is algebraically closed, Chow–Witt groups and Chow groups coincide as expected.

Example 5.3.3. For algebraically closed fields k, the map dim : $GW(k) \to \mathbb{Z}$ is an isomorphism by [Lam05, Proposition II.3.1], hence

$$\widetilde{\mathrm{CH}}^{i}_{\Delta}(kG\operatorname{-stab}) \cong \begin{cases} \mathbb{Z}/p^{n}\mathbb{Z} & i = 0, \\ 0 & i \neq 0 \end{cases}$$

since kG is local, i.e. projective and free modules coincide. Thus, symmetric forms in Im(α) have k-dimension a multiple of p^n . It follows from [Kl16a, Proposition 6.3.2] that

$$\widetilde{\operatorname{CH}}^{i}_{\Delta}(kG\operatorname{-\operatorname{stab}}) \cong \operatorname{CH}^{i}_{\Delta}(kG\operatorname{-\operatorname{stab}})$$

for all i.

For the next example, we restrict to the case of k being finite, since GW(k) is known for this special case. Again, it turns out that Chow–Witt and Chow groups coincide here.

Example 5.3.4. Let k be finite. We know by Proposition 5.2.6 that

$$\operatorname{GW}(kG\operatorname{-}\operatorname{mod})\cong \operatorname{GW}(k).$$

By [Scha85, Chapter 2, Theorem 3.3 and Lemma 3.7] (and the considerations of [Scha85, Chapter 2, §2,]) we have the isomorphism

$$(\dim, \det) : \mathrm{GW}(k) \xrightarrow{\cong} \mathbb{Z} \oplus \mathbb{Z}/2\mathbb{Z}$$

given by the dimension and the determinant. Note that [Scha85, Chapter 2, Theorem 3.3] can be applied since, by [Scha85, Chapter 2, Theorem 3.8], every 3-dimensional quadratic space is isotropic (in the sense of [Scha85]). By Theorem 5.2.3 and Proposition 5.2.6 we hence have

$$GW(kG\operatorname{-stab}) \cong GW(kG\operatorname{-mod}) / \operatorname{Im}(\alpha)$$
$$\cong GW(k) / U(\operatorname{Im}(\alpha))$$
$$\cong (\mathbb{Z} \oplus \mathbb{Z}/2\mathbb{Z}) / U(\operatorname{Im}(\alpha))$$

for $U : \mathrm{GW}(kG \operatorname{-mod}) \to \mathrm{GW}(k \operatorname{-mod})$ and $\alpha : \mathrm{GW}(kG \operatorname{-proj}) \to \mathrm{GW}(kG \operatorname{-mod})$ as before.

The subgroup $\text{Im}(\alpha)$ is generated by equivalence classes of symmetric objects (P, ϕ) , where P is projective. Since kG is local by Lemma 5.1.4, all projective modules are free, and hence have k-dimension a multiple of p^n . Moreover, classes of objects of the form (P, ϕ) can have arbitrary determinant in each dimension $m \cdot p^n$, $m \in \mathbb{Z}$, hence, the map (dim, det) induces a group isomorphism

$$U(\operatorname{Im}(\alpha)) \cong p^n \mathbb{Z} \oplus \mathbb{Z}/2\mathbb{Z}.$$

In the end, we have

$$GW(kG\text{-stab}) \cong \mathbb{Z} \oplus \mathbb{Z}/2\mathbb{Z}/U(\operatorname{Im}(\alpha))$$
$$\cong (\mathbb{Z} \oplus \mathbb{Z}/2\mathbb{Z})/(p^n\mathbb{Z} \oplus \mathbb{Z}/2\mathbb{Z})$$
$$\cong \mathbb{Z}/p^n\mathbb{Z}.$$

Hence,

$$\widetilde{\mathrm{CH}}^{i}_{\Delta}(kG\operatorname{-stab}) \cong \begin{cases} \mathbb{Z}/p^{n}\mathbb{Z} & i = 0, \\ 0 & i \neq 0. \end{cases}$$

Again, it follows from [Kl16a, Proposition 6.3.2] that

$$\widetilde{\operatorname{CH}}^{i}_{\Delta}(kG\operatorname{-\operatorname{stab}})\cong \operatorname{CH}^{i}_{\Delta}(kG\operatorname{-\operatorname{stab}})$$

for all i.

The reader might find it surprising that $\widetilde{\operatorname{CH}}_{\Delta}^{i}(kG\operatorname{-stab}) \cong \operatorname{CH}_{\Delta}^{i}(kG\operatorname{-stab})$ not only for k algebraically closed but also for finite k as we have seen above. What actually happens when dividing out $\operatorname{Im}(\alpha)$ is that we divide out such a big subgroup that symmetric objects of the same dimension become equivalent although they have different determinants, i.e. in particular are not isometric in $\operatorname{GW}(k)$. To get a better intuition of this circumstance, we consider the case $k = \mathbb{F}_3$ and $G = \mathbb{Z}/3\mathbb{Z}$ in the following.

Example 5.3.5. Let $k = \mathbb{F}_3$ and $G = \mathbb{Z}/3\mathbb{Z}$, i.e. p = 3 and n = 1. Since $\mathrm{GW}(k) \cong \mathbb{Z} \oplus \mathbb{Z}/2\mathbb{Z}$, each element in $\mathrm{GW}(k)$ is uniquely determined (up to isometry) by its dimension together with its determinant, where the determinant is an element of $k^{\times}/(k^{\times})^2 = \{[1], [-1]\}$ since -1 is not a square in \mathbb{F}_3 .

Dividing out $U(\text{Im}(\alpha))$, symmetric objects of equal dimension now are in the same equivalence class. We check dimensions 1 and 2:

We have $\langle 1 \rangle \sim \langle -1 \rangle$ since

$$\langle 1 \rangle + \langle -1, 1, 1 \rangle = \langle -1, 1, 1, 1 \rangle = \langle -1 \rangle + \langle 1, 1, 1 \rangle$$

with $\langle 1, 1, 1 \rangle$, $\langle -1, 1, 1 \rangle$ in $U(\operatorname{Im}(\alpha))$.

In dimension 2, we have

$$\langle 1,1\rangle + \langle -1,1,1\rangle = \langle -1,1,1,1,1\rangle = \langle -1,1\rangle + \langle 1,1,1\rangle$$

with $\langle -1, 1, 1 \rangle$, $\langle 1, 1, 1 \rangle \in U(\text{Im}(\alpha))$. Analogous considerations can be done for different values of p^n for $p \neq 2$.

Summarizing these results, we have the following.

Proposition 5.3.6. If $G = \mathbb{Z}/p^n\mathbb{Z}$, and k is either algebraically closed or finite of characteristic $p \neq 2$, we have that

$$\widetilde{\mathrm{CH}}^{i}_{\Delta}(kG\operatorname{-stab}) \cong \mathrm{CH}^{i}_{\Delta}(kG\operatorname{-stab}) \cong \begin{cases} \mathbb{Z}/p^{n}\mathbb{Z} & i = 0, \\ 0 & i \neq 0. \end{cases}$$

In particular, $GW(kG\operatorname{-stab}) \cong K_0(kG\operatorname{-stab}) \cong \mathbb{Z}/p^n\mathbb{Z}$.

Proof. If $G = \mathbb{Z}/p^n\mathbb{Z}$, the first statement follows directly from Example 5.3.3 and Example 5.3.4. The second statement follows from

$$\operatorname{GW}(kG\operatorname{-stab}) = \widetilde{\operatorname{CH}}^0_{\Delta}(kG\operatorname{-stab}) \cong \operatorname{CH}^0_{\Delta}(kG\operatorname{-stab}) = K_0(kG\operatorname{-stab})$$

by Proposition 5.3.2 and [Kl16a, Section 6.3].

We have just seen examples of tensor triangular Chow–Witt groups in the setting of representation theory. For $G = \mathbb{Z}/p^n\mathbb{Z}$, we obtained a general result, but also witnessed that, for k algebraically closed or k finite, Chow–Witt and Chow groups of kG-stab coincide.

In the next section, we will consider the "next complicated" case, namely $G = \mathbb{Z}/p\mathbb{Z} \times \mathbb{Z}/p\mathbb{Z}$.

5.4 Tensor Triangular Chow Groups for $G = \mathbb{Z}/p\mathbb{Z} \times \mathbb{Z}/p\mathbb{Z}$

We are in the situation of Convention 5.3.1, but with $G = \mathbb{Z}/p\mathbb{Z} \times \mathbb{Z}/p\mathbb{Z}$.

In [Kl16a], Klein computes the Chow groups for $G = \mathbb{Z}/2\mathbb{Z} \times \mathbb{Z}/2\mathbb{Z}$ and char(k) = 2, but since we work with symmetric bilinear forms, characteristic 2 is usually a problem as mentioned in Remark 3.1.4. For this reason, we consider fields k of characteristic $p \neq 2$ and $G = \mathbb{Z}/p\mathbb{Z} \times \mathbb{Z}/p\mathbb{Z}$.

Remark 5.4.1. Note that there has been progress on the matter of characteristic 2 for Grothendieck–Witt and Witt groups. For example, works by Schlichting which we use here such as [Sch10a] do not make any assumptions on the characteristic. However, some of the tools we use do not have an analog for characteristic 2 yet. Since the focus of this work is merely to see some basic examples, we will not consider this case.

To get a sensible intuition of what the Chow–Witt groups of kG-stab could possibly look like, we start with computing its tensor triangular Chow groups following [Kl16a] (here for $\mathbb{Z}/p\mathbb{Z} \times \mathbb{Z}/p\mathbb{Z}, p \neq 2$ as opposed to $\mathbb{Z}/2\mathbb{Z} \times \mathbb{Z}/2\mathbb{Z}$).

As for $G = \mathbb{Z}/p^n\mathbb{Z}$, we first determine the filtration components of the tensor triangulated category kG-stab. By [Car96, Thm 7.6] (compare [BBC09, Example 1.2]), it follows that $\mathcal{V}_G(k) = \mathbb{P}^1$, i.e. the modules of complexity ≤ 1 yield a proper subcategory kG-stab₍₀₎ of kG-stab. The filtration components of kG-stab thus are

$$kG-\operatorname{stab}_{(i)} := \begin{cases} kG-\operatorname{stab} & i \ge 1, \\ kG-\operatorname{stab}_{(0)} & i = 0, \\ 0 & i < 0, \end{cases}$$

and the only non-trivial quotients are

$$kG$$
- $\operatorname{stab}_{(0)}/kG$ - $\operatorname{stab}_{(-1)} = kG$ - $\operatorname{stab}_{(0)}$

and

$$kG\operatorname{-stab}_{(1)}/kG\operatorname{-stab}_{(0)} = kG\operatorname{-stab}/kG\operatorname{-stab}_{(0)}.$$

Hence, we distinguish the following non-trivial degrees i for $CH_i^{\Delta}(kG\operatorname{-stab})$.

 $\frac{i=0}{\text{First note that}}$

$$Z_0^{\Delta}(kG\operatorname{-stab}) := K_0((kG\operatorname{-stab}_{(0)}/kG\operatorname{stab}_{(-1)})^{\natural})$$

= $K_0((kG\operatorname{-stab}_{(0)})^{\natural}) = K_0(kG\operatorname{-stab}_{(0)})$

since kG-stab₍₀₎ is idempotent complete as a thick subcategory of an idempotent complete category. It follows that

$$\operatorname{CH}^{0}_{\Delta}(kG\operatorname{-\operatorname{stab}}) := Z^{0}_{\Delta}(kG\operatorname{-\operatorname{stab}})/j^{0} \circ q^{0}(\ker(i^{1})) = K_{0}(kG\operatorname{-\operatorname{stab}}_{(0)})/\ker(i^{1}),$$

for $i^1 : K_0(kG\operatorname{-stab}_{(0)}) \to K_0(kG\operatorname{-stab})$ and by the isomorphism theorem of abelian groups

$$\operatorname{CH}^{0}_{\Delta}(kG\operatorname{-stab}) = K_{0}(kG\operatorname{-stab}_{(0)})/\operatorname{ker}(i^{1}) \cong \operatorname{Im}(i^{1}).$$

 $\underline{i=1}$

We do not know whether the quotient kG-stab/kG-stab $_{(0)}$ is idempotent complete, which was a condition until now. So, we use the model-theoretic description of tensor triangular Chow groups introduced in Definition 4.6.2. It is applicable since kG-stab admits a model by [Kl16b, Example 2.31].

The coboundaries (whose definition does not differ from Balmer's definition) are defined as

$$B^1_{\Delta}(kG\operatorname{-stab}) = j^1 \circ q^1(\ker(i^2)),$$

where

$$i^2: K_0(kG\operatorname{-stab}_{(1)}) \to K_0(kG\operatorname{-stab}_{(2)})$$

is the identity on $K_0(kG\operatorname{-stab})$ since $kG\operatorname{-stab}_{(i)} = kG\operatorname{-stab}$ for $i \geq 1$. Hence,

$$B^{1}_{\Delta}(kG\text{-}\operatorname{stab}) = j^{1} \circ q^{1}(\ker(i^{2})) = j^{1} \circ q^{1}(0) = 0.$$

It follows that

$$CH^{1}_{\Delta'}(kG\operatorname{-stab}) := Z^{1}_{\Delta'}(kG\operatorname{-stab})/B^{1}_{\Delta}(kG\operatorname{-stab}) = Z^{1}_{\Delta'}(kG\operatorname{-stab}).$$

By Definition 4.6.2,

$$Z^{1}_{\Delta'}(kG\operatorname{-stab}) := \ker(q'_{0} \circ d'_{1}) \subset K_{0}((kG\operatorname{-stab}/kG\operatorname{-stab}_{(0)})^{\natural})$$

with

$$d'_1: K_0((kG\operatorname{-stab}/kG\operatorname{-stab}_{(0)})^{\sharp}) \to \mathbb{K}_{-1}(kG\operatorname{-stab}_{(0)})$$

and

$$q_1': \mathbb{K}_{-1}(kG\operatorname{-stab}_{(0)}) \to \mathbb{K}_{-1}(kG\operatorname{-stab}_{(0)}/kG\operatorname{-stab}_{(-1)}) \\ \to \mathbb{K}_{-1}((kG\operatorname{-stab}_{(0)}/kG\operatorname{-stab}_{(-1)})^{\natural}).$$

Since kG-stab₍₋₁₎ = 0, the map q'_1 is the identity on $\mathbb{K}_{-1}(kG$ -stab₍₀₎). Consequently,

$$CH^{1}_{\Delta'}(kG\operatorname{-stab}) = Z^{1}_{\Delta'}(kG\operatorname{-stab}) = \ker(d'_{1}).$$

In conclusion, we have shown:

Proposition 5.4.2. Let $G = \mathbb{Z}/p\mathbb{Z} \times \mathbb{Z}/p\mathbb{Z}$ and k a field of prime characteristic $p \neq 2$ and $n \in \mathbb{N}$. Then, the tensor triangular Chow groups of kG-stab are

$$CH^{i}_{\Delta}(kG\text{-}stab) = \begin{cases} 0 & i \neq 0, 1\\ Im(i^{1}) & i = 0,\\ ker(d'_{1}) & i = 1, \end{cases}$$

where $i^1 : K_0(kG\operatorname{-stab}_{(0)}) \to K_0(kG\operatorname{-stab})$ and $d'_1 : K_0((kG\operatorname{-stab}/kG\operatorname{-stab}_{(0)})^{\natural}) \to \mathbb{K}_{-1}(kG\operatorname{-stab}_{(0)}).$

For i = 0, it amounts to computing the image of i^1 .

Remark 5.4.3. For the case p = 2, Klein does so in [Kl16a, Lemma 6.4.4]. He uses a complete characterization of the indecomposable modules of complexity 1 as the non-projective indecomposable modules of even dimension, see [Kl16a, Corollary 6.4.3]. At this point, concrete results about p = 2 are applied, which is why we use different arguments here.

Recall that, by Example 5.1.6, the group algebra of kG for $G = \mathbb{Z}/p\mathbb{Z} \times \mathbb{Z}/p\mathbb{Z}$ is given by

$$k(\mathbb{Z}/p\mathbb{Z} \times \mathbb{Z}/p\mathbb{Z}) \cong k[x, y]/(x^p, y^p).$$

We know from [Ben98b, Corollary 5.8.5] that if p does not divide the dimension of a module, then its support variety already is $\mathcal{V}_G(k)$, which in our case equals \mathbb{P}^1 . Hence, if p does not divide the dimension of a module, then its complexity is 2.

Example 5.4.4. Let p = 3. We want to give an example of an indecomposable 3-dimensional module of complexity 1. One can extend this example to p > 3 since this is a general procedure.

Let J = (x, y) be the radical of kG, then $J^2 = (x^2, y^2, xy)$. We consider the indecomposable module

$$M = kG/J^2 = k[x, y]/(x^2, y^2, xy).$$

It is 3-dimensional with k-generators 1, x and y. To see that it has complexity 1, we construct a minimal resolution.

We define $f_0: kG \to M$ as the canonical projection which has kernel ker $(f_0) = (x^2, y^2, xy)$ as a kG-module. Next, we construct

$$f_1 : kG^3 \to kG, (1,0,0) \mapsto x^2, (0,1,0) \mapsto y^2, (0,0,1) \mapsto xy,$$

such that $\ker(f_0) = \operatorname{Im}(f_1)$. We obtain the description of its kernel as $\ker(f_1) = ((x, 0, 0), (0, y, 0), (0, 0, x^2), (0, 0, y^2))$ and construct a map

$$f_2: kG^4 \to kG^3, (1,0,0,0) \mapsto (x,0,0), (0,1,0,0) \mapsto (0,y,0), (0,0,1,0) \mapsto (0,0,x^2), (0,0,0,1) \mapsto (0,0,y^2),$$

with $\text{Im}(f_2) = \text{ker}(f_1)$ and $\text{ker}(f_2) = ((x^2, 0, 0, 0), (0, y^2, 0, 0), (0, 0, x, 0), (0, 0, 0, y))$. We continue inductively by always sending the generators of kG^4 to the generators of the kernel from the step before. This way, we obtain that all f_i for $i \ge 3$ odd and for $i \ge 4$, respectively, are equal and we have a minimal resolution

$$\dots \to kG^4 \xrightarrow{f_3} kG^4 \xrightarrow{f_2} kG^3 \xrightarrow{f_1} kG \xrightarrow{f_0} M \to 0.$$

This procedure is described in [Car96, p.14]; the resolution is minimal by [Car96, Theorem 4.1].

Now, let $\lambda := 5$. We observe that $\dim(kG^3) = 3 \leq 5 \cdot 2^{1-1} = 5$ and $\dim(kG^4) = 4 \leq 5 \cdot i^{1-1}$ where $i \geq 3$ is the index of kG^4 in the free resolution. Hence, the complexity of M is indeed 1 whereas its dimension is 3, i.e. in particular divisible by 3.

On the other hand, the module $k \oplus k \oplus k$ is three-dimensional, but has complexity 2 by Proposition 5.1.30 and [Ben98b, Theorem 5.1.1 (ii)].

As we have just seen, there are modules of dimension 3 with different complexity.

However, for general p, such modules of the same dimension but different complexity are in the same equivalence class in $K_0(kG \operatorname{-mod}) \stackrel{\text{dim}}{\cong} \mathbb{Z}$. Consequently, we can still compute $\operatorname{Im}(i^1)$. By the same arguments as in the case $G = \mathbb{Z}/p^n\mathbb{Z}$, we know that the dimension yields an isomorphism

$$\overline{\dim}_k : K_0(kG\operatorname{-stab}) \xrightarrow{\cong} \mathbb{Z}/p^2\mathbb{Z}.$$

Restricting this isomorphism to $\text{Im}(i^1) \subset K_0(kG\text{-stab})$, we obtain an injective group homomorphism

$$\overline{\dim}_k : \operatorname{Im}(i^1) \to \mathbb{Z}/p^2\mathbb{Z}.$$

Let A be the subgroup of $K_0(kG\text{-stab}) \cong \mathbb{Z}/p^2\mathbb{Z}$ generated by modules of dimension divisible by p. We have that

$$A \cong p(\mathbb{Z}/p^2\mathbb{Z}) \cong \mathbb{Z}/p\mathbb{Z}$$

via dimension. Modules of complexity 1, i.e. modules in kG-mod₍₀₎, all have dimension divisible by p by contraposition of [Ben98b, Corollary 5.8.5] and the fact that $\mathcal{V}_G(k) = \mathbb{P}^1$. Thus, $\operatorname{Im}(i^1) \subset A$. On the other hand, there is a kG-module of dimension p and complexity 1, namely kG/J^2 for J = (x, y) by Example 5.4.4. Since $1 = c_G(kG/J^2) = c_G((kG/J^2)^n)$ by [Ben98b, Theorem 5.1.1 (ii)], the composition

$$\operatorname{Im}(i^1) \xrightarrow{\cong} A \xrightarrow{\cong} \mathbb{Z}/p\mathbb{Z}$$

is also surjective. It follows that $\operatorname{Im}(i^1) \cong \mathbb{Z}/p\mathbb{Z}$.

Applying these computations to Proposition 5.4.2, we can now deduce the following.

Proposition 5.4.5. Let $G = \mathbb{Z}/p\mathbb{Z} \times \mathbb{Z}/p\mathbb{Z}$ and k a field with char(k) = $p \neq 2$. Then,

$$CH^{i}_{\Delta}(kG\text{-stab}) = \begin{cases} 0 & i \neq 0, 1, \\ \mathbb{Z}/p\mathbb{Z} & i = 0, \\ \ker(d'_{1}) & i = 1, \end{cases}$$

where $d'_1: K_0((kG\operatorname{-stab}/kG\operatorname{-stab}_{(0)})^{\natural}) \to \mathbb{K}_{-1}(kG\operatorname{-stab}_{(0)}).$

For i = 1, we will not proceed further since it would exceed the extent of this work.

5.5 Tensor Triangular Chow–Witt Groups for $G = \mathbb{Z}/p\mathbb{Z} \times \mathbb{Z}/p\mathbb{Z}$

Again, we are in the situation of Convention 5.3.1, but with $G = \mathbb{Z}/p\mathbb{Z} \times \mathbb{Z}/p\mathbb{Z}$.

We would like to take a closer look at the defining diagram for Chow–Witt groups of kG-stab for $G = \mathbb{Z}/p\mathbb{Z} \times \mathbb{Z}/p\mathbb{Z}$, at first for general k of characteristic $p \neq 2$.

We have seen the filtration components above, so we also consider the following cases where $\mathcal{T} := kG$ -stab. Since we do not know in general whether kG-stab/kG-stab₍₀₎ is idempotent complete, the defining diagram is considered on the level of models as in Definition 4.6.4. $\underline{i=0}$

Since kG-stab₍₀₎ is idempotent complete as a thick subcategory of an idempotent complete category, the defining diagram has the form

We obtain

$$\widetilde{\mathrm{CH}}_{\Delta}^{0}(kG\operatorname{-stab}) = 0_{d}^{-1}(\mathrm{Im}(0))/(\mathrm{id}(\mathrm{ker}(\tilde{i})))$$
$$= \mathrm{GW}(kG\operatorname{-stab}_{(0)})/\operatorname{ker}(\tilde{i})$$
$$\cong \mathrm{Im}(\tilde{i}),$$

where the last isomorphism follows from the isomorphism theorem for abelian groups.

 $\underline{i=1}$

As in the case of tensor triangular Chow groups, we use the model-theoretic version Definition 4.6.4 since we do not know whether kG-stab/kG-stab $_{(0)}$ is idempotent-complete.

The oriented coboundaries are defined as

$$\widetilde{B}^1_{\Delta'}(kG\operatorname{-stab}) := \widetilde{q}'_1(\ker(\widetilde{i}^2)),$$

where

$$\tilde{i}^2 : \mathrm{GW}^{[1]}(kG\operatorname{-stab}_{(1)}) \to \mathrm{GW}^{[1]}(kG\operatorname{-stab}_{(2)})$$

is the identity on $GW^{[1]}(kG\operatorname{-stab})$ since $kG\operatorname{-stab}_{(i)} = kG\operatorname{-stab}$ for $i \geq 1$. Hence,

$$\widetilde{B}^1_{\Delta'}(kG\operatorname{-stab}) = \widetilde{q}'_1(\ker(\widetilde{i}^2)) = \widetilde{q}'_1(0) = 0.$$

It follows that

$$\widetilde{\operatorname{CH}}^{1}_{\Delta'}(kG\operatorname{-\operatorname{stab}}) := \widetilde{Z}^{1}_{\Delta'}(kG\operatorname{-\operatorname{stab}})/\widetilde{B}^{1}_{\Delta'}(kG\operatorname{-\operatorname{stab}}) = \widetilde{Z}^{1}_{\Delta'}(kG\operatorname{-\operatorname{stab}}).$$

By Definition 4.6.4,

$$\widetilde{Z}^1_{\Delta'}(kG\operatorname{\!-stab}) = \ker(\widetilde{q}'_{0,\mathrm{W}} \circ \widetilde{d}'_1) \subset \mathrm{GW}^{[1]}((kG\operatorname{\!-stab}/kG\operatorname{\!-stab}_{(0)})^{\natural})$$

with

$$\tilde{d}'_1 : \mathrm{GW}^{[1]}((kG\operatorname{-stab}/kG\operatorname{-stab}_{(0)})^{\natural}) \to \mathrm{W}^{[2]}(kG\operatorname{-stab}_{(0)})$$

and

$$\begin{split} \tilde{q}'_{0,W} &: W^{[2]}(kG\operatorname{-stab}_{(0)}) \to W^{[2]}(kG\operatorname{-stab}_{(0)}/kG\operatorname{-stab}_{(-1)}) \\ &\to W^{[2]}((kG\operatorname{-stab}_{(0)}/kG\operatorname{-stab}_{(-1)})^{\natural}). \end{split}$$

Since kG-stab₍₋₁₎ = 0, the map $\tilde{q}'_{0,W}$ is the identity on W^[2](kG-stab₍₀₎). Consequently,

$$\widetilde{\operatorname{CH}}_{\Delta'}^1(kG\operatorname{-\operatorname{stab}}) = \widetilde{Z}_{\Delta'}^1(kG\operatorname{-\operatorname{stab}}) = \ker(\widetilde{d}'_1).$$

We have thus proven the following

Proposition 5.5.1. For $G = \mathbb{Z}/p\mathbb{Z} \times \mathbb{Z}/p\mathbb{Z}$ and k a field of characterisite $p \neq 2$, we have isomorphisms

$$\widetilde{\mathrm{CH}}^{i}_{\Delta'}(kG\operatorname{\!-stab}) = \begin{cases} 0 & i \neq 0, 1, \\ \mathrm{Im}(\tilde{i}) & i = 0, \\ \ker(\tilde{d}'_{1}) & i = 1 \end{cases}$$

for \tilde{i} : GW(kG-stab₍₀₎) \rightarrow GW(kG-stab) and \tilde{d}'_1 : GW^[1]((kG-stab/kG-stab₍₀₎)^{\natural}) \rightarrow W^[2](kG-stab₍₀₎).

For more concrete cases of k one can say a bit more, but we can only give partial results here.

Example 5.5.2. For k be algebraically closed, let

$$U : \mathrm{GW}(kG \operatorname{-} \mathrm{mod}) \to \mathrm{GW}(k \operatorname{-} \mathrm{mod})$$

be the morphism induced by the forgetful functor and

$$\alpha : \mathrm{GW}(kG\operatorname{-proj}) \to \mathrm{GW}(kG\operatorname{-mod})$$

the morphism induced by the inclusion. To compute the Chow–Witt group in degree 0, we consider the inclusion

$$i : \mathrm{GW}(kG\operatorname{-stab}_{(0)}) \to \mathrm{GW}(kG\operatorname{-stab})$$

and observe that

$$\operatorname{GW}(kG\operatorname{-stab}) \cong \operatorname{GW}(kG\operatorname{-mod})/\operatorname{Im}(\alpha) \cong \operatorname{GW}(k)/U(\operatorname{Im}(\alpha)) \cong \mathbb{Z}/p^2\mathbb{Z}$$

by Theorem 5.2.3 and Proposition 5.2.6; recall that by Example 5.1.6 the residue field of kG is k. The last isomorphism follows from the dimension isomorphism $GW(k) \cong \mathbb{Z}$ and the fact that kG is local (see Lemma 5.1.4), hence, free and projective modules coincide. So, their dimension is always a multiple of p^2 and a kG-module of dimension p^2 exists, namely kG itself.

As in the case of Chow groups, the image Im(i) is now spanned by modules of complexity 1. These always have dimension divisible by p by contraposition of [Ben98b, Corollary 5.8.5] and support variety \mathbb{P}^1 . There is indeed a kG-module of dimension p, namely kG/J^2 from Example 5.4.4. Analogously, it follows that the restriction of the isomorphism $\overline{\dim}_k : \text{GW}(kG\text{-stab}) \to \mathbb{Z}/p^2\mathbb{Z}$ to $\text{Im}(\tilde{i})$

$$\overline{\dim}_k : \operatorname{Im}(\tilde{i}) \to \mathbb{Z}/p^2\mathbb{Z}$$

is injective with image $\mathbb{Z}/p\mathbb{Z}$. It follows by Proposition 5.5.1, that

$$\widetilde{\operatorname{CH}}^0_{\Delta}(kG\operatorname{-stab})/\operatorname{Im}(\tilde{i}) \cong p(\mathbb{Z}/p^2\mathbb{Z}) \cong \mathbb{Z}/p\mathbb{Z}.$$

For the computation of kernel of \tilde{d}'_1 , or even $\mathrm{GW}^{[i]}(kG\operatorname{-stab})$ as a first approach, we lack the necessary tools at the moment and will not pursue this further here. Therefore, we have that for k algebraically closed

$$\widetilde{\mathrm{CH}}^{i}_{\Delta}(kG\operatorname{-stab}) = \begin{cases} 0 & i \neq 0, 1, \\ \mathbb{Z}/p\mathbb{Z} & i = 0, \\ \ker(\widetilde{d}'_{1}) & i = 1, \end{cases}$$

where $\tilde{d}'_1 : \mathrm{GW}^{[1]}((kG\operatorname{-stab}/kG\operatorname{-stab}_{(0)})^{\natural}) \to \mathrm{W}^{[2]}(kG\operatorname{-stab}_{(0)}).$

In degree 0, our computation coincides with the case of Chow groups. To compare degree 1, further steps will need to be made.

After having treated algebraically closed fields, we now want to consider finite fields.

Example 5.5.3. Let k be finite and U, α as above. Analogously to Example 5.3.4, we consider the following isomorphism

$$(\dim, \det) : \mathrm{GW}(k) \xrightarrow{\cong} \mathbb{Z} \oplus \mathbb{Z}/2\mathbb{Z}.$$

By Theorem 5.2.3 and Proposition 5.2.6 we, hence, have

$$GW(kG\text{-stab}) \cong GW(kG\text{-mod}) / \operatorname{Im}(\alpha)$$
$$\cong GW(k) / U(\operatorname{Im}(\alpha))$$
$$\cong (\mathbb{Z} \oplus \mathbb{Z}/2\mathbb{Z}) / U(\operatorname{Im}(\alpha))$$

Again, as in Example 5.3.4, kG is local by Lemma 5.1.4, all projective modules are free, hence, have a dimension a multiple of p^2 and arbitrary determinant in each dimension. It follows that $U(\text{Im}(\alpha)) \cong p^2 \mathbb{Z} \oplus \mathbb{Z}/2\mathbb{Z}$ and, thus,

$$GW(kG-\text{stab}) \cong (\mathbb{Z} \oplus \mathbb{Z}/2\mathbb{Z})/(p^2\mathbb{Z} \oplus \mathbb{Z}/2\mathbb{Z})$$
$$\cong \mathbb{Z}/p^2\mathbb{Z} \cong K_0(kG-\text{stab}).$$

To determine the image of \tilde{i} : $\mathrm{GW}(kG\operatorname{-stab}_{(0)}) \to \mathrm{GW}(kG\operatorname{-stab})$, recall that, since $\mathcal{V}_G(k) = \mathbb{P}^1$ and by contraposition of [Ben98b, Corollary 5.8.5], modules in the image of \tilde{i} have dimension a multiple of p^2 but arbitrary determinants. Thus, as above, $\mathrm{Im}(\tilde{i}) \cong p(\mathbb{Z}/p^2\mathbb{Z}) \cong p\mathbb{Z}/p^2\mathbb{Z} \cong \mathbb{Z}/p\mathbb{Z}$ and we obtain

$$\widetilde{\mathrm{CH}}^{i}_{\Delta'}(kG\operatorname{-stab}) = \begin{cases} 0 & i \neq 0, 1, \\ \mathbb{Z}/p\mathbb{Z} & i = 0, \\ \ker(\widetilde{d}'_{1}) & i = 1. \end{cases}$$

Again, Chow- and Chow-Witt groups coincide in degree 0 and further progress on degree 1 may be made in future work. As before, we lack the necessary technical tools to compute the first shifted Grothendieck–Witt groups and kernels or images of maps between them.

However, from the examples, we can deduce the following results. The restriction to the cases where G is an elementary abelian p-group or $G = \mathbb{Z}/p^n\mathbb{Z}$ is due to the assumptions in Proposition 5.2.6.

Corollary 5.5.4. If $G \cong (\mathbb{Z}/p\mathbb{Z})^r$ is a finite elementary abelian *p*-group and *k* is algebraically closed or finite of characteristic $p \neq 2$, we have that $GW(kG\operatorname{-stab}) \cong K_0(kG\operatorname{-stab})$.

Proof. Let U, α be as above. For G elementary abelian we have that

 $\operatorname{GW}(kG\operatorname{-stab}) \cong \operatorname{GW}(kG\operatorname{-mod})/\operatorname{Im}(\alpha) \cong \operatorname{GW}(k)/U(\operatorname{Im}(\alpha))$

by Theorem 5.2.3 and Proposition 5.2.6.

For k algebraically closed, $GW(k) \cong \mathbb{Z} \cong K_0(k)$ by dimension, and hence by [TaWa91, Proposition 1] GW(kG-stab) $\cong K_0(kG$ -stab).

If k is finite, $\mathrm{GW}(k) \cong \mathbb{Z} \oplus \mathbb{Z}/2\mathbb{Z}$ via dimension and determinant. Then, the subgroup $U(\mathrm{Im}(\alpha))$ is isomorphic to $p^r\mathbb{Z} \oplus \mathbb{Z}/2\mathbb{Z}$ analogously to Example 5.5.3. Hence, $\mathrm{GW}(kG\operatorname{-stab}) \cong \mathbb{Z}/p^r\mathbb{Z} \cong K_0(kG\operatorname{-stab})$ using again [TaWa91, Proposition 1], where both isomorphisms are given by the dimension. \Box

In this chapter, we have made some very first computations concerning Chow-Witt groups of tensor triangulated categories outside the field of algebraic geometry. For $G = \mathbb{Z}/p^n\mathbb{Z}$, $p \neq 2$, we have seen that the only non-trivial Chow-Witt group is in degree 0 and isomorphic to $\mathbb{Z}/p^n\mathbb{Z}$ and coincides with the non-oriented case. For $G = \mathbb{Z}/p\mathbb{Z} \times \mathbb{Z}/p\mathbb{Z}$ and $p \neq 2$, the 0-th Chow-Witt group again coincides with the Chow group being $\mathbb{Z}/p\mathbb{Z}$. The other non-trivial degree is 1, but we lack technical tools such as an analogon of Proposition 5.2.6 to do further steps here. Our computations moreover show that the Chow-Witt groups of kG-stab usually also heavily depends on the Grothendieck-Witt group of the base field k, which are hard to determine in general.

Chapter 6

The Spectrum of Artin Motives

After having seen how Chow–Witt groups of schemes can be generalized to the setting of tensor triangulated categories, we now want to investigate the role of Chow–Witt groups in the theory of motives. The first aim is to refine results of [BG23a] before computing some examples of Balmer spectra of Artin Milnor–Witt motives.

To begin with, this chapter recalls the construction of Voevodsky's derived category of motives DM(k; R) in Section 6.1 and arguments and results of [BG23a] in Section 6.2 connecting Artin motives, permutation modules, and Mackey functors.

Convention 6.0.1. Let R be a commutative ring. In the rest of this work, let k be a perfect field with a fixed separable closure \bar{k} and absolute Galois group $\Gamma := \text{Gal}(\bar{k}/k)$. If not mentioned otherwise, all schemes will be separated and of finite type over their respective base field (often k).

6.1 The Derived Category of Motives

We recall the construction of Voevodsky's derived category of motives DM(k; R) over a field k with coefficients in a ring R. As an introductory lecture on this topic, we recommend [MVW06, Voe00], which we use as references for the contents of this section. The latter is the original work.

Consider the following diagram demonstrating the construction setps. We start with the category of finite correspondences discussed below and consider presheaves on this category. After that, one passes to the derived category of tsheaves with transfers, usually for topologies such as Nisnevich, étale, or Zariski. In the next step, we force \mathbb{A}^1 -invariance to define the derived category of *effective motives*. Finally, we formally invert the Tate twist which is usually done by passing to spectrum objects, but we will not focus on this part here since we only deal with effective motives in this work.



We start with the construction of the category of finite correspondences.

Definition 6.1.1. Let X be a smooth connected scheme over k and Y any separated scheme over k. An *elementary correspondence* from X to Y is defined as an irreducible closed subset $W \subset X \times Y$ whose associated integral subscheme is finite and surjective over X. If X is non-connected, an elementary correspondence from X to Y is an elementary correspondence from a connected component of X to Y. A *finite correspondence* is an element of the free abelian group $\operatorname{Cor}_k(X, Y)$ generated by the elementary correspondences.

Definition 6.1.2. We define the category Cor_k of finite correspondences as follows. The objects are smooth separated schemes over k denoted by Sm_k . Its morphism sets are given by the abelian groups $\operatorname{Cor}_k(X,Y)$. If X is not connected with connected components X_i , $i \in I$, we have $\operatorname{Cor}_k(X,Y) = \bigoplus_{i \in I} \operatorname{Cor}_k(X_i,Y)$. Composition of correspondences is illustrated in [MVW06, Figure 17A.1.]. For a ring R, we set the category $\operatorname{Cor}_{k,R} := \operatorname{Cor}_k \otimes_{\mathbb{Z}} R$, i.e., tensoring every morphism group between two objects with R.

Remark 6.1.3. Note that $\operatorname{Cor}_{k,R}$ is a symmetric monoidal additive category with coproduct the disjoint union of k-schemes.

Example 6.1.4. Let $f : X \to Y$ be a morphism in Sm_k and X connected, then the graph Γ_f is an elementary correspondence from X to Y. If X is not connected, the sum of components of the graph yields a finite correspondence. We obtain a functor

$$\gamma: \mathrm{Sm}_k \to \mathrm{Cor}_k,$$

being the identity on objects and sending a morphism to its graph.

In the next step of the construction of DM(k; R), we pass to (pre-)sheaves with transfers.

Definition 6.1.5. A presheaf with transfers is an additive, *R*-linear contravariant functor $\operatorname{Cor}_{k,R}^{\operatorname{op}} \to R$ -Mod. The category of *R*-linear additive presheaves on $\operatorname{Cor}_{k,R}$ will be denoted by $\operatorname{PSh}_R(\operatorname{Cor}_{k,R})$.

A presheaf with transfers is a *t*-sheaf with transfers if it is a sheaf in the *t*-topology when restricted to Sm_k , where *t* can be Zariski (Zar), Nisnevich (Nis), or étale (et). We denote the corresponding category by $\text{Sh}_t(\text{Cor}_{k,R})$.

Example 6.1.6. Constant presheaves on Sm_k are presheaves with transfers. The classical Chow groups $\text{CH}^i(-)$ are presheaves with transfers, see [MVW06, Example 2.5].

An important class of presheaves with transfers will be the representable objects. They are essential to define motivic complexes and the Tate twist we later wish to invert.

Definition 6.1.7. Let X be an object of Sm_k , then $R_{\text{tr}}(X)$ is defined as the presheaf with transfers given by $R_{\text{tr}}(X)(U) := \text{Cor}_{k,R}(U,X)$, also denoted the presheaf with transfers represented by X.

Remark 6.1.8. By [Voe00, Lemma 3.1.2], $R_{tr}(X)$ is a Nis-sheaf with transfers.

Definition 6.1.9. Let $q \in \mathbb{N}$. Then, we define the *motivic complex* R(q) as the complex of presheaves with transfers given by

$$R(q) := C_* R_{\mathrm{tr}} \left(\mathbb{G}_m^{\wedge q} \right) [-q],$$

where \mathbb{G}_m is the multiplicative group and $C_*R_{\mathrm{tr}}(\mathbb{G}_m^{\wedge q}[-q])$ the complex obtained from the simplicial presheaf $U \mapsto R_{\mathrm{tr}}(\mathbb{G}_m^{\wedge q})[-q](U \times \Delta^{\bullet})$ and Δ^{\bullet} is the cosimplicial scheme defined as

$$\Delta^{n} := \operatorname{Spec}(k[x_{0}, ..., x_{n}]) / (\sum_{i=0}^{n} x_{i} = 1).$$

Example 6.1.10. We consider the following examples of motivic complexes.

- (i) For a general ring R we have $R(q) = \mathbb{Z}(q) \otimes R$.
- (ii) If q = 0, then $\mathbb{Z}(0) = C_*(\mathbb{Z})$, which is quasi-isomorphic to \mathbb{Z} .
- (iii) For q = 1 we have $\mathbb{Z}(1) = C_*\mathbb{Z}_{tr}(\mathbb{G}_m)[-1]$, which is quasi-isomorphic to $\mathcal{O}^*[-1]$ by [MVW06, Theorem 4.1].

In the next step, we force \mathbb{A}^1 -invariance to define effective motives.

Definition 6.1.11. We denote by $\mathcal{E}_{\mathbb{A}^1}$ the smallest thick (and closed under arbitrary direct sums) subcategory of $D(\operatorname{Sh}_{\operatorname{Nis}}(\operatorname{Cor}_{k,R}))$ containing maps of the form $R_{\operatorname{tr}}(X \times \mathbb{A}^1) \to R_{\operatorname{tr}}(X)$ and by $W_{\mathbb{A}^1}$ the class of maps with cone in $\mathcal{E}_{\mathbb{A}^1}$.

Then, we define the triangulated category of effective motives as the localization ${}^{}$

$$\mathrm{DM}^{\mathrm{eff}}(k;R) := \mathrm{D}(\mathrm{Sh}_{\mathrm{Nis}}(\mathrm{Cor}_{k,R}))[W_{\mathbb{A}^1}^{-1}],$$

which coincides with the Verdier quotient $D(Sh_{Nis}(Cor_{k,R}))/\mathcal{E}_{\mathbb{A}^1}$, compare [Nee05, Proposition 2.1.24].

By the motive M(X) of X in Sm_k we denote the object $R_{\text{tr}}(X)$ in $\text{DM}^{\text{eff}}(k; R)$.

Moreover, we define the triangulated category of geometric effective motives $DM^{\text{eff,gm}}(k; R)$ as the localization of the category $K_b(\text{Cor}_{k,R}^{\natural})$ with respect to the thick subcategory \bar{T} from [Voe00, p.191], i.e., forcing \mathbb{A}^1 -invariance.

Finally, we pass to the triangulated category of motives in the last step. Since we will only consider effective motives in the rest of this work, we will not go into detail. For this, the reader is referred to [MVW06, Voe00].

If t = N is, we omit it from the notation.

Definition 6.1.12. We define the triangulated category of motives DM(k; R) as the category obtained from $DM^{\text{eff}}(k; R)$ after formally inverting the Tate twist operation $\otimes R(1)$.

Similarly, we set the triangulated category of geometric motives $DM^{gm}(k; R)$ as the category obtained from $DM^{eff,gm}(k; R)$ after formally inverting the Tate twist operation. By abuse of notation, we denote by the motive M(X) of a smooth scheme X the class of $R_{tr}(X)$ in DM(k; R).

Remark 6.1.13. The category $\text{DM}^{\text{eff}}(k; R)$ is a tensor triangulated category. It is compactly generated by motives M(X) of smooth schemes. The category DM(k; R) is a tensor triangulated category in which every object is isomorphic to some M(X)(-n) for a smooth scheme X and some $n \ge 0$.

The geometric motives form a rigid tensor triangulated category $DM^{gm}(k; R)$ by [Voe00, Theorem 4.3.7]. It is moreover essentially small and idempotent complete, hence, so is $DM^{eff,gm}(k; R)$. Moreover, the category of geometric (effective) motives coincides with the thick subcategory of compact objects of $DM^{(eff)}(k; R)$.

Remark 6.1.14. The triangulated category of Tate motives is the localizing subcategory of DM(k; R) generated by Tate twists R(n) for $n \in \mathbb{Z}$ and is a large tensor triangulated category. The triangulated category of geometric Tate motives is the thick subcategory of $DM^{gm}(k; R)$ generated by Tate twists R(n) for $n \in \mathbb{Z}$. It is a small, rigid, idempotent complete tensor triangulated category.

After having recalled the construction of Voevodsky's derived category of motives, we will now recollect the most important results from [BG23a].

6.2 Artin Motives, Mackey Functors, and Permutation Modules

In this section, we recall constructions and main results from [BG23a] on the connection between Artin motives, permutation modules, and Mackey functors. The recollection is coarse and only included for the reader's convenience. For more details the reader is referred to the original work [BG23a], which we will follow here.

The aim of [BG23a] is to prove equivalences of tensor triangulated categories in the following diagram, where all the candidates will be recalled briefly.



A part of these equivalences already appears in [Voe00, Proposition 3.4.1]. We begin with the upper part of the triangle.

Definition 6.2.1. The derived category of Artin motives DAM(k; R) is defined as the localizing subcategory of $DM^{\text{eff}}(k; R)$ generated by the motives of 0-dimensional smooth k-schemes. Its compact part, the derived category of geometric Artin motives, is denoted by $DAM^{\text{gm}}(k; R)$. It can also be described as the thick subcategory of $DM^{\text{eff},\text{gm}}(k; R)$ generated by motives M(X) of zero-dimensional smooth k-schemes X.

Remark 6.2.2. The category DAM(k; R) is a tensor triangulated category which is compactly generated by motives M(X), where X is a 0-dimensional smooth scheme. Its compact part $DAM^{gm}(k; R)$ is an essentially small, idempotent complete, rigid tensor triangulated category by [HPS97, Theorem 2.1.3] (compare [Bal10b, Definition 44]). Moreover, one does not distinguish effective and noneffective (geometric) Artin motives by [BG23a, Remark 7.9].

Let $\operatorname{Cor}_{k,R}^0$ be the full subcategory of $\operatorname{Cor}_{k,R}$ spanned by the 0-dimensional smooth k-schemes. Balmer and Gallauer prove the following.

Proposition 6.2.3. ([BG23a, Corollary 7.10]) The inclusion $\iota : \operatorname{Cor}_{k,R}^0 \to \operatorname{Cor}_{k,R}$ induces an equivalence of tensor triangulated categories

 $\mathcal{D}(\operatorname{Sh}_{\operatorname{Nis}}(\operatorname{Cor}_{k,R}^0)) \xrightarrow{\simeq} \mathcal{D}\mathcal{AM}(k;R)$

that restricts to an equivalence on the level of compacts

 $\mathrm{K}_b((\mathrm{Cor}^0_{k,R})^{\natural}) \xrightarrow{\simeq} \mathrm{DAM}^{\mathrm{gm}}(k;R),$

where $(-)^{\natural}$ denotes the idempotent completion.

We now give the definition of permutation modules in the left corner of the triangle from Equation 6.1.

Let Γ be an arbitrary profinite group. Later, it will denote the absolute Galois group of k as in Convention 6.0.1.

In the category Γ – Sets, the objects are sets equipped with a continuous Γ action. Morphisms are simply Γ -equivariant maps. By Γ - sets we denote the full subcategory consisting of *finite* sets with a continuous Γ -action.

Moreover, we define the category $\operatorname{Mod}(\Gamma; R)$ with objects being *R*-modules endowed with the discrete topology and a continuous Γ -action. For X in Γ -Sets, we define R(X) to be the free *R*-module on the basis X. The Γ -action on X can be *R*-linearly extended to R(X). Hence, R(X) is an object in $\operatorname{Mod}(\Gamma; R)$. We obtain a functor

$$R(-): \Gamma - \text{Sets} \to \text{Mod}(\Gamma; R)$$

that is symmetric monoidal, where the tensor product of two modules over R is equipped with the diagonal Γ -action.

Note that by [BG23a, Proposition 2.6] the category $Mod(\Gamma; R)$ is *Grothendieck* abelian with $\{R(\Gamma/H) \mid H \leq \Gamma \text{ open subgroup}\}$ as a set of finitely presented generators.

Definition 6.2.4. We call modules in the essential image of R(-) permutation modules, the full subcategory of such is denoted by $Perm(\Gamma; R)$. Modules in the essential image of $R(-): \Gamma$ -sets $\rightarrow Mod(\Gamma; R)$ also yield a full subcategory which will be denoted by $Perm(\Gamma; R)$, the finitely generated permutation modules.

For Balmer's and Gallauer's definition of the derived category of permutation modules $D \operatorname{Perm}(\Gamma; R)$, the reader is referred to [BG23a, Definition 3.6]. Here, a Γ -acyclic complex is an object C in K $\operatorname{Perm}(\Gamma; R)$ such that the complex C^H of H-fixed points is acyclic for every open subgroup $H \leq \Gamma$. By [BG23a, Corollary 3.13], D $\operatorname{Perm}(\Gamma; R)$ is a tensor triangulated category.

For the right corner of the triangle from Equation 6.1, we need to introduce (cohomological) Mackey functors.

We will go into more detail when introducing the oriented counterpart in the following Construction 8.1.2. For a more concerete definition of span(Γ), the reader is referred to [ThWe95, Section 2].

Definition 6.2.5. ([ThWe95, Section 2]) We denote by span(Γ) the following category. Objects are Γ -sets and morphisms $X \to Y$ are given by isomorphisms classes of spans $X \leftarrow Z \to Y$, where $[X \leftarrow Z \to Y] = [X \leftarrow Z' \to Y]$ if there is an isomorphism $Z \cong Z'$ making the resulting diagrams commute. Composition is defined by pulling back.

Now, $\Omega(\Gamma)$ denotes the category obtained from span(Γ) by group completing the Hom sets, where the monoidal structure is given by the disjoint union. The resulting category is additive. By extending the scalars formally to an arbitrary ring R, we obtain an additive category denoted by $\Omega_R(\Gamma)$. **Definition 6.2.6.** An (*R*-linear) Mackey functor is an additive functor M: $\Omega(\Gamma)^{\text{op}} \to R$ -Mod. Furthermore, if M sends spans $\Gamma/H \leftarrow \Gamma/K \to \Gamma/H$ for every open subgroup $K \leq H$ to multiplication by [H:K], M is called a *cohomo*logical Mackey functor. We denote the resulting abelian categories by $\operatorname{Mack}_{R}(\Gamma)$ and $\operatorname{Mack}_{R}^{\operatorname{coh}}(\Gamma)$, respectively.

If we extend scalars to R formally, Mackey functors are defined as R-linear additive functors $M : \Omega_R(\Gamma)^{\text{op}} \to R$ -Mod, but we do not introduce a separate notation for this case.

Moreover, [BG23a, Proposition 4.17] tells us that we can view $\operatorname{perm}(\Gamma; R)$ as a quotient

$$\operatorname{perm}(\Gamma; R) \simeq \frac{\Omega_R(\Gamma)}{J_R(\Gamma)},$$

where $J_R(\Gamma)$ is the ideal generated by differences

$$(\Gamma/H \leftarrow \Gamma/K \to \Gamma/H) - [H:K] \cdot \mathrm{id}_{\Gamma/H}$$

for $K \leq H \leq \Gamma$. We consequently have the following equivalent description of cohomological Mackey functors, which is cited in [BG23a, Corollary 4.22]).

Proposition 6.2.7. ([Yos83, Theorem 4.3]) There is an equivalence of *R*-linear Grothendieck abelian categories

$$\operatorname{Mack}_{R}^{\operatorname{coh}}(\Gamma) \simeq \operatorname{PSh}_{R}(\operatorname{perm}(\Gamma; R)),$$

where $PSh_R(-)$ denotes the category of *R*-linear additive presheaves with values in *R*-modules.

Now, let k and Γ be as in Convention 6.0.1.

To see how cohomological Mackey functors fit into Equation 6.1, we construct a functor

$$\Psi_{\overline{k}} : \operatorname{Sh}_{\operatorname{et}}(\operatorname{Cor}_{k,R}) \to \operatorname{Mod}(\Gamma; R)$$
$$M \mapsto \operatorname{colim}_{k'} M(\operatorname{Spec}(k')),$$

where k' runs over finite field extensions of k contained in \overline{k} ; the colimit is taken in the category of R-modules and the $\Gamma = \text{Gal}(\overline{k}/k)$ -action is canonical. The functor $\Psi_{\overline{k}}$ is an equivalence of tensor categories by [BG23a, Lemma 6.13].

Proposition 6.2.8. ([BG23a, Proposition 6.14]) Let k be a field with a fixed separable closure \bar{k} and absolute Galois group $\Gamma := \text{Gal}(\bar{k}/k)$. The composition $\Psi_{\bar{k}} \circ a_{\text{et}} : \text{Sh}_{\text{Nis}}(\text{Cor}_{k,R}) \to \text{Mod}(\Gamma; R)$, where a_{et} denotes the étale sheafification functor, restricts to an equivalence of tensor categories

$$\operatorname{Cor}_k^0 \otimes R \simeq \operatorname{perm}(\Gamma; R).$$

Following [BG23a, Remark 6.16], we can extend $\Psi_{\overline{k}}$ to the Nisnevich-topology to get the following result.

Corollary 6.2.9. ([BG23a, Corollary 6.17]) Let k be a field with a fixed separable closure \bar{k} and absolute Galois group $\Gamma := \text{Gal}(\bar{k}/k)$. The functor

$$\Psi_{\overline{k}} : \operatorname{Sh}_{\operatorname{Nis}}(\operatorname{Cor}_{k,R}^{0}) \xrightarrow{\simeq} \operatorname{Mack}_{R}^{\operatorname{coh}}(\Gamma)$$

is an equivalence of abelian categories.

Together with Proposition 6.2.3, this induces the equivalence of the right-hand side of the triangle in Equation 6.1.

In order to establish the equivalence at the bottom of the triangle in Equation 6.1, one defines a functor, called the *fixed-point functor*, via

$$FP: Mod(\Gamma; R) \to PSh_R(perm(\Gamma; R)) \simeq Mack_R^{coh}(\Gamma)$$
$$M \mapsto Hom_{Mod(\Gamma; R)}(-, M)|_{perm(\Gamma; R)}$$

which can be seen as a restricted version of Yoneda. It admits a left adjoint $LP : PSh_R(perm(\Gamma; R)) \simeq Mack_R^{coh}(\Gamma) \rightarrow Mod(\Gamma; R)$ defined by left Kan extension (see[BG23a, Remark 5.4]).

By [BG23a, Lemma 5.5], FP is *R*-linear, lax monoidal, fully faithful and preserves filtered colimits. To justify the name *fixed-point* functor, fix a Γ -module *M*. Then, the image of a generator $R(\Gamma/H)$ of perm($\Gamma; R$) under FP(*M*) is isomorphic to M^H for any open subgroup $H \leq \Gamma$.

We want to understand compact objects better. Balmer and Gallauer prove the following using Neeman–Thomason localization, which can be transferred to different settings using Corollary 6.2.12 below.

Lemma 6.2.10. ([BG23a, Corollary 3.10]) Let Γ be a profinite group. There is a canonical equivalence

 $D \operatorname{Perm}(\Gamma; R)^c \simeq \operatorname{thick}(\operatorname{perm}(\Gamma; R)) = K_b(\operatorname{perm}(\Gamma; R)^{\natural})$

between the compact part of the triangulated category $D \operatorname{Perm}(\Gamma; R)$ and the thick subcategory of $K(\operatorname{Perm}(\Gamma; R))$ generated by finitely generated permutation modules.

Finally, [BG23a] deduces an equivalence between the derived category of permutation modules and the derived category of cohomological Mackey functors.

Proposition 6.2.11. ([BG23a, Corollary 5.7]) Let Γ be a profinite group. The fixed-point functor FP induces an equivalence of tensor triangulated categories

$$\operatorname{FP} : \operatorname{DPerm}(\Gamma; R) \xrightarrow{\simeq} \operatorname{D}(\operatorname{Mack}_{R}^{\operatorname{coh}}(\Gamma))$$

The results are summarized in the following diagram.

Corollary 6.2.12. ([BG23a, Corollary 7.10]) Let k be a field with a fixed separable closure \bar{k} and absolute Galois group $\Gamma := \text{Gal}(\bar{k}/k)$. There are equivalences of tensor triangulated categories

$$\mathrm{DAM}(k; R) \xleftarrow{\simeq} \mathrm{D}(\mathrm{Sh}_{\mathrm{Nis}}(\mathrm{Cor}_{k,R}^{0}))$$
$$\Psi_{k}^{-1} \uparrow \simeq$$
$$\mathrm{D}\operatorname{Perm}(\Gamma; R) \xrightarrow{\simeq} \mathrm{D}(\mathrm{Mack}_{R}^{\mathrm{coh}}(\Gamma))$$

which restrict to equivalences on the level of compacts

$$DAM^{gm}(k; R) \xleftarrow{\simeq} K_b((Cor^0_{k,R})^{\natural})$$
$$\overbrace{\sim}^{\simeq} K_b(perm(\Gamma; R)^{\natural})$$

Proof. The statements follow from Proposition 6.2.3, Corollary 6.2.9 and Proposition 6.2.11. $\hfill \Box$

Having seen the main results of [BG23a] and the construction of the derived category of motives DM(k; R), the next chapter aims at refining the first step to an oriented setting. For this, the construction of the derived category of Milnor–Witt motives $\widetilde{DM}(k; R)$ is recalled in Section 7.1 before we generalize Proposition 6.2.3 to the Milnor–Witt setting in the following sections.

Chapter 7

The Derived Category of Artin Milnor–Witt Motives

This chapter aims at extending some results of [BG23a] to the category DM(k; R) of Milnor–Witt motives. In the end, we want to construct the following categories and prove equivalences of tensor triangulated categories between them.

$$\widetilde{\text{DAM}}(k; R) \xleftarrow{\simeq} D(\text{Sh}_{\text{Nis}}(\widetilde{\text{Cor}}_{k,R}^{0})) \qquad (7.1)$$
$$\downarrow^{\simeq} D(\widetilde{\text{Mack}}_{R}^{\text{coh}}(k)).$$

In Remark 8.2.7, we will comment on why we do not present an analog of the category $D \operatorname{Perm}(\Gamma; R)$ in our case.

We begin with a recapitulation of the construction of the derived category of Milnor–Witt motives $\widetilde{DM}(k; R)$ in the first section. Then, we discuss further details on the category $\widetilde{Cor}_{k,R}$ of finite Milnor–Witt correspondences when restricting it to 0-dimensional schemes as in [BG23a]. In Section 7.3, we introduce Milnor–Witt Artin motives and prove the upper equivalence of the diagram above under the assumption that Conjecture 7.3.10 holds. For geometric Artin motives, we will prove the upper equivalence without assuming a conjecture to be true. The remaining equivalence will be treated in the Chapter 8.

Convention 7.0.1. For this chapter, let k be a perfect field of characteristic $\neq 2$ and R a ring.

7.1 Milnor–Witt Motives

In this section, we recall the theory of Milnor–Witt (=MW) motives following [CF22] and [DF22]. As for DM(k; R), we have similar steps to follow. The main difference here lies in the underlying category of finite correspondences, which refines to *finite MW-correspondences*. The other steps then follow in an analog

manner. Consider the following diagram illustrating the construction of DM(k; R).



Because it will be a central object of study, we dedicate the first subsection to the construction of the category $\widetilde{\text{Cor}}_{k,R}$ of finite MW-correspondences. The rest of the construction will follow in Subsection 7.1.2.

7.1.1 Milnor–Witt Correspondences

The idea why one would like to enlarge the category $\operatorname{Cor}_{k,R}$ of finite correspondences is that, for example, Chow–Witt groups (see Section 4.2) are not captured here. To understand why the concept of "decorating with \sim " is interesting, the reader is, for example, referred to the introduction of [BC+22].

In order to create a category $\operatorname{Cor}_{k,R}$ it therefore seems sensible to equip finite correspondences with symmetric bilinear forms, in this case over the function field of each irreducible component of the *support* of a finite correspondence. Together with some conditions on the symmetric bilinear form, this is exactly what the definition of MW-correspondences comes down to. However, due to technical reasons, we define the group $\operatorname{Cor}_{k,R}(X,Y)$ in terms of Chow–Witt groups.

Remark 7.1.1. At first, we need to adapt our definition of Chow–Witt groups of schemes from Definition 4.2.2 slightly. Let X be a scheme and (i, L) a graded line bundle over X in the sense of [Fas20, Section 1.3]. One can define the Chow–Witt groups $\widetilde{\operatorname{CH}}^n(X, (i, L))$ twisted by a graded line bundle (i, L). The addition of an
integer i to the line bundle L over X is only a technical convenience when treating products.

What we will be interested in now is the Chow–Witt group supported on a closed subset. For this, first recall that we can write Chow–Witt groups as the sheaf cohomology $\widetilde{\operatorname{CH}}^n(X,(i,L)) = H^n_X(X,\mathbf{K}^{\mathrm{MW}}_n(i,L))$, compare [AF16, Theorem 2.3.4]. Then, we define for a closed subset $Z \subset X$ the *n*-th Chow–Witt group supported on Z as $\widetilde{\operatorname{CH}}^n_Z(X,(i,L)) := H^n_Z(X,\mathbf{K}^{\mathrm{MW}}_n(i,L))$.

Having seen the definition of Chow–Witt groups with support, we want to define finite Milnor–Witt correspondences as the directed limit of such Chow–Witt groups.

Definition 7.1.2. Let X and Y be smooth schemes over a perfect field k of characteristic $\neq 2$ and Y equidimensional and let $T \subset X \times Y$ be a closed subset. We call T an *admissible subset* if the canonical morphism $T \to X$ is finite and maps each irreducible component of T surjectively onto an irreducible component of X. The poset of admissible subsets partially ordered by inclusion is denoted by $\mathcal{A}(X,Y)$.

Definition 7.1.3. Let X and Y be smooth schemes over a perfect field k of characteristic $\neq 2$ and Y equidimensional. Then, we define the group of *finite* Milnor-Witt correspondences

$$\widetilde{\operatorname{Cor}}_k(X,Y) := \varinjlim_{T \in \mathcal{A}(X,Y)} \widetilde{\operatorname{CH}}_T^{\dim(Y)}(X \times Y, \omega_Y),$$

where ω_Y is the graded line bundle $(\dim(Y), p_Y^* \omega_{Y/k}), p_Y : X \times Y \to Y$ being the canonical projection and $\omega_{Y/k} := \det(\Omega_{Y/k})$ is the determinant bundle of the differential sheaf as in Remark 4.2.16.

If Y is not equidimensional, i.e. $Y = \coprod_j Y_j$ with each Y_j equidimensional, we define

$$\widetilde{\operatorname{Cor}}_k(X,Y) := \prod_j \widetilde{\operatorname{Cor}}_{k,R}(X,Y_j).$$

For a ring R, we set $\widetilde{\operatorname{Cor}}_{k,R}(X,Y) := \widetilde{\operatorname{Cor}}_k(X,Y) \otimes_{\mathbb{Z}} R$.

We obtain a category $\operatorname{Cor}_{k,R}$ with objects smooth k-schemes and morphisms groups from X to Y given by $\operatorname{Cor}_{k,R}(X,Y)$. The composition is defined as in [CF22, Section 4.2].

Remark 7.1.4. When considering the set of elementary correspondences from Definition 6.1.1 a partially ordered set ordered by inclusion, it coincides with the poset of admissible subsets from Definition 7.1.2 since a scheme is integral if and only if it is reduced and irreducible.

Example 7.1.5. For arbitrary smooth schemes X, we have $\operatorname{Cor}_k(X, \operatorname{Spec}(k)) = \widetilde{\operatorname{CH}}^0(X) = \mathbf{K}_0^{\mathrm{MW}}(X).$

On the other hand, if Y has dimension d, we have

$$\widetilde{\operatorname{Cor}}_{k}(\operatorname{Spec}(k), Y) = \bigoplus_{y \in Y^{(d)}} \widetilde{\operatorname{CH}}^{d}_{\{y\}}(Y, \omega_{Y}) = \bigoplus_{y \in Y^{(d)}} \operatorname{GW}(k(y), \omega_{k(y)/k}).$$

For more details, see [CF22, Example 4.1.5].

As for the classical case, we can ask ourselves how to turn morphisms in Sm_k into finite MW-correspondences.

Example 7.1.6. Analogously to the functor $\gamma : \operatorname{Sm}_k \to \operatorname{Cor}_k$ from Example 6.1.4, we can assign to each morphism $f : X \to Y$ in Sm_k its oriented graph $\tilde{\gamma}_f$, which is constructed as follows.

Let Y be of dimension d_Y and $f: X \to Y$ a morphism in Sm_k . Further, let $\Gamma_f: X \to X \times Y$ be its graph. It follows that $\Gamma_f(X)$ is of codimension d_Y in $X \times Y$ and we obtain a finite pushforward $(\Gamma_f)_*: \mathbf{K}_0^{\mathrm{MW}}(X) \to \widetilde{\operatorname{CH}}_{\Gamma_f}^{d_Y}(X \times Y, \omega_Y)$. Eventually, we define the oriented graph of f as $\tilde{\gamma}_f := (\Gamma_f)_*(\langle 1 \rangle)$. For more details, see [CF22, Section 4.3].

In particular, we set $\tilde{\gamma}_{id} =: 1_X$ for X = Y. It is the identity of the composition of MW-correspondences defined in [CF22, Section 4.2].

This way, we obtain a functor

$$\tilde{\gamma}: \mathrm{Sm}_k \to \widetilde{\mathrm{Cor}}_k$$

being the identity on objects and sending a morphism f to its oriented graph $\tilde{\gamma}_f$.

In the beginning, we mentioned that finite Milnor–Witt correspondences can be considered finite correspondences in the sense of Voevodsky. To further comment on this, we introduce the notion of the *support* of a finite MW-correspondence.

Definition 7.1.7. For smooth X, equidimensional smooth Y, and $\alpha \in \text{Cor}_k(X, Y)$, we define the support of α as

$$\operatorname{supp}(\alpha) := \overline{\{x \in (X \times Y)^{(\dim(Y))} \mid \text{the component of } \alpha \text{ in } \\ \overline{K_0^{\operatorname{MW}}(k(x), \Lambda(x) \otimes (\omega_Y)_x) \text{ is nonzero}\}},$$

where $\Lambda(x)$ denotes the one-dimensional k(x)-vector space $\wedge^n(\mathfrak{m}/\mathfrak{m}^2)^*$. Here, \wedge denotes the exterior power, $n = \dim(Y)$, $(-)^*$ denotes the dual, and \mathfrak{m}_x is the maximal ideal corresponding to x in $\mathcal{O}_{X,x}$.

Remark 7.1.8. As promised, we want to shed some light on the connection between finite Milnor–Witt correspondences and finite correspondences in the sense of Voevodsky mentioned above. For this, let $R = \mathbb{Z}$.

Moreover, let X and Y be smooth schemes and dim(Y) = d. We describe finite Milnor–Witt correspondences alternatively (compare [CF22, Chapter 4]) as *admissible*, unramified elements in $G := \bigoplus_{x \in (X \times Y)^{(d)}} \text{GW}(k(x), \omega_{k(x)/k})$. Here, an element α of G is unramified if it is in the kernel of the residue map $G \to \bigoplus_{x \in (X \times Y)^{(d+1)}} W(k(x), \omega_{k(x)/k})$ (compare Section 4.2) and admissible if $\operatorname{supp}(\alpha)$ is the union of the closure of admissible points. A point $x \in (X \times Y)^{(d)}$ is admissible if its closure is finite surjective over X.

Now, the support defined above can be interpreted as the union of the closure of points where the symmetric bilinear forms are non-trivial. The support of an admissible, unramified element in G is a finite correspondence in the sense of Voevodsky by [CF22, Lemma 4.1.8] using Remark 7.1.11 below. On the other hand, one can equip the support of a finite correspondence with a symmetric bilinear form that is moreover unramified and admissible and obtains a finite Milnor–Witt correspondence.

As usual, we are interested in the categorical properties of the construction. First, we see that $\widetilde{\text{Cor}}_{k,R}$ is an additive category with disjoint union as direct sum.

Moreover, we can define a tensor product for finite MW-correspondences that turns the category $\widetilde{\text{Cor}}_{k,R}$ into a symmetric monoidal (or tensor) category following [CF22, Section 4.4], where more details can be found.

Definition 7.1.9. We define the tensor product $X \otimes X'$ in $\operatorname{Cor}_{k,R}$ of smooth schemes X and X' as the underlying cartesian product $X \times X'$.

Let $\alpha \in \widetilde{\operatorname{CH}}_{T_1}^{d_1}(X \times Y, \omega_Y) \otimes R$ and $\beta \in \widetilde{\operatorname{CH}}_{T_2}^{d_2}(X' \times Y', \omega_{Y'}) \otimes R$ for some admissible subsets $T_1 \subset X \times Y$ and $T_2 \subset X' \times Y'$. The exterior product of Chow–Witt groups as defined in [Fas07, Chapter 4] yields a cycle

$$\alpha \times \beta \in \widetilde{\operatorname{CH}}_{T_1 \times T_2}^{d_1 + d_2} (X \times Y \times X' \times Y', p_Y^* \omega_{Y/k} \otimes p_{Y'}^* \omega_{Y'/k}) \otimes R.$$

We define the *tensor product* $\alpha \otimes \beta$ as

$$\alpha \otimes \beta := \sigma_*(\alpha \times \beta),$$

where

$$\sigma: X \times Y \times X' \times Y' \xrightarrow{\cong} X \times X' \times Y \times Y'$$

is the transpose isomorphism. It is not only a cycle but also a finite Milnor-Witt correspondence between $X \times X'$ and $Y \times Y'$, see [CF22, Section 4.4] for details. Hence, we can define for $\alpha \in \widetilde{\operatorname{Cor}}_{k,R}^0(X,Y)$ and $\beta \in \widetilde{\operatorname{Cor}}_{k,R}^0(X',Y')$ their tensor product as $\alpha \otimes \beta := \sigma_*(\alpha \times \beta)$.

Lemma 7.1.10. ([CF22, Lemma 4.4.2]) The category $\widetilde{\text{Cor}}_{k,R}$ is symmetric monoidal by \otimes defined above.

Remark 7.1.11. By replacing Chow–Witt groups with Chow groups in the construction of $\widetilde{\text{Cor}}_{k,R}$, one recovers classical finite correspondences $\text{Cor}_{k,R}$, see [CF22, Remark 4.3.3].

In particular, the forgetful homomorphisms

$$\widetilde{\operatorname{CH}}^d_T(X \times Y, \omega_Y) \to \operatorname{CH}^d_T(X \times Y)$$

yield an additive functor

$$\pi: \operatorname{Cor}_{k,R} \to \operatorname{Cor}_{k,R}$$

such that $\gamma: \mathrm{Sm}_k \to \mathrm{Cor}_k$ is the composition $\pi \circ \tilde{\gamma}: \mathrm{Sm}_k \to \widetilde{\mathrm{Cor}}_{k,R} \to \mathrm{Cor}_{k,R}$.

On the other hand, the hyperbolic map $H_{X,Y} : \operatorname{CH}^d_T(X \times Y) \to \widetilde{\operatorname{CH}}^d_T(X \times Y, \omega_Y)$ does not yield a functor $\operatorname{Cor}_{k,R} \to \widetilde{\operatorname{Cor}}_{k,R}$ since identity and composition are not preserved by $H_{X,Y}$, compare [CF22, p.22].

In the next part, we proceed with the construction of the derived category of Milnor–Witt motives indicated at the beginning of this chapter.

7.1.2 Construction of the Derived Category of Milnor–Witt Motives

As in the construction of DM(k; R), the next step is the passage to presheaves and sheaves with transfers, in this case with MW-*transfers*. As in the classical case, sheaves with MW-transfers satisfy the sheaf property on the underlying category of smooth k-schemes.

Definition 7.1.12. A presheaf with MW-transfers is a contravariant, additive *R*-linear functor $\widetilde{\text{Cor}}_k^{\text{op}} \to R$ -Mod. The corresponding category of presheaves with MW-transfers will be denoted by $\text{PSh}_R(\widetilde{\text{Cor}}_{k,R})$.

Let t be the Zariski, Nisnevich, or étale topology, denoted by Zar, Nis, or et, respectively. A t-sheaf with MW-transfers is a presheaf with MW-transfers whose restriction to Sm_k is a t-sheaf. We denote the corresponding category of t-sheaves with MW-transfers by $\text{Sh}_t(\widetilde{\text{Cor}}_{k,R})$.

Again, we will focus on one specific type of presheaves, namely the representable ones.

Definition 7.1.13. Let X be a smooth k-scheme. We denote the representable presheaf with MW-transfers $\widetilde{\text{Cor}}_{k,R}(-,X)$ by $\tilde{c}_R(X)$. If $R = \mathbb{Z}$, we omit it from the notation.

Example 7.1.14. By [CF22, Lemma 5.0.3], the functor $X \mapsto \mathbf{K}_{j}^{\mathrm{MW}}(X) \otimes R$ is a presheaf with MW-transfers. We have that $\tilde{c}_{R}(\mathrm{Spec}(k)) = \mathbf{K}_{0}^{\mathrm{MW}} \otimes R$ for a field k.

Remark 7.1.15. The presheaf $\tilde{c}_R(X)$ is a Zar-sheaf with MW-transfers, but in general not a Nis-sheaf with transfers, see [CF22, Example 5.2.5], contrarily to the setting of (pre-)sheaves with transfers in the sense of Voevodsky as mentioned in Remark 6.1.8.

We have a forgetful functor $\widetilde{\mathcal{O}}$: $\operatorname{Sh}_t(\widetilde{\operatorname{Cor}}_{k,R}) \to \operatorname{PSh}_R(\widetilde{\operatorname{Cor}}_{k,R})$ which has a left adjoint \widetilde{a} by [DF22, Proposition 1.2.11 (1)], where \widetilde{a} is the sheafification functor. We define the following.

Definition 7.1.16. Let t be Zar, Nis, or et. For a smooth scheme X, we set $\tilde{R}_t(X) := \tilde{a}(\tilde{c}_R(X))$. When t = Nis, we sometimes omit it from the notation.

Remark 7.1.17. By definition, $\tilde{R}_t(X)$ is a *t*-sheaf with MW-transfers. It is called the *t*-sheaf associated to $\tilde{c}_R(X)$ and all *t*-sheaves of this form generate the abelian category $\mathrm{Sh}_t(\mathrm{Cor}_{k,R})$. Moreover, we obtain a tensor product $\tilde{\otimes}$ on $\mathrm{Sh}_t(\mathrm{Cor}_{k,R})$ uniquely characterized by the property $\tilde{R}_t(X)\tilde{\otimes}\tilde{R}_t(Y) = \tilde{R}_t(X \times Y)$ and commuting with colimits. For more details, see [DF22, 1.2.14].

Definition 7.1.18. Let $\mathcal{T}_{\mathbb{A}^1}$ denote the localizing triangulated subcategory of $D(Sh_t(\widetilde{Cor}_{k,R}))$ generated by complexes

$$\dots \to 0 \to \hat{R}_t(\mathbb{A}^1 \times X) \to \hat{R}_t(X) \to 0 \to \dots$$

We define the category of effective MW-motives $\widetilde{DM}_{t}^{\text{eff}}(k; R)$ as the localization of $D(\operatorname{Sh}_{t}(\operatorname{Cor}_{k,R}))$ with respect to $\mathcal{T}_{\mathbb{A}^{1}}$. The category of geometric effective MWmotives $\widetilde{DM}_{t}^{\text{eff},\text{gm}}(k; R)$ is defined as the Verdier localization of $(K_{b}(\operatorname{Cor}_{k,R}))^{\natural}$ with respect to the thick triangulated subcategory containing complexes of the form (1) and (2) from [DF22, Definition 3.2.22]. Note that $(K_{b}(\operatorname{Cor}_{k,R}))^{\natural} \cong K_{b}((\operatorname{Cor}_{k,R})^{\natural})$ by [BS01, Corollary 2.12] (the same arguments hold for K instead of D).

Definition 7.1.19. Let X be a smooth scheme. Its MW-motive $\widetilde{M}(X)$ is defined as the class of $\widetilde{R}_t(X)$ in $\widetilde{DM}_t^{\text{eff}}(k; R)$.

Remark 7.1.20. By [DF22, 3.2.4], $\widetilde{DM}_{t}^{\text{eff}}(k; R)$ is a tensor triangulated category with internal Hom. For t = Nis, the motives $\widetilde{M}(X)$ for X a smooth k-scheme form a family of compact generators as seen in [DF22, Remark 3.2.24]. The category $\widetilde{DM}_{t}^{\text{eff,gm}}(k; R)$ is also tensor triangulated with internal Hom by [BC+22, Introduction, (MW1)].

As in the case of Voevodsky's derived category of motives, the next step would be to formally invert the Tate twist. In this setting, we have the following definition.

Definition 7.1.21. The object

$$\tilde{R}(1) := \widetilde{M}(\mathbb{P}^1) / \widetilde{M}(\{\infty\})[-2]$$

in $\widetilde{\mathrm{DM}}_t^{\mathrm{eff}}(k; R)$ is called the *Tate twist*.

Again, we will not go into detail about how to invert the Tate twist since we will only concentrate on effective Milnor–Witt motives in this work. The usual procedure of inverting the Tate twist operation, also called \mathbb{P}^1 -stabilization, is usually done by passage to spectrum objects, hence the name MW-*motivic* spectra. For more details, see [DF22, Section 3.3].

Definition 7.1.22. We define the *derived category of* MW-*motives* DM(k; R) (also the *category of* MW-*motivic spectra*) as the category obtained from the derived category of effective motives $\widetilde{DM}_{Nis}^{eff}(k; R)$ by formally inverting the Tate twist operation $\otimes \tilde{R}(1)$.

Remark 7.1.23. We know from [DF22, Proposition 3.3.4] that DM(k; R) is a tensor triangulated category.

Having recalled the construction of the derived category of Milnor–Witt motives, we can now begin with formulating and proving Diagram 7.1 given at the beginning of this chapter.

7.2 Sheaves with Milnor–Witt Transfers on Zero-Dimensional Schemes

In this section, we will become more familiar with the category $\operatorname{Cor}_{k,R}$ when allowing only 0-dimensional smooth k-schemes.

Definition 7.2.1. We denote by $\widetilde{\text{Cor}}_{k,R}^0$ the full tensor subcategory of $\widetilde{\text{Cor}}_{k,R}$ spanned by 0-dimensional smooth schemes.

For smooth k-schemes of dimension 0, the definition of finite Milnor–Witt correspondences simplifies in the following way.

Example 7.2.2. Let X and Y be smooth k-schemes. We know from Example 7.1.5 that $\widetilde{\operatorname{Cor}}_{k,R}(X, \operatorname{Spec}(k)) = \widetilde{\operatorname{CH}}^0(X)$. If Y is of dimension d, then $\widetilde{\operatorname{Cor}}_{k,R}(\operatorname{Spec}(k), Y) = \bigoplus_{y \in Y^{(d)}} \widetilde{\operatorname{CH}}^d_{\{y\}}(Y, \omega_Y) \otimes R$. We can go even further and show that for 0-dimensional X and Y

$$\widetilde{\operatorname{Cor}}^0_{k,R}(X,Y) = \widetilde{\operatorname{CH}}^0(X \times Y) \otimes R$$

and

$$\widetilde{\operatorname{Cor}}^0_{k,R}(X,Y) = \widetilde{\operatorname{CH}}^0(X \times Y) \otimes R = \widetilde{Z}^0(X \times Y) \otimes R.$$

It suffices to show that $X \times Y$ is an element in $\mathcal{A}(X, Y)$. In this case, $X \times Y$ is a final object in $\mathcal{A}(X, Y)$. We then have

$$\widetilde{\operatorname{Cor}}^0_{k,R}(X,Y) = \widetilde{\operatorname{CH}}^0_{X \times Y}(X \times Y) \otimes R = \widetilde{\operatorname{CH}}^0(X \times Y) \otimes R.$$

The canonical projection $X \times Y \to X$ is surjective and maps an irreducible component of $X \times Y$ onto an irreducible component of X, compare [CF22, Section 4.1]. For the finiteness of the projection, it suffices to show that $Y \to \text{Spec}(k)$ is finite since base change of a finite morphism is finite. We assume Y = Spec(K) for a finite separable field extension K of k, because a morphism $\coprod_{i=1}^{n} \text{Spec}(K_i) \to$ Spec(k) is finite if each $\text{Spec}(K_i) \to \text{Spec}(k)$ is. But $\text{Spec}(K) \to \text{Spec}(k)$ is finite since K is a finite separable field extension of k, hence K is finitely generated as a k-vector space.

Remark 7.2.3. We did not include any twists in the above notion for the following reason. Let X and Y in $\widetilde{\text{Cor}}_{k,R}^0$ be connected, i.e. X = Spec(K) and Y = Spec(L)

for some finite separable field extensions $K, L \supset k$. Let $p_X : X \times Y \to X$ and $p_Y : X \times Y \to Y$ be the canonical projections.

In the original definition of $\operatorname{Cor}_{k,R}(X,Y)$, the Chow–Witt group of $X \times Y$ is twisted by $p_Y^* \omega_{Y/k}$. Since L is a separable field, $\omega_{Y/k}$ is canonically isomorphic to L (see [CF22, Lemma 2.2.1]). It follows that $p_Y^* \omega_{Y/k}$ is canonically isomorphic to $\omega_{X \times Y}$. By [Har77, Exercise II.8.3], $\omega_{X \times Y} \cong p_X^* \omega_X \otimes p_Y^* \omega_Y$ and since X = $\operatorname{Spec}(K)$ and $Y = \operatorname{Spec}(L)$, this is canonically equivalent to $\mathcal{O}_{X \times Y} \otimes \mathcal{O}_{X \times Y} \cong$ $\mathcal{O}_{X \times Y}$ again by [CF22, Lemma 2.2.1]. For a smooth non-connected scheme X, ω_X is defined connected component by connected component, so, this observation linearly extends to non-connected X and Y.

Remark 7.2.4. The functor $\pi : \widetilde{\operatorname{Cor}}_{k,R} \to \operatorname{Cor}_{k,R}$, induced by the forgetful homomorphism from Chow–Witt to Chow groups, now restricts to a functor $\pi : \widetilde{\operatorname{Cor}}_{k,R}^0 \to \operatorname{Cor}_{k,R}^0$. It is the identity on objects and sends a morphism in $\widetilde{\operatorname{CH}}^0(X \times Y) \otimes R$ to the respective element in $\operatorname{CH}^0(X \times Y) \otimes R$.

Recall from [CF22, Section 4.3, p.21] that the hyperbolic homomorphism $H_{X,Y}$: $\operatorname{Cor}_{k,R}(X,Y) \to \widetilde{\operatorname{Cor}}_{k,R}(X,Y)$ does not yield a functor $\operatorname{Cor}_{k,R} \to \widetilde{\operatorname{Cor}}_{k,R}$ since identity morphisms and composition are not preserved:

The composition $\pi_{X,Y} \circ H_{X,Y}$ is the multiplication by 2 (see [CF22, Section 3, p.11]), where $\pi_{X,Y} : \widetilde{\operatorname{Cor}}_{k,R}(X,Y) \to \operatorname{Cor}_{k,R}(X,Y)$ is the forgetful homomorphism. This is also true when considering only 0-dimensional schemes.

The composition $H_{X,Y} \circ \pi_{X,Y}$ is not the identity either. For example, let X = Y = Spec(k) for an algebraically closed field k and $R = \mathbb{Z}$. By [Lam05, Proposition II.3.1] and [Wei13, Lemma II.2.1], the dimension yields isomorphisms $\text{GW}(k) \cong \mathbb{Z}$ and $\text{K}_0(k) \cong \mathbb{Z}$. It follows that

$$\pi_{X,X} : \operatorname{Cor}_k(X,X) \cong \operatorname{GW}(k) \cong \mathbb{Z} \to \mathbb{Z} \cong \operatorname{K}_0(k) \cong \operatorname{Cor}_k(X,X)$$

is the identity, whereas $H_{X,X}$ is multiplication by 2.

Remark 7.2.5. We recall from Remark 7.1.15 that $\tilde{c}_R(X) := \operatorname{Cor}_{k,R}(-,X)$ is a Zariski sheaf with Milnor-Witt transfers, but in general not a Nisnevich sheaf.

However, the Nisnevich and Zariski topology coincide on Sm_k^0 as mentioned in [BG23a, Remark 6.16]. Recall that *t*-sheaves with Milnor–Witt transfers and those with classical transfers are both defined to fulfill the sheaf property only on the underlying category Sm_k , so the same considerations apply to our case. We thus have

$$PSh_{R}(\widetilde{Cor}_{k,R}^{0}) = Sh_{Zar}(\widetilde{Cor}_{k,R}^{0}) = Sh_{Nis}(\widetilde{Cor}_{k,R}^{0}),$$

and, in particular, $\tilde{c}_R(X)$ is a sheaf in the Nisnevich topology for X in Sm_k^0 .

Even more, any presheaf on $\widetilde{\operatorname{Cor}}_{k,R}^0$ is already a Zar-/Nis-sheaf with Milnor-Witt transfers: We can apply [BG23a, Remark 6.16] to the Milnor-Witt case since, in both cases, the sheaf condition needs to be satisfied only on the underlying presheaf on Sm_k^0 by definition.

However, for X not 0-dimensional, $\tilde{c}_R(X)$ is in general not a Nisnevich sheaf, see [CF22, Example 5.2.5].

Remark 7.2.6. By [DF22, Section 3.1, p.59], the derived category $D(Sh_t(Cor_{k,R}))$ is a tensor triangulated category with internal Hom, hence so is $D(Sh_t(Cor_{k,R}))$.

In this section, we have seen that the category $\widetilde{\operatorname{Cor}}_{k,R}^0$ of finite Milnor–Witt correspondences simplifies when restricting to 0-dimensional schemes; the morphism groups between X and Y are the Chow–Witt groups of the fiber product $X \times Y$. Moreover, although the representable presheaf is in general not a Nisnevich (nor an étale) sheaf with MW-transfers, it does enjoy this property when considering $\widetilde{\operatorname{Cor}}_{k,R}^0$.

In the next section, we will introduce Artin Milnor–Witt motives and establish the horizontal equivalence of Diagram 7.1 from the beginning of this chapter, assuming that Conjecture 7.3.10 is true. For geometric Artin Milnor–Witt motives, we will show the equivalence without assuming a conjecture to hold.

7.3 Artin Milnor–Witt Motives

In this section, we want to prove the upper equivalence of Diagram 7.1, which we recall here for convenience. We begin with introducing Milnor–Witt Artin motives in the upper left corner.



Definition 7.3.1. Let k be a perfect field of characteristic $\neq 2$, R a commutative ring, and t = Nis, Zar, or et. We define the category $\widetilde{\text{DAM}}_t(k; R) := \text{Loc}(\widetilde{\text{Cor}}_{k,R}^0)$ of Artin MW-motives as the localizing subcategory of $\widetilde{\text{DM}}_t^{\text{eff}}(k; R)$ generated by the motives of 0-dimensional smooth k-schemes. Moreover, we define the category of geometric Artin MW-motives as its compact part $\widetilde{\text{DAM}}_t^{\text{gm}}(k; R) :=$ $(\widetilde{\text{DAM}}_t(k; R))^c$.

Convention 7.3.2. If omitted from the notation, t = Nis.

Proposition 7.3.3. Let k be an infinite perfect field of characteristic $\neq 2$ and R a ring. There is a canonical equivalence of tensor triangulated categories

$$\widetilde{\mathrm{DAM}}^{\mathrm{gm}}(k;R) := (\widetilde{\mathrm{DAM}}(k;R))^c \simeq \mathrm{thick}(\widetilde{\mathrm{Cor}}_{k,R}^0) \simeq \mathrm{K}_b((\widetilde{\mathrm{Cor}}_{k,R}^0)^{\natural})$$

between the compact part of the tensor triangulated category DAM(k; R) and the thick triangulated subcategory of $\widetilde{DAM}(k; R)$ generated by motives $\widetilde{M}(X)$ of zerodimensional smooth k-schemes X, denoted thick $(\widetilde{Cor}_{k,R}^0)$. *Proof.* Using [DF22, Remark 3.2.24], the first equivalence follows analogously to [BG23a, Notation 7.8] from Neeman-Thomason-Localization in [BG23a, Recollection 3.8(b)], compare [Nee92, Theorem 2.1].

For the second equivalence, it suffices to show that $\widetilde{\operatorname{Cor}}_{k,R}^{0}(-,Y)$ considered a complex in degree 0 denoted by $\widetilde{M}(Y)$ is already Nisnevich-local and \mathbb{A}^{1} -local in $K_{b}((\widetilde{\operatorname{Cor}}_{k,R})^{\natural})$ for a smooth 0-dimensional k-scheme Y, i.e. local with respect to objects satisfying conditions (1) and (2) of [DF22, Definition 3.2.22]. By Remark 7.2.5, $\widetilde{\operatorname{Cor}}_{k,R}^{0}(-,Y)$ is already a Nisnevich-sheaf with Milnor–Witt transfers, hence, Nisnevich-local. In particular, this shows that the map $i : K_{b}(\widetilde{\operatorname{Cor}}_{k,R}^{0}) \to$ $D(\operatorname{Sh}_{\operatorname{Nis}}(\widetilde{\operatorname{Cor}}_{k,R}))$ is a fully faithful embedding.

It remains to show that $\widetilde{M}(Y)$ is \mathbb{A}^1 -local, but this follows from $\widetilde{M}(Y)$ being \mathbb{A}^1 -local in $D(\operatorname{Sh}_{\operatorname{Nis}}(\operatorname{Cor}_{k,R}))$ by fully faithfulness of i, so we prove this statement instead. The proof is analog to the proof of the second statement of [BG23a, Proposition 7.4]. Since the category $\operatorname{Cor}_{k,R}^0$ is additive, it suffices to show the statement for schemes Y of the form $\operatorname{Spec}(L)$ for a finite separable field extension L/k. This means that the map

$$\operatorname{Hom}_{\operatorname{D}(\operatorname{Sh}_{\operatorname{Nis}}(\widetilde{\operatorname{Cor}}_{k,R}^{0}))}(X, T^{i}(\operatorname{Spec}(L))) \to \operatorname{Hom}_{\operatorname{D}(\operatorname{Sh}_{\operatorname{Nis}}(\widetilde{\operatorname{Cor}}_{k,R}^{0}))}(\mathbb{A}^{1}_{X}, T^{i}(\operatorname{Spec}(L)))$$

is bijective for all $i \in \mathbb{Z}$ and for all smooth k-schemes X.

By Lemma 7.3.6, objects in $\widetilde{\operatorname{Cor}}_{k,R}^0$ are their own duals. It follows that

$$\operatorname{Hom}_{\operatorname{D}(\operatorname{Sh}_{\operatorname{Nis}}(\widetilde{\operatorname{Cor}}_{k,R}^{0}))}(X, T^{i}(\operatorname{Spec}(L))) \cong \operatorname{Hom}_{\operatorname{D}(\operatorname{Sh}_{\operatorname{Nis}}(\widetilde{\operatorname{Cor}}_{k,R}^{0}))}(X', T^{i}(\operatorname{Spec}(k)))$$

and

$$\operatorname{Hom}_{\mathcal{D}(\operatorname{Sh}_{\operatorname{Nis}}(\widetilde{\operatorname{Cor}}_{k,R}^{0}))}(\mathbb{A}^{1}_{X}, T^{i}(\operatorname{Spec}(L))) \cong \operatorname{Hom}_{\mathcal{D}(\operatorname{Sh}_{\operatorname{Nis}}(\widetilde{\operatorname{Cor}}_{k,R}^{0}))}(\mathbb{A}^{1}_{X'}, T^{i}(\operatorname{Spec}(k)))$$

for $X' = X \otimes \text{Spec}(L)$. Hence, it amounts to show that

$$\operatorname{Hom}_{\operatorname{D}(\operatorname{Sh}_{\operatorname{Nis}}(\widetilde{\operatorname{Cor}}_{k,R}^{0}))}(X', T^{i}(\operatorname{Spec}(k))) \to \operatorname{Hom}_{\operatorname{D}(\operatorname{Sh}_{\operatorname{Nis}}(\widetilde{\operatorname{Cor}}_{k,R}^{0}))}(\mathbb{A}^{1}_{X'}, T^{i}(\operatorname{Spec}(k)))$$

is bijective. By [DF22, Corollary 3.1.8] (which is the Milnor–Witt analog of [Voe00, Proposition 3.1.8]), this is equivalent to showing that the map

$$H^i_{\text{Nis}}(X', \operatorname{Spec}(k)) \to H^i_{\text{Nis}}(\mathbb{A}^1_{X'}, \operatorname{Spec}(k))$$

is bijective, i.e. that $\widetilde{M}(\operatorname{Spec}(k))$, which is \mathbf{K}_0^{MW} by Example 7.1.5, is strictly \mathbb{A}^1 -invariant. But this follows from [DF22, Theorem 3.2.10] and the fact that \mathbf{K}_0^{MW} is \mathbb{A}^1 -invariant by [Mor12, p.86].

Remark 7.3.4. One could hope to drop the assumption of k being infinite in the above proposition.

Remark 7.3.5. The (Voevodsky) motive M(X) of a 0-dimensional smooth scheme X is its own tensor dual in $\text{DM}_t^{\text{eff}}(k; R)$ and $\text{DM}_t(k; R)$, i.e. M(X) is isomorphic to hom(M(X), 1). In particular, it is rigid in the sense of Definition 2.2.1, see [BG23a, Remark 7.9]. In this remark, Balmer and Gallauer use the equivalence between the derived category of permutation modules and $D(\text{Sh}_{\text{Nis}}(\text{Cor}_{k,R}^0))$ and show that objects in perm $(\Gamma; R)$ are already their own tensor duals. However, one can also see directly that objects in $\text{Cor}_{k,R}^0$ enjoy this property and we generalize this idea to $\widetilde{\text{Cor}}_{k,R}^0$ as follows.

Lemma 7.3.6. The objects in $\widetilde{\operatorname{Cor}}_{k,R}^0$ are their own tensor duals.

Proof. We will use that $\widetilde{\operatorname{Cor}}_{k,R}^0(X,Y) \cong \widetilde{\operatorname{Cor}}_{k,R}^0(Y,X)$ by Example 7.2.2. The category $\widetilde{\operatorname{Cor}}_{k,R}^0$ is additively generated by objects of the form $\operatorname{Spec}(K)$ for $K \supset k$ a separable finite field extension, so, it suffices to show that $\operatorname{Spec}(K)$ is rigid with dual $\operatorname{Spec}(K)$. Recall from Definition 7.1.9 that the tensor product in $\widetilde{\operatorname{Cor}}_{k,R}^0$ is given by the underlying cartesian product of schemes. To show that objects in $\widetilde{\operatorname{Cor}}_{k,R}^0$ are rigid, by Remark 2.2.2 we need to find (co-)evaluation maps

$$\eta : \operatorname{Spec}(k) \to \operatorname{Spec}(K) \times \operatorname{Spec}(K) \in \operatorname{Hom}_{\widetilde{\operatorname{Cor}}_{k,R}^0}(\operatorname{Spec}(k), \operatorname{Spec}(K) \times \operatorname{Spec}(K))$$
$$\cong \operatorname{GW}(K \otimes K) \otimes R$$

and

$$\varepsilon : \operatorname{Spec}(K) \times \operatorname{Spec}(K) \to \operatorname{Spec}(k) \in \operatorname{Hom}_{\widetilde{\operatorname{Cor}}_{k,R}^0}(\operatorname{Spec}(k), \operatorname{Spec}(K) \times \operatorname{Spec}(K))$$
$$\cong \operatorname{GW}(K \otimes K) \otimes R$$

fulfilling the triangle identities

$$(\mathrm{id}_{\mathrm{Spec}(K)}\otimes\varepsilon)(\eta\otimes\mathrm{id}_{\mathrm{Spec}(K)})=\mathrm{id}_{\mathrm{Spec}(K)}$$

and

$$(\varepsilon \otimes \mathrm{id}_{\mathrm{Spec}(K)})(\mathrm{id}_{\mathrm{Spec}(K)} \otimes \eta) = \mathrm{id}_{\mathrm{Spec}(K)}$$

where $\operatorname{id}_{\operatorname{Spec}(K)} \in \operatorname{Hom}_{\operatorname{Cor}_{k,R}^0}(\operatorname{Spec}(K), \operatorname{Spec}(K)) \cong \operatorname{GW}(K \otimes K) \otimes R$. We choose $\eta := \varepsilon := \langle 1 \rangle \in \operatorname{GW}(K \otimes K) \otimes R$. The triangle identities now easily follow from the fact that η, ε , and $\operatorname{id}_{\operatorname{Spec}(K)}$ all represent the identity element in $\operatorname{GW}(K \otimes K) \otimes R$.

We have shown that objects in the category of finite MW-correspondences are their own tensor duals and, in particular, rigid. $\hfill \Box$

Remark 7.3.7. We can now generalize [BG23a, Remark 7.9]. By Voevodsky's cancellation theorem [Voe00, Theorem 4.3.1], we know that the passage from geometric effective (Artin) motives to geometric non-effective (Artin) motives embodies turning each object rigid; for Milnor–Witt motives, this is shown by [DF22, Theorem 3.3.9]. But, by the above considerations, the Milnor–Witt motive $\widetilde{M}(X)$ of a smooth 0-dimensional scheme X is its own tensor dual. In particular, it is rigid, so we do not distinguish effective and non-effective geometric Artin Milnor–Witt motives. Since the motives of the form $\widetilde{M}(X)$ with $\dim(X) = 0$ form a set of compact generators of $\widetilde{DAM}(k, R)$, it follows that we neither distinguish effective and non-effective Artin Milnor–Witt motives in general, just as in loc. cit.

We obtain the following structural properties for the category of (geometric) Artin motives.

Lemma 7.3.8. The category $\widetilde{DAM}(k; R)$ is a tensor triangulated category with internal Hom and the category $\widetilde{DAM}^{gm}(k; R)$ is an essentially small, idempotent complete, rigid tensor triangulated category.

Proof. The category $\widetilde{\text{DM}}^{\text{eff}}(k; R)$ is tensor triangulated and possesses an internal Hom adjoint to the tensor product by [DF22, Remark 3.2.4]. As a localizing subcategory, $\widetilde{\text{DAM}}(k; R)$ is again tensor triangulated with internal Hom. It is closed under \otimes since the tensor product is induced by the tensor product in $\text{Sh}_t(\widetilde{\text{Cor}}_{k,R}^0)$). It commutes with colimits and satisfies $\widetilde{R}_t(X) \otimes \widetilde{R}_t(Y) = \widetilde{R}_t(X \times Y)$ ([DF22, (1.2.14.a)]), where $X \times Y$ is again 0-dimensional, compare Remark 8.1.1. It follows from the \otimes -hom adjunction and Lemma 7.3.6 that $\widetilde{\text{DAM}}(k; R)$ is also closed under the internal Hom.

As the compact part of a compactly generated tensor triangulated category, the statement for $\widetilde{DAM}^{gm}(k; R)$ follows from [HPS97, Theorem 2.1.3] (compare [Bal10b, Definition 44]) under consideration of Proposition 7.3.3.

To show the equivalence between $\widetilde{\text{DAM}}^{\text{eff}}(k; R)$ and $\text{DSh}_{\text{Nis}}(\widetilde{\text{Cor}}_{k,R}^{0})$, we construct the left adjoint functor $\tilde{\iota}_! : \text{D}(\text{Sh}_{\text{Nis}}(\widetilde{\text{Cor}}_{k,R}^{0})) \to \text{D}(\text{Sh}_{\text{Nis}}(\widetilde{\text{Cor}}_{k,R}))$ and show that it is fully faithful and remains so under composition with the quotient functor to $\widetilde{\text{DM}}^{\text{eff}}(k)$ just as in the proof of [BG23a, Lemma 6.9]. It will follow that its essential image is precisely $\widetilde{\text{DAM}}^{\text{eff}}(k; R)$.

Construction 7.3.9. We consider the inclusion functor $\tilde{\iota} : \widetilde{\text{Cor}}_{k,R}^0 \to \widetilde{\text{Cor}}_{k,R}^0$. It induces a functor on the level of presheaves with values in *R*-modules

$$\tilde{\iota}^* : \mathrm{PSh}_R(\widetilde{\mathrm{Cor}}_{k,R}) \to \mathrm{PSh}_R(\widetilde{\mathrm{Cor}}_{k,R}^0),$$

defined as $\tilde{\iota}^*(F) = F \circ \tilde{\iota}$.

The category of *R*-modules is complete by [MacL71, p. 111] and $\widetilde{\operatorname{Cor}}_{k,R}^{0}$ is locally small. More generally, it is small since its objects are the same as the objects in $\operatorname{Sm}_{k}^{0}$, which form a set. Hence, we can apply [KS06, Theorem 2.3.3] to see that the left adjoint

$$\tilde{\iota}_! : \mathrm{PSh}_R(\widetilde{\mathrm{Cor}}_{k,R}^0) \to \mathrm{PSh}_R(\widetilde{\mathrm{Cor}}_{k,R})$$

constructed as in [SGA4, (5.11)] exists.

Since $\tilde{\iota}$ is fully faithful, $\tilde{\iota}_{!}$ is fully faithful as well and there are isomorphisms

$$\operatorname{id}_{\operatorname{PSh}_R(\widetilde{\operatorname{Cor}}_{k,R})} \cong \tilde{\iota}^* \tilde{\iota}_{!}$$

by [KS06, Theorem 2.3.3]. In particular, we have isomorphisms $\widetilde{\operatorname{Cor}}_{k,R}^0(-,X) \cong \tilde{\iota}^* \tilde{\iota}_!(\widetilde{\operatorname{Cor}}_{k,R}^0(-,X)) = \tilde{\iota}_!(\widetilde{\operatorname{Cor}}_{k,R}^0(-,X)) \circ \tilde{\iota}$ for each X in Sm_k^0 . Since $\tilde{\iota}$ is the inclusion, it follows that

$$\tilde{\iota}_{!}(\widetilde{\operatorname{Cor}}_{k,R}^{0}(-,X)) = \widetilde{\operatorname{Cor}}_{k,R}^{0}(-,X) = \widetilde{\operatorname{Cor}}_{k,R}(-,X)$$

for all X in Sm_k^0 .

Now, since $\tilde{\iota}$ is merely the inclusion, the functor $\tilde{\iota}^*$ preserves *t*-sheaves with MW-transfers and hence induces a functor

$$\operatorname{Sh}_t(\widetilde{\operatorname{Cor}}_{k,R}^0) \leftarrow \operatorname{Sh}_t(\widetilde{\operatorname{Cor}}_{k,R}) : \tilde{\iota}^*$$

which admits a left adjoint

$$\tilde{\iota}_{!}: \operatorname{Sh}_{t}(\widetilde{\operatorname{Cor}}_{k,R}^{0}) \to \operatorname{Sh}_{t}(\widetilde{\operatorname{Cor}}_{k,R})$$

defined as the composition $\mathrm{PSh}_R(\widetilde{\mathrm{Cor}}_{k,R}^0) \xrightarrow{\tilde{\iota}_!} \mathrm{PSh}_R(\widetilde{\mathrm{Cor}}_{k,R}) \xrightarrow{\tilde{a}_t} \mathrm{Sh}_t(\widetilde{\mathrm{Cor}}_{k,R})$ restricted to $\mathrm{Sh}_t(\widetilde{\mathrm{Cor}}_{k,R}^0)$, where \tilde{a}_t denotes the sheafification functor with respect to the topology t. Hence, we have constructed $\tilde{\iota}_! : \mathrm{Sh}_t(\widetilde{\mathrm{Cor}}_{k,R}^0) \to \mathrm{Sh}_t(\widetilde{\mathrm{Cor}}_{k,R})$ as a left adjoint functor.

It moreover induces a functor

$$\tilde{\iota}_{!}: \mathrm{D}(\mathrm{Sh}_{t}(\widetilde{\mathrm{Cor}}_{k,R}^{0})) \to \mathrm{D}(\mathrm{Sh}_{t}(\widetilde{\mathrm{Cor}}_{k,R}))$$

on the level of derived categories which we will use later.

One might conjecture that [BG23a, Lemma 6.9] allows for a refinement to Chow-Witt correspondences. For t = et, one can use that $\text{Sh}_{\text{et}}(\widetilde{\text{Cor}}_{k,R}) \simeq \text{Sh}_{\text{et}}(\text{Cor}_{k,R})$ by [DF22, Corollary 1.2.15 (3)].

Conjecture 7.3.10. The inclusion $\tilde{\iota} : \widetilde{\operatorname{Cor}}_{k,R}^0 \to \widetilde{\operatorname{Cor}}_{k,R}$ and the étale sheafification functor \tilde{a}_{et} induce a commutative square of left adjoint tensor triangulated functors with the horizontal arrows being fully faithful.

$$\begin{array}{c} \operatorname{Sh}_{\operatorname{Nis}}(\widetilde{\operatorname{Cor}}_{k,R}^{0}) \xrightarrow{\tilde{\iota}_{!}} \operatorname{Sh}_{\operatorname{Nis}}(\widetilde{\operatorname{Cor}}_{k,R}) \\ & & \\ &$$

The result motivates the following proposition, showing that $\tilde{\iota}_1$ remains fully faithful when passing to MW-motives, which is a refinement of [BG23a, Proposition 7.4].

For the rest of this subsection, we assume that Conjecture 7.3.10 is true.

Proposition 7.3.11. Let k be a perfect field of characteristic $\neq 2$ and R a commutative ring. If Conjecture 7.3.10 is true, the functor

$$\tilde{\iota}_{!}: \mathrm{D}(\mathrm{Sh}_{\mathrm{Nis}}(\widetilde{\mathrm{Cor}}_{k,R}^{0})) \to \mathrm{D}(\mathrm{Sh}_{\mathrm{Nis}}(\widetilde{\mathrm{Cor}}_{k,R}))$$

induced by inclusion satisfies the following properties.

- (i) It is tensor triangulated and fully faithful.
- (ii) The composition with the quotient functor

$$\widetilde{\mathcal{L}}_{!}: \mathcal{D}(\mathrm{Sh}_{\mathrm{Nis}}(\widetilde{\mathrm{Cor}}_{k,R}^{0})) \to \mathcal{D}(\mathrm{Sh}_{\mathrm{Nis}}(\widetilde{\mathrm{Cor}}_{k,R})) \to \widetilde{\mathrm{DM}}^{\mathrm{eff}}(k;R)$$

remains fully faithful.

Proof. Analogously to the proof of [BG23a, Proposition 7.4.(a)], we can deduce the first statement from Conjecture 7.3.10 since $\tilde{\iota}_1$ is tensor, fully faithful, triangulated, and admits a triangulated right adjoint $\tilde{\iota}^*$, so, the unit of the adjunction remains an isomorphism at the derived level.

To see that the composition with the quotient functor remains fully faithful, it suffices to show that the image of a 0-dimensional smooth scheme in $D(Sh_{Nis}(Cor_{k,R}))$ is \mathbb{A}^1 -local, but we have already shown this in the proof of Proposition 7.3.3.

Now, the following corollary refines the last equivalence of [BG23a, Corollary 7.10].

Corollary 7.3.12. Let k be a perfect field of characteristic $\neq 2$ and R a commutative ring. If Conjecture 7.3.10 is true, there is an equivalence of tensor triangulated categories

$$\tilde{\iota}_{!}: \mathrm{D}(\mathrm{Sh}_{\mathrm{Nis}}(\widetilde{\mathrm{Cor}}_{k,R}^{0})) \xrightarrow{\simeq} \widetilde{\mathrm{DAM}}(k;R).$$

Proof. By Construction 7.3.9, the fully faithful functor $\tilde{\iota}_{!} : D(Sh_{Nis}(\widetilde{Cor}_{k,R}^{0})) \to \widetilde{DM}_{Nis}^{eff}(k;R)$ has image $\widetilde{DAM}(k;R)$.

Chapter 8

Cohomological Milnor–Witt Mackey Functors

It might help to have a copy of [BG23a] at hand while reading this chapter.

After having proven the upper equivalence of Diagram 7.1 in the last chapter, we want to construct the vertical equivalence of

$$\widetilde{\mathrm{DAM}}(k; R) \xleftarrow{\simeq} \mathrm{D}(\mathrm{Sh}_{\mathrm{Nis}}(\widetilde{\mathrm{Cor}}_{k,R}^{0}))$$
$$\downarrow \simeq$$
$$\mathrm{D}(\widetilde{\mathrm{Mack}}_{R}^{\mathrm{coh}}(k))$$

refining the equivalence in [BG23a, Corollary 6.17].

We stay in the world of algebraic geometry; a Milnor–Witt analog of permutation modules will merely be commented in Remark 8.2.7. In Section 8.1, we introduce a category $\widetilde{\Omega}_R(k)$ closely related to classical span categories and investigate its relation to $\widetilde{\operatorname{Cor}}_{k,R}^0$. The idea of Section 8.2 then is to construct a category of cohomological Milnor–Witt Mackey functors $\widetilde{\operatorname{Mack}}_R^{\operatorname{coh}}(k)$ and show that $\widetilde{\operatorname{Mack}}_R^{\operatorname{coh}}(k) \simeq \operatorname{Sh}_{\operatorname{Nis}}(\widetilde{\operatorname{Cor}}_{k,R}^0)$.

Basics on symmetric bilinear forms and Grothendieck–Witt groups of fields and rings used here are taken from [Scha85, Lam05].

Convention 8.0.1. Let R be a commutative ring. In this chapter, let $\Gamma := \operatorname{Gal}(\bar{k}/k)$ be the absolute Galois group of a perfect field k of characteristic $\neq 2$ for a fixed algebraic closure \bar{k} . For a subgroup H of Γ and an element $g \in \Gamma$, we set ${}^{g}H := gHg^{-1}$ and $H^{g} := g^{-1}Hg$. We will only consider open subgroups of Γ .

8.1 Span Categories

To begin with, we need an analog $\widetilde{\Omega}_R(k)$ of the additive category $\Omega_R(k)$ defined in [ThWe95, Section 2] and recalled in [BG23a, Remark 6.5]. The construction is motivated by the idea of obtaining a functor $\widetilde{\Omega}_R(k) \to \widetilde{\operatorname{Cor}}_{k,R}^0$ that sends a morphism to its oriented pushforward, analogously to the non-oriented case. After defining the category $\widetilde{\Omega}_R(k)$ we will go into more detail about how $\widetilde{\Omega}_R(k)$ and $\widetilde{\operatorname{Cor}}_{k,R}^0$ relate.

We start with an observation on tensor products of 0-dimensional smooth k-schemes.

Remark 8.1.1. Recall that the fiber product of 0-dimensional smooth k-schemes is isomorphic to an object of Sm_k^0 . In the case $Z = \operatorname{Spec}(K_Z)$, $Y = \operatorname{Spec}(K_Y)$ and $V = \operatorname{Spec}(K_V)$ we have that

$$Z \times_Y V = \operatorname{Spec}(K_Z) \times_{\operatorname{Spec}(K_Y)} \operatorname{Spec}(K_V) = \operatorname{Spec}(K_Z \otimes_{K_Y} K_V).$$

It suffices to restrict to this case since Sm_k^0 is additively generated by such schemes and pulling back is an additive functor. Now, the tensor product of fields is not always a field, but, in our case, it is isomorphic to a finite product of finite separable field extensions Ω_i of K_Y

$$K_Z \otimes_{K_Y} K_V \cong \prod_{i=1}^n \Omega_i,$$

see [Milne20, Theorem 1.18]. Hence,

$$\operatorname{Spec}(K_Z \otimes_{K_Y} K_V) \cong \operatorname{Spec}(\prod_{i=1}^n \Omega_i) = \prod_{i=1}^n \operatorname{Spec}(\Omega_i)$$

is isomorphic to a finite disjoint union of spectra of finite separable field extensions over K_Y and, thus, over k, i.e. in particular, isomorphic to an object of Sm_k^0 .

Concretely, if $K_Z = K_Y(\alpha)$ for some $\alpha \in K_Z$, the minimal polynomial f of α has a decomposition in $K_V[X]$ into monic irreducible polynomials $f = f_1 \cdot \ldots \cdot f_n$. Then, we have $\Omega_i = K_V[X]/(f_i)$ for all i by the Chinese Remainder Theorem since K_Z is separable over K_Y .

In the special case $K := K_Z = K_V$ and $L := K_Y$, let $L(\alpha) = K$ for some $\alpha \in K$ and let f be the minimal polynomial of α , hence, K = L[X]/(f). Then, we have

$$K \otimes_L K \cong K \otimes_L L[X]/(f) \cong K[X]/(f).$$

In particular, since $\alpha \in K$, the linear polynomial $X - \alpha$ is a factor of f in K[X], so,

$$K[X]/(X - \alpha) \cong K$$

is one of the Ω_i from above. It follows that the tensor product $K \otimes_L K$ contains at least one copy of K.

Let us now come to the definition of $\widehat{\Omega}(k)$, which is a refinement of [ThWe95, Section 2].

Construction 8.1.2. Let k be a perfect field of characteristic $\neq 2$ and R a commutative ring. At first, we define the category $\widetilde{\text{span}}(k)$ as follows. The **objects** are the same as in Sm_k^0 , i.e., finite disjoint unions of spectra of finite separable field extensions of k.

Morphisms $f : X \to Y$ are given by isomorphism classes of so-called *(ori-ented) spans* consisting of pairs of morphisms

$$X \leftarrow Z := \prod_{i=1}^{n} \operatorname{Spec}(L_i) \to Y$$

in $\operatorname{Sm}_{k}^{0}$, where L_{i} is a finite separable field extension of k for each i, together with an equivalence class of symmetric bilinear forms

$$\langle \alpha_i \rangle \in Z^0(\operatorname{Spec}(L_i)) = \operatorname{GW}(L_i)$$

for each *i*, where $\alpha_i \in L_i^{\times}$, i.e. the rank of $\langle \alpha_i \rangle$ is 1. We denote such a morphism by $X \leftarrow \coprod_{i=1}^n (\operatorname{Spec}(L_i), \langle \alpha_i \rangle) \to Y$ or $\coprod_{i=1}^n (\operatorname{Spec}(L_i), \langle \alpha_i \rangle) \to X \times Y$.

Two spans $X \leftarrow \coprod_{i=1}^{n} (\operatorname{Spec}(L_i), \langle \alpha_i \rangle) \to Y$ and $X \leftarrow \coprod_{j=1}^{m} (\operatorname{Spec}(L_j), \langle \alpha_j \rangle) \to Y$ are **isomorphic** if there is an isomorphism $f : Z = \coprod_{i=1}^{n} \operatorname{Spec}(L_i) \to Z' = \coprod_{i=1}^{m} \operatorname{Spec}(L_j)$, so, in particular, n = m, making the diagram



commute and satisfying the following property. For all index pairs (i, j) with $f(\operatorname{Spec}(L_i)) = \operatorname{Spec}(L'_j)$, we have that $f'(\langle \alpha_i \rangle) = \langle \alpha'_j \rangle \in \widetilde{Z}^0(\operatorname{Spec}(L'_j))$, where $f': \widetilde{Z}^0(Z) \xrightarrow{\cong} \widetilde{Z}^0(Z')$ is the isomorphism on oriented 0-cocycles induced by f.

The **composition** in $\widetilde{\text{span}}(k)$ is defined as follows. Let Z = Spec(L) and V = Spec(K) for K, L finite separable field extensions of k. Consider morphisms $X \xleftarrow{r} (Z, \langle \alpha \rangle) \longrightarrow Y$ and $Y \xleftarrow{} (V, \langle \beta \rangle) \xrightarrow{t} W$ and let Y be connected w.l.o.g. Consider the pullback square



Since $\widetilde{Z}^0(Z) = \widetilde{\operatorname{CH}}^0(Z)$ for all Z in Sm_k^0 , we can apply the pullback construction of Chow–Witt groups from [Fas08, Corollaire 10.4.3] in order to obtain elements

$$(p_Z)^*(\langle \alpha \rangle) \in Z^0(Z \times_Y V)$$

and

$$(p_V)^*(\langle \beta \rangle) \in Z^0(Z \times_Y V).$$

We fix an isomorphism $\varphi : Z \times_Y V \xrightarrow{\cong} \coprod_{i=1}^n \operatorname{Spec}(\Omega_i)$ for finite separable field extensions Ω_i/k for all *i*, which exists by Remark 8.1.1. It induces an isomorphism φ' on oriented 0-cocycles. Then, $\varphi'((p_Z)^*(\langle \alpha \rangle))$ and $\varphi'((p_V)^*(\langle \beta \rangle))$ have the form

$$\langle \alpha_1' \rangle \oplus ... \oplus \langle \alpha_n' \rangle \in \bigoplus_{i=1}^n \widetilde{Z}^0(\operatorname{Spec}(\Omega_i)) \cong \bigoplus_{i=1}^n \operatorname{GW}(\Omega_i)$$

and

$$\langle \beta_1' \rangle \oplus ... \oplus \langle \beta_n' \rangle \in \bigoplus_{i=1}^n \widetilde{Z}^0(\operatorname{Spec}(\Omega_i)) \cong \bigoplus_{i=1}^n \operatorname{GW}(\Omega_i),$$

respectively, where $\alpha'_i, \beta'_i \in \Omega^{\times}_i$ for all *i*. Let

$$\langle \alpha_i' \rangle \cdot \langle \beta_i' \rangle \in \widetilde{Z}^0(\operatorname{Spec}(\Omega_i)) = \widetilde{\operatorname{CH}}^0(\operatorname{Spec}(\Omega_i))$$

denote the intersection product of Chow–Witt groups (see [Fas07, Definition 6.1]), which coincides with the multiplication in $GW(\Omega_i)$ by [Fas07, Theorem 7.6]. The composition is then defined as the morphism

$$X \xleftarrow{r \circ p_Z} \prod_{i=1}^n (\operatorname{Spec}(\Omega_i), \langle \alpha'_i \rangle \cdot \langle \beta'_i \rangle) \xrightarrow{t \circ p_V} W.$$

The construction is linearly extended to general Z, V, Y in Sm_k^0 . The composition is independent of the choice of φ , as Remark 8.1.3 shows.

One checks that the **identity element** in $\operatorname{Hom}_{\widetilde{\operatorname{span}}(k)}(Z, Z)$ for an object $Z = \prod_{i=1}^{n} \operatorname{Spec}(L_i)$ is the morphism

$$Z \xleftarrow{\operatorname{id}} \prod_{i=1}^{n} (\operatorname{Spec}(L_i), \langle 1 \rangle) \xrightarrow{\operatorname{id}} Z,$$

where the L_i are finite separable field extensions of k. The composition is **asso**ciative since the multiplication in Grothendieck–Witt groups is associative and the composition in the non-oriented counterpart span(k) is as well.

We have a **monoid structure** on $\operatorname{Hom}_{\widetilde{\operatorname{span}}(k)}(X,Y)$ given by

$$[X \leftarrow \coprod_{i=1}^{n} (\operatorname{Spec}(L_{i}), \langle \alpha_{i} \rangle) \to Y] + [X \leftarrow \coprod_{j=1}^{m} (\operatorname{Spec}(L'_{j}), \langle \alpha'_{j} \rangle) \to Y]$$
$$:= [X \leftarrow (\coprod_{k=1}^{n+m} (\operatorname{Spec}(L_{k}), \langle \alpha_{k} \rangle) \to Y],$$

where $L_{n+j} := L'_j$ and $\langle \alpha_{n+j} \rangle := \langle \alpha'_j \rangle$ for all $j \in \{1, ..., m\}$. One checks that the empty set gives rise to a zero element.

The **disjoint union** equips $\widetilde{\text{span}}(k)$ with finite products and coproducts, see Remark 8.1.4.

We denote by $\Omega(k)$ the additive category after group completing the Homsets and imposing the following equivalence relation on Hom-sets. The morphism group $\operatorname{Hom}_{\widetilde{\Omega}(k)}(X,Y)$ is then the group of isomorphism classes of oriented spans dividing out the equivalence relation

$$[X \leftarrow (Z, \langle a \rangle) \to Y] + [X \leftarrow (Z, \langle b \rangle) \to Y]$$

= $[X \leftarrow (Z, \langle a + b \rangle) \to Y] + [X \leftarrow (Z, \langle ab(a + b) \rangle) \to Y],$

where X, Y are in $\operatorname{Sm}_{k}^{0}, Z = \operatorname{Spec}(L)$ for some finite separable field extension L/k, $a, b, a + b \in L^{\times}$, and we procede componentwise for general Z. This relation is inspired by the definition of a Grothendieck–Witt group of a field via relations, compare [Lam05, Theorem II.4.3]. Note that the relation

$$[X \leftarrow (Z, \langle a \rangle) \to Y] = [X \leftarrow (Z, \langle ab^2 \rangle) \to Y] \in \operatorname{Hom}_{\widetilde{\Omega}(k)}(X, Y)$$

is already satisfied by construction.

We may **extend scalars** to a given commutative ring R in order to obtain a category $\widetilde{\Omega}_R(k)$. While objects are the same as in $\widetilde{\Omega}(k)$, oriented spans f: $(\coprod_{i=1}^n (\operatorname{Spec}(L_i), \langle \alpha_i \rangle)) \to X \times Y, \ \alpha_i \in L_i^x$, now satisfy $\langle \alpha_i \rangle \in \widetilde{Z}^0(\operatorname{Spec}(L_i)) \otimes R) =$ $\operatorname{GW}(L_i) \otimes R$.

Remark 8.1.3. Let the notation be as in Construction 8.1.2. The composition in $\widetilde{\operatorname{span}}_R(k)$ (and therefore in $\widetilde{\Omega}(k)$) does not depend on the choice of the isomorphism $\varphi : Z \times_Y V \xrightarrow{\cong} \coprod_{i=1}^n \operatorname{Spec}(\Omega_i)$. Take another isomorphism $\psi : Z \times_Y V \xrightarrow{\cong} \coprod_{i=1}^n \operatorname{Spec}(\Omega_i)$. We obtain isomorphisms

$$\bigoplus_{i=1}^{n} \widetilde{Z}^{0}(\operatorname{Spec}(\Omega'_{i}))) \otimes R \xrightarrow{(\psi^{-1})'} \widetilde{Z}^{0}(Z \times_{Y} V) \otimes R \xrightarrow{\varphi'} \bigoplus_{i=1}^{n} \widetilde{Z}^{0}(\operatorname{Spec}(\Omega_{i}))) \otimes R.$$

Then, the resulting compositions for the different choices φ and ψ are in the same isomorphism class in $\operatorname{Hom}_{\widetilde{\operatorname{span}}(k)}(X,W)$ via the isomorphism $\varphi \circ \psi^{-1}$ since it fulfills

$$(\varphi' \circ (\psi^{-1})')(\psi'((p_Z)^*(\langle \alpha \rangle))) = \varphi'((p_Z)^*(\langle \alpha \rangle))$$

and

$$(\varphi' \circ (\psi^{-1})')(\psi'((p_V)^*(\langle \beta \rangle))) = \varphi'((p_V)^*(\langle \beta \rangle)).$$

Remark 8.1.4. Let us briefly verify that the disjoint union is product and coproduct in $\widetilde{\text{span}}(k)$ and its non-oriented counterpart span(k) from [ThWe95, Section 2].

First, we show that the disjoint union is a product. Let A, B, T, X, Y be connected 0-dimensional k-schemes. This observation extends to the non-connected

case. Let $T \longleftarrow A \xrightarrow{f_X} X$ and $T \longleftarrow B \xrightarrow{f_Y} Y$ be spans in span(k). We construct a span

$$f = [T \longleftarrow A \coprod B \xrightarrow{(f_X, f_Y)} X \coprod Y].$$

Moreover, we set

$$\pi_X = [X \coprod Y \xleftarrow{i_X} X \xrightarrow{\mathrm{id}} X]$$

and

$$\pi_Y = [X \coprod Y \xleftarrow{i_Y} Y \xrightarrow{\mathrm{id}} Y].$$

Since the fiber product $(A \coprod B) \times_{X \coprod Y} X$ is simply A, the composition $\pi_X \circ f$ is the span

$$T \leftarrow (A \coprod B) \times_{X \coprod Y} X \to X$$
$$= T \longleftarrow A \xrightarrow{f_X} X$$

Similarly, $\pi_Y \circ f = f_Y$. One checks that for a different choice of f, at least one of the equations $\pi_X \circ f = f_X$ and $\pi_Y \circ f = f_Y$ is not satisfied. Hence, the choice of f is unique.

Now, consider spans $T \leftarrow (A, \langle \alpha \rangle) \xrightarrow{f_X} X$ and $T \leftarrow (B, \langle \beta \rangle) \xrightarrow{f_Y} Y$ in $\widehat{\text{span}}(k)$. We equip the spans π_X, π_Y , and f from above with the forms $\langle 1 \rangle \in \widetilde{Z}^0(X), \langle 1 \rangle \in \widetilde{Z}^0(Y)$, and $\langle \alpha \rangle \coprod \langle \beta \rangle$, respectively, where the latter notation means $\langle \alpha \rangle \in \widetilde{Z}^0(A)$ and $\langle \beta \rangle \in \widetilde{Z}^0(B)$. From the above description of the span $\pi_X \circ f = f_X$ it follows that the composition of $T \leftarrow (A \coprod B, \langle \alpha \coprod \langle \beta \rangle) \xrightarrow{(f_X, f_Y)} X \coprod Y$ and $X \coprod Y \xleftarrow{i_X} X \xrightarrow{\text{id}} X$ is again $T \leftarrow (A, \langle \alpha \rangle) \xrightarrow{f_X} X$. The same follows for the component Y. The choice of f in span(k) has been unique and if we equip f with a different form than $\langle \alpha \rangle \coprod \langle \beta \rangle$, the respective diagram does not commute. Hence, the choice of f together with $\langle \alpha \rangle \coprod \langle \beta \rangle$ is unique as well.

This shows that the disjoint union is a product in both $\operatorname{span}(k)$ and $\widetilde{\operatorname{span}}(k)$. It can be seen directly that $\operatorname{span}(k) \simeq \operatorname{span}(k)^{\operatorname{op}}$ and $\widetilde{\operatorname{span}}(k) \simeq \widetilde{\operatorname{span}}(k)^{\operatorname{op}}$, so product and coproduct coincide.

We now turn back to the category $\widetilde{\Omega}_R(k)$. There is an additive forgetful functor

$$V: \Omega_R(k) \to \Omega_R(k)$$

that is the identity on objects and is induced by the forgetful homomorphism $V: \widetilde{Z}^0(Z) \otimes R \to Z^0(Z) \otimes R$ on morphism groups.

We also obtain a functor from $\widetilde{\Omega}_R(k)$ to $\widetilde{\operatorname{Cor}}_{k,R}^{0}$ induced by the pushforward of Chow–Witt groups. For an overview of the general construction of the pushforward of Chow–Witt groups, the reader is referred to [Fas20, Section 2.3]. Recall from Remark 7.2.3 that we may fix isomorphisms trivializing the twists in an canonical way in our case.

Lemma 8.1.5. The functor $\tilde{\varepsilon} : \widetilde{\Omega}_R(k) \to \widetilde{\operatorname{Cor}}_{k,R}^0$ is well-defined with respect to the equivalence relation on Hom-sets.

Proof. Let $f_{\alpha} := [(\operatorname{Spec}(L), \langle \alpha \rangle) \xrightarrow{f} X \times Y]$ be in $\widetilde{\Omega}_{R}(k)$, $\alpha, \beta, \alpha + \beta \in L^{\times}$, and $f_{\beta}, f_{\alpha+\beta}, f_{\alpha\beta(\alpha+\beta)}$ the same spans but equipped with the respective bilinear forms $\langle \beta \rangle, \langle \alpha+\beta \rangle$ and $\langle \alpha\beta(\alpha+\beta) \rangle$. We have to check that $\tilde{\varepsilon}(f_{\alpha}+f_{\beta}) = \tilde{\varepsilon}(f_{\alpha+\beta}+f_{\alpha\beta(\alpha+\beta)})$. By additivity of $\tilde{\varepsilon}$, we have

 $\tilde{\varepsilon}(f_{\alpha} + f_{\beta}) = \tilde{\varepsilon}(f_{\alpha}) + \tilde{\varepsilon}(f_{\beta}) = f_{*}(\langle \alpha \rangle) + f_{*}(\langle \beta \rangle).$

Since f_* is a group homomorphism ([Lam05, Corollary VII.1.5]) and because the relation $\langle \alpha \rangle + \langle \beta \rangle = \langle \alpha + \beta \rangle + \langle \alpha \beta (\alpha + \beta) \rangle$ holds in GW(L) $\otimes R$, it follows that

$$\tilde{\varepsilon}(f_{\alpha} + f_{\beta}) = f_*(\langle \alpha \rangle + \langle \beta \rangle) = f_*(\langle \alpha + \beta \rangle + \langle \alpha \beta (\alpha + \beta) \rangle) = f_*(\langle \alpha + \beta \rangle) + f_*(\langle \alpha \beta (\alpha + \beta) \rangle)) = \tilde{\varepsilon}(f_{\alpha+\beta}) + \tilde{\varepsilon}(f_{\alpha\beta(\alpha+\beta)}) = \tilde{\varepsilon}(f_{\alpha+\beta} + f_{\alpha\beta(\alpha+\beta)}).$$

Lemma 8.1.6. Let k be a perfect field of characteristic $\neq 2$ and R a ring. The pushforward of Chow–Witt groups induces an additive functor

$$\widetilde{\varepsilon}:\widetilde{\Omega}(k)\to\widetilde{\operatorname{Cor}}_k^0$$

being the identity on objects. It sends a span $f: (Z = \operatorname{Spec}(L), \langle \alpha \rangle) \to X \times Y$ to the oriented pushforward $\tilde{\varepsilon}(f) := f_*(\langle \alpha \rangle) \in \widetilde{\operatorname{CH}}^0(X \times Y) = \widetilde{Z}^0(X \times Y)$ and we extend this definition linearly for general Z in Sm_k^0 . Extending scalars to a ring R, we obtain an additive functor $\tilde{\varepsilon}: \widetilde{\Omega}_R(k) \to \widetilde{\operatorname{Cor}}_{k,R}^0$.

Proof. Additivity is given by construction. To see that the identity is preserved, it thus suffices to check that $\operatorname{Spec}(L) \leftarrow (\operatorname{Spec}(L), \langle 1 \rangle) \to \operatorname{Spec}(L)$ with $L \supset k$ a finite separable field extension and $1 \in L$ is preserved. We have by Example 8.1.9 below that

$$\tilde{\varepsilon}(\operatorname{Spec}(L) \leftarrow (\operatorname{Spec}(L), \langle 1 \rangle) \to \operatorname{Spec}(L)) \\ = (\operatorname{Tr}_{L}^{L})_{*}(\langle 1 \rangle) = \operatorname{id}_{*}(\langle 1 \rangle) = \langle 1 \rangle \in \widetilde{Z}^{0}(\operatorname{Spec}(L) \times \operatorname{Spec}(L)).$$

The fact that composition is preserved follows directly from the definition of the composition in $\widetilde{\operatorname{Cor}}_k^0$ introduced in [CF22, Section 4.2]. The statement after extending scalars follows immediately.

Let us consider examples and special cases in order to understand $\tilde{\varepsilon}$ better. Since, in our case, we only treat (finite coproducts of) zero-dimensional regular k-schemes, the definition simplifies and essentially breaks down to the Scharlau transfer of Grothendieck–Witt groups with respect to the transfer map, compare [Lam05, Chapter VII, Section 1]. Remark 8.1.7. Let $X \times Y \cong \operatorname{Spec}(\prod_{i=1}^{n} \Omega_i) = \coprod_{i=1}^{n} \operatorname{Spec}(\Omega_i)$ for X, Y in Sm_k^0 (by Remark 8.1.1), i.e. Ω_i is a finite separable field extension of k for every i. Recall from [BG23a, Remark 6.5] that a morphism $f : Z \to X \times Y, Z = \operatorname{Spec}(L)$, factors through one of the points $\operatorname{Spec}(\Omega_j)$ of $X \times Y$, and, thus, the pushforward $f_* : \operatorname{GW}(L) \otimes R = \widetilde{\operatorname{CH}}^0(Z) \otimes R \to \widetilde{\operatorname{CH}}^0(X \times Y) \otimes R = \bigoplus_{i=1}^{n} \operatorname{GW}(\Omega_i) \otimes R$ is given by the composition

$$f_*: \mathrm{GW}(L) \otimes R \xrightarrow{(\mathrm{Tr}_{\Omega_j}^L)_*} \mathrm{GW}(\Omega_j) \otimes R \xrightarrow{(\mathrm{Tr}_{\Omega_j}^L)_*} \mathrm{GW}(\Omega_i) \otimes R ,$$

where $(\operatorname{Tr}_{\Omega_j}^L)_*$ is the Scharlau transfer of the field trace $\operatorname{Tr}_{\Omega_j}^L : L \to \Omega_j$, compare [Lam05, Chapter VII, Section 1]. It sends a one-dimensional form $\langle \alpha \rangle$, $\alpha \in L^{\times}$, to the scaled trace form

$$(\operatorname{Tr}_{\Omega_j}^L)_{\alpha} : L \times L \to \Omega_j$$

 $(x, y) \mapsto \operatorname{Tr}_{\Omega_i}^L(\alpha x y)$

of dimension $[L:\Omega_j]$.

To understand the functor $\tilde{\varepsilon}$ intuitively, we give some concrete examples for the case $k = \mathbb{R}$, hence, $\otimes = \otimes_{\mathbb{R}}$.

Example 8.1.8. Let $k = \mathbb{R}$ and $R = \mathbb{Z}$.

(i) Consider the span

$$\operatorname{Spec}(\mathbb{C}) \xleftarrow{\operatorname{id}} \operatorname{Spec}(\mathbb{C}) \xrightarrow{\overline{(-)}} \operatorname{Spec}(\mathbb{C})$$

corresponding to the ring homomorphism

$$\mathbb{C} \otimes \mathbb{C} \cong \mathbb{C} \times \mathbb{C} \to \mathbb{C} \\ u \otimes v \mapsto (uv, u\overline{v}) \mapsto u\overline{v}$$

hence, yielding the map

$$\begin{aligned} f: \operatorname{Spec}(\mathbb{C}) &\to \operatorname{Spec}(\mathbb{C} \times \mathbb{C}) \cong \operatorname{Spec}(\mathbb{C}) \amalg \operatorname{Spec}(\mathbb{C}) \\ (0) &\mapsto \mathbb{C} \times (0) \end{aligned}$$

factoring through the second component of $\operatorname{Spec}(\mathbb{C}) \amalg \operatorname{Spec}(\mathbb{C})$.

The pushforward $f_* : \mathrm{GW}(\mathbb{C}) \to \mathrm{GW}(\mathbb{C} \otimes \mathbb{C}) \cong \mathrm{GW}(\mathbb{C}) \oplus \mathrm{GW}(\mathbb{C})$ is now the composition of the identity $\mathrm{id} = (\mathrm{Tr}_{\mathbb{C}}^{\mathbb{C}})_*$ and the inclusion into the second component, hence the only equivalence class of one-dimensional forms $\langle 1 \rangle \in$ $\mathrm{GW}(\mathbb{C}) \cong \mathbb{Z}$ is sent to itself in the second component of $\mathrm{GW}(\mathbb{C}) \oplus \mathrm{GW}(\mathbb{C})$ under $\tilde{\varepsilon}$. (ii) Consider the span

$$\operatorname{Spec}(\mathbb{R}) \xleftarrow{\pi} \operatorname{Spec}(\mathbb{C}) \xrightarrow{\pi} \operatorname{Spec}(\mathbb{R})$$

corresponding to multiplication $m : \mathbb{R} \otimes \mathbb{R} \cong \mathbb{R} \to \mathbb{C}$ yielding

$$g: \operatorname{Spec}(\mathbb{C}) \to \operatorname{Spec}(\mathbb{R})$$
$$(0)_{\mathbb{C}} \mapsto (0)_{\mathbb{R}}.$$

Thus, g_* is the Scharlau transfer $(\operatorname{Tr}_{\mathbb{R}}^{\mathbb{C}})_* : \operatorname{GW}(\mathbb{C}) \to \operatorname{GW}(\mathbb{R})$ sending $\langle 1 \rangle \in \operatorname{GW}(\mathbb{C})$ to its trace form $(\operatorname{Tr}_{\mathbb{R}}^{\mathbb{C}})_1$.

To determine the matrix representation of $(\operatorname{Tr}_{\mathbb{R}}^{\mathbb{C}})_1$, we choose the \mathbb{R} -basis $e_1 = 1, e_2 = i$ of \mathbb{C} and determine

$$\operatorname{Tr}_{\mathbb{R}}^{\mathbb{C}}(e_i \cdot e_j)$$

for $i, j \in \{1, 2\}$. The multiplication by i is given by the matrix $\begin{pmatrix} 0 & -1 \\ 1 & 0 \end{pmatrix}$, so, for $e_i \neq e_j$, we have $\operatorname{Tr}_{\mathbb{R}}^{\mathbb{C}}(e_i e_j) = \operatorname{Tr}_{\mathbb{R}}^{\mathbb{C}}(i) = \operatorname{tr}\left(\begin{pmatrix} 0 & -1 \\ 1 & 0 \end{pmatrix}\right) = 0 + 0 = 0$. Moreover, $\operatorname{Tr}_{\mathbb{R}}^{\mathbb{C}}(1) = 1 + 1 = 2$ and $\operatorname{Tr}_{\mathbb{R}}^{\mathbb{C}}(i^2) = -1 + (-1) = -2$, so the trace form $(\operatorname{Tr}_{\mathbb{R}}^{\mathbb{C}})_1$ is $\begin{pmatrix} 2 & 0 \\ 0 & -2 \end{pmatrix} = \langle 2, -2 \rangle \cong \langle 1, -1 \rangle = \mathbb{H} \in \operatorname{GW}(\mathbb{R})$.

Hence, the image of the span

$$\operatorname{Spec}(\mathbb{R}) \xleftarrow{i} (\operatorname{Spec}(\mathbb{C}), \langle 1 \rangle) \xrightarrow{i} \operatorname{Spec}(\mathbb{R})$$

under $\tilde{\varepsilon}$ in $\widetilde{\operatorname{Cor}}_k^0(\operatorname{Spec}(\mathbb{R}), \operatorname{Spec}(\mathbb{R}))$ is the hyperbolic form $\langle 1, -1 \rangle \in \operatorname{GW}(\mathbb{R})$.

For our purposes, it will be important to understand where spans of the following special type are sent to under $\tilde{\varepsilon}$, just as in [BG23a, Remark 6.5].

Example 8.1.9. Let L/K/k be finite separable field extensions and consider the span

$$f = [\operatorname{Spec}(K) \xleftarrow{\pi} (\operatorname{Spec}(L), \langle \alpha \rangle) \xrightarrow{\pi} \operatorname{Spec}(K)].$$

We know from [BG23a, Remark 6.5] that the underlying map f without $langle\alpha$ factors through Spec(K), hence its image in $\widetilde{\text{Cor}}_{k,R}^0$ is given by

$$\tilde{\varepsilon}(f) = f_*(\langle \alpha \rangle) = (\operatorname{Tr}_K^L)_*(\langle \alpha \rangle) = (\operatorname{Tr}_K^L)_\alpha \in \operatorname{GW}(K) \otimes R \longrightarrow \operatorname{GW}(K \otimes K) \otimes R$$

Forgetting the symmetric bilinear form $\langle \alpha \rangle$, the span f gets send to [L:K] $\mathrm{id}_{\mathrm{Spec}(K)}$ under $\varepsilon : \Omega_R(k) \to \mathrm{Cor}_{k,R}^0$, see loc.cit.

More generally, the functor $\tilde{\varepsilon}$ is a generalization of the functor ε in the following sense.

Lemma 8.1.10. Let k be a perfect field of characteristic $\neq 2$ and R a ring. The following diagram commutes, where the horizontal arrows are the forgetful functors.



Proof. Let $R = \mathbb{Z}$ for better readability. All arrows are the identity on objects, hence we only have to check the statement for morphisms. Since all functors are additive, let $f : \operatorname{Spec}(K_1) \leftarrow \operatorname{Spec}(L) \to \operatorname{Spec}(K_2)$ be a prototypical span in $\Omega(k)$ and $\langle \alpha \rangle \in \operatorname{GW}(L), \ \alpha \in L^{\times}$. We know that f factors through some point $P = \operatorname{Spec}(k(P)) := f(\operatorname{Spec}(L))$ of $\operatorname{Spec}(K_1) \times \operatorname{Spec}(K_2)$. Then, f_* factors as

$$f_*: Z^0(\operatorname{Spec}(L)) \cong \mathbb{Z} \to Z^0(P) \cong \mathbb{Z} \xrightarrow{\operatorname{incl.}} Z^0(\operatorname{Spec}(K_1) \times \operatorname{Spec}(K_2))$$
$$[\operatorname{Spec}(L)] = 1 \mapsto [L:k(P)] \cdot P$$

and, in the oriented case, f_* factors as

$$f_* : \mathrm{GW}(L) \to \mathrm{GW}(k(P)) \xrightarrow{\mathrm{incl.}} \mathrm{GW}(\mathrm{Spec}(K_1) \times \mathrm{Spec}(K_2))$$
$$\langle \alpha \rangle \mapsto (\mathrm{Tr}_{k(P)}^L)_*(\langle \alpha \rangle).$$

The forgetful functor $V : \mathrm{GW}(k(P)) \to K_0(k(P)) \cong \mathbb{Z}$ is given by dimension and since $\dim((\mathrm{Tr}_{k(P)}^L)_*(\langle \alpha \rangle)) = [L : k(P)]$ by Remark 8.1.7, the diagram commutes. The proof extends directly to general R.

We construct the following ideal of homomorphisms inspired by [BG23a, Notation 4.15].

Definition 8.1.11. We define the two-sided ideal $\widetilde{J}_R(k)$ of homomorphisms in $\widetilde{\Omega}_R(k)$ as the ideal generated, i.e. consisting of linear combinations (with respect to the sum as defined in Construction 8.1.2) with coefficients in R, by spans of the form

$$[\operatorname{Spec}(L') \xleftarrow{\pi_L^{L'}} (\operatorname{Spec}(L), \langle \alpha \rangle) \xrightarrow{\pi_L^{L'}} \operatorname{Spec}(L')] \\ - \sum_{i=1}^{[L:L']} [\operatorname{Spec}(L') \xleftarrow{\operatorname{id}} (\operatorname{Spec}(L'), \langle \alpha_i \rangle) \xrightarrow{\operatorname{id}} \operatorname{Spec}(L')]$$

for L/L'/k finite separable field extensions of k and $\langle \alpha \rangle \in \mathrm{GW}(L) \otimes R$, where

$$\langle \alpha_1, ..., \alpha_{[L:L']} \rangle = \tilde{\varepsilon}([\operatorname{Spec}(L') \xleftarrow{\pi_L^{L'}} (\operatorname{Spec}(L), \langle \alpha \rangle) \xrightarrow{\pi_L^{L'}} \operatorname{Spec}(L')]) = (\operatorname{Tr}_{L'}^L)_*(\alpha)$$

is a matrix representation of the trace form of $\alpha \in L^{\times}$ in $\mathrm{GW}(L') \otimes R$.

Note that two such matrix representations may differ by permutation of the entries or multiplication of the *i*-th entry by a square β_i^2 for $\beta_i \in (L')^{\times}$ and $i \in \{1, ..., [L : L']\}$, see [Scha85, Lemma 3.5]. Since the addition of oriented spans is commutative, permutation of entries by some permutation π yields

$$\sum_{i=1}^{[L:L']} [\operatorname{Spec}(L') \xleftarrow{\operatorname{id}} (\operatorname{Spec}(L'), \langle \alpha_{\pi(i)} \rangle) \xrightarrow{\operatorname{id}} \operatorname{Spec}(L')]$$
$$= \sum_{i=1}^{[L:L']} [\operatorname{Spec}(L') \xleftarrow{\operatorname{id}} (\operatorname{Spec}(L'), \langle \alpha_i \rangle) \xrightarrow{\operatorname{id}} \operatorname{Spec}(L')].$$

On the other hand, multiplying α_i with a square yields isomorphic oriented spans by Construction 8.1.2, hence,

$$\sum_{i=1}^{[L:L']} [\operatorname{Spec}(L') \xleftarrow{\operatorname{id}} (\operatorname{Spec}(L'), \langle \alpha_i \beta_i^2 \rangle) \xrightarrow{\operatorname{id}} \operatorname{Spec}(L')]$$
$$= \sum_{i=1}^{[L:L']} [\operatorname{Spec}(L') \xleftarrow{\operatorname{id}} (\operatorname{Spec}(L'), \langle \alpha_i \rangle) \xrightarrow{\operatorname{id}} \operatorname{Spec}(L')].$$

Moreover, there is the Scharlau relation in GW(L') for forms of dimension 2. Let w.l.o.g. [L:L'] = 2. Then,

$$\begin{split} &[\operatorname{Spec}(L') \xleftarrow{\operatorname{id}} (\operatorname{Spec}(L'), \langle \alpha_1 \rangle) \xrightarrow{\operatorname{id}} \operatorname{Spec}(L')] \\ &+[\operatorname{Spec}(L') \xleftarrow{\operatorname{id}} (\operatorname{Spec}(L'), \langle \alpha_2 \rangle) \xrightarrow{\operatorname{id}} \operatorname{Spec}(L')] \\ &=[\operatorname{Spec}(L') \xleftarrow{\operatorname{id}} (\operatorname{Spec}(L'), \langle \alpha_1 + \alpha_2 \rangle) \xrightarrow{\operatorname{id}} \operatorname{Spec}(L')] \\ &+[\operatorname{Spec}(L') \xleftarrow{\operatorname{id}} (\operatorname{Spec}(L'), \langle \alpha_1 \alpha_2 (\alpha_1 + \alpha_2) \rangle) \xrightarrow{\operatorname{id}} \operatorname{Spec}(L')] \end{split}$$

by definition of $\widetilde{\Omega}_R(k)$. Consequently, the description of the ideal is independent of the choice of the α_i .

This yields an equivalence relation on each Hom-set of the form $\operatorname{Hom}_{\widetilde{\Omega}_R(k)}(\operatorname{Spec}(H), \operatorname{Spec}(H))$ in $\widetilde{\Omega}_R(k)$ by defining the equivalence class of f in the quotient of abelian groups $\operatorname{Hom}_{\widetilde{\Omega}_R(k)}(\operatorname{Spec}(H), \operatorname{Spec}(H))/\widetilde{J}_R(k)$ as $[f] := f + \widetilde{J}_R(k)$.

We extend this equivalence relation via composition in the following way. Let $f = f_1 \circ f_2 \circ f_3$, $n \in \mathbb{N}$ be an arbitrary composition of some morphism $f \in \text{Hom}_{\tilde{\Omega}_R(k)}(\text{Spec}(K), \text{Spec}(L))$. If f_2 is an element of $\text{Hom}_{\tilde{\Omega}_R(k)}(\text{Spec}(H), \text{Spec}(H))$ for some finite separable field extension H/k, we say

$$f \sim f_1 \circ f'_2 \circ f_3$$

for each $f'_2 \in [f_2] = f_2 + \widetilde{J}_R(k)$. We denote the quotient category defined this way by $\widetilde{\Omega}_R(k)/\widetilde{J}_R(k)$.

We need the following lemma that characterizes prototypical spans in $\Omega(k)$ just as in [BG23a, Remark 4.14], inspired by [ThWe95, Proposition 2.2].

Lemma 8.1.12. Let K and H be open subgroups of $\Gamma := \operatorname{Gal}(k/k)$ for a fixed separable closure \bar{k} of a perfect field k with $\operatorname{char}(k) \neq 2$. Every morphism in $\operatorname{Hom}_{\widetilde{\Omega}_R(k)}(\operatorname{Spec}(\bar{k}^K), \operatorname{Spec}(\bar{k}^H))$ is equivalent modulo $\widetilde{J}_R(k)$ to a linear combination of spans of the form

$$\operatorname{Spec}(\bar{k}^{K}) \xleftarrow{}{}^{g_{\pi_{K^{g}\cap H}^{K}}} \operatorname{Spec}(\bar{k}^{K^{g}\cap H}, \langle \beta \rangle) \xrightarrow{}{}^{\pi_{K^{g}\cap H}^{H}} \operatorname{Spec}(\bar{k}^{H}) ,$$

where $[g] \in K \setminus \Gamma/H$.

Proof. Recall that there is a well-known equivalence $\operatorname{Sm}_k^0 \simeq \Gamma$ -sets, by which every smooth connected 0-dimensional k-scheme is equivalent to $\operatorname{Spec}(\bar{k}^H)$ for some open subgroup $H \leq \Gamma$.

Therefore, we can apply [ThWe95, Proposition 2.2] (compare [BG23a, Remark 4.14]) and conclude that we can rewrite a prototypical span in $\Omega_R(k)$

$$\operatorname{Spec}(\bar{k}^K) \xleftarrow{g_{\pi_L^K}} \operatorname{Spec}(\bar{k}^L) \xrightarrow{\pi_L^H} \operatorname{Spec}(\bar{k}^H)$$

as the composition

$$[\operatorname{Spec}(\bar{k}^{K}) \xleftarrow{}^{g_{\pi_{K^{g}\cap H}^{K}}} \operatorname{Spec}(\bar{k}^{K^{g}\cap H}) = \operatorname{Spec}(\bar{k}^{K^{g}\cap H})]$$

$$\circ[\operatorname{Spec}(\bar{k}^{K^{g}\cap H}) \xleftarrow{}^{\pi_{L}^{K^{g}\cap H}} \operatorname{Spec}(\bar{k}^{L}) \xrightarrow{}^{\pi_{L}^{K^{g}\cap H}} \operatorname{Spec}(\bar{k}^{K^{g}\cap H})]$$

$$\circ[\operatorname{Spec}(\bar{k}^{K^{g}\cap H}) = \operatorname{Spec}(\bar{k}^{K^{g}\cap H}) \xrightarrow{}^{\pi_{K^{g}\cap H}^{H}} \operatorname{Spec}(\bar{k}^{H})]$$

and it follows that we can rewrite a prototypical span

$$\operatorname{Spec}(\bar{k}^K) \xleftarrow{g_{\pi_L^K}} (\operatorname{Spec}(\bar{k}^L), \langle \alpha \rangle) \xrightarrow{\pi_L^H} \operatorname{Spec}(\bar{k}^H)$$

in $\widetilde{\Omega}_R(k)$ as the composition

$$[\operatorname{Spec}(\bar{k}^{K}) \xleftarrow{}^{g_{\pi_{K^{g}\cap H}}} (\operatorname{Spec}(\bar{k}^{K^{g}\cap H}), \langle 1 \rangle) = \operatorname{Spec}(\bar{k}^{K^{g}\cap H})]$$

$$\circ[\operatorname{Spec}(\bar{k}^{K^{g}\cap H}) \xleftarrow{}^{\pi_{L}^{K^{g}\cap H}} (\operatorname{Spec}(\bar{k}^{L}), \langle \alpha \rangle) \xrightarrow{}^{\pi_{L}^{K^{g}\cap H}} \operatorname{Spec}(\bar{k}^{K^{g}\cap H})]$$

$$\circ[\operatorname{Spec}(\bar{k}^{K^{g}\cap H}) = (\operatorname{Spec}(\bar{k}^{K^{g}\cap H}), \langle 1 \rangle) \xrightarrow{}^{\pi_{K^{g}\cap H}^{H}} \operatorname{Spec}(\bar{k}^{H})]$$

since $\langle 1 \rangle \otimes \langle \alpha \rangle \otimes \langle 1 \rangle = \langle 1 \cdot \alpha \cdot 1 \rangle = \langle \alpha \rangle$ in $\mathrm{GW}(\bar{k}^L) \otimes R$. By definition of $\widetilde{J}_R(k)$, the middle term is equivalent modulo $\widetilde{J}_R(k)$ to

$$\sum_{i=1}^{n} [\operatorname{Spec}(\bar{k}^{K^{g} \cap H}) \leftarrow (\operatorname{Spec}(\bar{k}^{K^{g} \cap H}), \langle \alpha_{i} \rangle) \to \operatorname{Spec}(\bar{k}^{K^{g} \cap H})]$$

for $n = [\bar{k}^L : \bar{k}^{K^g \cap H}]$, where $\langle \alpha_1 \rangle \perp ... \perp \langle \alpha_n \rangle$ is the trace form of $\alpha \in L$, i.e. the image of $\langle \alpha \rangle$ under the pushforward

$$(\operatorname{Tr}_{\bar{k}^{K^g}\cap H}^{\bar{k}^L})_*: \operatorname{GW}(\bar{k}^L)\otimes R \to \operatorname{GW}(\bar{k}^{K^g\cap H})\otimes R.$$

It follows that our original span is equivalent modulo $\widetilde{J}_R(k)$ to

$$[\operatorname{Spec}(\bar{k}^{K}) \xleftarrow{}^{g_{\pi_{K^{g}\cap H}^{K}}} (\operatorname{Spec}(\bar{k}^{K^{g}\cap H}), \langle 1 \rangle) = \operatorname{Spec}(\bar{k}^{K^{g}\cap H})]$$

$$\circ \sum_{i=1}^{n} ([\operatorname{Spec}(\bar{k}^{K^{g}\cap H}) \leftarrow (\operatorname{Spec}(\bar{k}^{K^{g}\cap H}), \langle \alpha_{i} \rangle) \to \operatorname{Spec}(\bar{k}^{K^{g}\cap H})])$$

$$\circ [\operatorname{Spec}(\bar{k}^{K^{g}\cap H}) = (\operatorname{Spec}(\bar{k}^{K^{g}\cap H}), \langle 1 \rangle) \xrightarrow{}^{\pi_{K^{g}\cap H}^{H}} \operatorname{Spec}(\bar{k}^{H})]$$

which equals

$$\sum_{i=1}^{n} [\operatorname{Spec}(\bar{k}^{K}) \xleftarrow{g_{\pi_{K^{g}\cap H}}} (\operatorname{Spec}(\bar{k}^{K^{g}\cap H}), \langle \alpha_{i} \rangle) \xrightarrow{\pi_{K^{g}\cap H}^{H}} \operatorname{Spec}(\bar{k}^{H})].$$

Remark 8.1.13. At this point, we would have liked to have proven an analog of [BG23a, Proposition 4.17], namely that the functor $\tilde{\varepsilon} : \tilde{\Omega}_R(k) \to \widetilde{\operatorname{Cor}}_{k,R}^0$ induces an equivalence of additive categories

$$\widetilde{\varepsilon}': \frac{\widetilde{\Omega}_R(k)}{\widetilde{J}_R(k)} \xrightarrow{\simeq} \widetilde{\operatorname{Cor}}_{k,R}^0.$$

The idea was to construct an R-submodule \widetilde{M} of $\operatorname{Hom}_{\widetilde{\Omega}_R(k)}(\operatorname{Spec}(\bar{k}^K), \operatorname{Spec}(\bar{k}^H))$ generated by spans of the form

$$s_g^{\alpha} := [\operatorname{Spec}(\bar{k}^K) \xleftarrow{g_{\pi_{Kg_{\cap H}}^K}} (\operatorname{Spec}(\bar{k}^{K^g \cap H}), \langle \alpha \rangle) \xrightarrow{\pi_{Kg_{\cap H}}^H} \operatorname{Spec}(\bar{k}^H)]$$

for $[g] \in K \setminus \Gamma / H$ and we wanted to show that

$$\operatorname{Hom}_{\widetilde{\operatorname{Cor}}_{k}^{0}\otimes R}(\operatorname{Spec}(\bar{k}^{K}),\operatorname{Spec}(\bar{k}^{H}))\cong \widetilde{M}\cong \frac{\operatorname{Hom}_{\widetilde{\Omega}_{R}(k)}(\operatorname{Spec}(\bar{k}^{K}),\operatorname{Spec}(\bar{k}^{H}))}{\widetilde{J}_{R}(k)},$$

where the composition of the isomorphisms is given by ε' . However, it is not clear whether

$$\tilde{\varepsilon}: \widetilde{M} \xrightarrow{\tilde{\varepsilon}} \operatorname{Hom}_{\widetilde{\operatorname{Cor}}_{k,R}^0}(\operatorname{Spec}(\bar{k}^K), \operatorname{Spec}(\bar{k}^H)) \cong \operatorname{GW}(\bar{k}^K \otimes \bar{k}^H) \otimes R$$

is injective in general.

8.2 Milnor-Witt Mackey Functors

Having constructed the span category $\widehat{\Omega}(k)$ and the passage to finite MW correspondences, we now define (cohomological) Mackey functors. We define Mackey functors in a different way than the direct analog of [BG23a, Definition 4.5] since its oriented version is not as concrete as its non-oriented counterpart. Our definition is inspired by [KY15, Proposition 2], which states an equivalent definition of cohomological Mackey functors via factorizations. One could show these definitions coincide in the Milnor–Witt case as well, but this would require some form of an equivalence as discussed in Remark 8.1.13.

Definition 8.2.1. Let k be a perfect field of characteristic $\neq 2$ and R a commutative ring. We define *MW Mackey functors* to be additive functors

$$(\widetilde{\Omega}(k))^{\mathrm{op}} \to R\text{-}\mathrm{Mod}$$
.

A cohomological MW Mackey functor is a MW Mackey functor that factors through $\widetilde{\operatorname{Cor}}_k^0$ via $\tilde{\varepsilon}$. We will denote the abelian categories of (cohomological) MW Mackey functors by $\widetilde{\operatorname{Mack}}^{\operatorname{coh}}(k) \subset \widetilde{\operatorname{Mack}}(k)$.

As in the non-oriented case, we may extend scalars to R, denoting the resulting categories by $\widetilde{\operatorname{Mack}}_{R}^{\operatorname{coh}}(k) \subset \widetilde{\operatorname{Mack}}_{R}(k)$. The (cohomological) MW Mackey functors in this case are R-linear additive functors $\widetilde{\Omega}_{R}(k)^{\operatorname{op}} \to R$ - Mod (which factor through $\widetilde{\operatorname{Cor}}_{k,R}^{0}$).

Conjecture 8.2.2. The factorization via $\tilde{\varepsilon}$ is unique.

The conjecture is true if the equivalence from Remark 8.1.13 holds, but we do not expect the latter to be true in this form. However, it might still be possible to show the uniqueness of the factorization independently.

For the rest of this work, we assume Conjecture 8.2.2 to be true.

The vertical equivalence of Diagram 7.1 is the following. After forgetting \sim , the result can be considered a refinement of [BG23a, Corollary 6.17].

Corollary 8.2.3. We assume that Conjecture 8.2.2 holds. Let k be a perfect field of characteristic $\neq 2$ and R a commutative ring. Consider the functor

$$\tilde{\Phi}: \widetilde{\operatorname{Mack}}_{R}^{\operatorname{coh}}(k) \to \operatorname{PSh}_{R}(\widetilde{\operatorname{Cor}}_{k,R}^{0})$$

sending a cohomological MW Mackey functor M that factors as



to $\tilde{\Phi}(M) := M'$ and a natural transformation $\varphi : M_1 \Rightarrow M_2$, defined via $\varphi_X : M_1(X) \to M_2(X)$ for all objects X in $\tilde{\Omega}(k)$, to $\tilde{\Phi}(\varphi) := \varphi' : M'_1 \Rightarrow M'_2$ defined via $\varphi'_{\tilde{\varepsilon}(X)} : M'_1(\tilde{\varepsilon}(X)) \to M'_2(\tilde{\varepsilon}(X)).$

It is an equivalence of Grothendieck abelian categories.

Proof. The factorization is unique by Conjecture 8.2.2, so the statement follows directly. \Box

Remark 8.2.4. Note that M' is, in particular, a Nisnevish sheaf with MW-transfers since every presheaf on $\widetilde{\operatorname{Cor}}_{k,R}^0$ is automatically one by Remark 7.2.5. The above equivalence hence extends to an equivalence

$$\widetilde{\operatorname{Mack}}_{R}^{\operatorname{coh}}(k) \xrightarrow{\simeq} \operatorname{Sh}_{\operatorname{Nis}}(\widetilde{\operatorname{Cor}}_{k,R}^{0}) = \operatorname{PSh}_{R}(\widetilde{\operatorname{Cor}}_{k,R}^{0}).$$

Let us say a few words about representation theoretical analogs.

We denote by $\Gamma := \operatorname{Gal}(\overline{k}/k)$ the absolute Galois group of a perfect field k of characteristic $\neq 2$ for a fixed algebraic closure \overline{k} . The category $\widetilde{\Omega}(k)$ from Construction 8.1.2 has a representation theoretical analog. For this, we recall the explicit description of the following equivalence (on the left-hand side determined by evaluation on additive generators)

$$\begin{array}{ccc} \operatorname{Sm}_k^0 \xrightarrow{\simeq} \Gamma \text{-sets} \\ X \mapsto X(\bar{k}) \\ \operatorname{Spec}(\bar{k}^H) \longleftrightarrow \Gamma/H. \end{array}$$

Using this equivalence, we can directly transfer Construction 8.1.2 generalizing [BG23a, Definition 4.5/Recollection 4.2]. The definition is merely sketched here.

Definition 8.2.5. Let $\Gamma := \operatorname{Gal}(\overline{k}/k)$ be the absolute Galois group of a perfect field k of characteristic $\neq 2$ for a fixed algebraic closure \overline{k} and R a commutative ring. Analogously to Construction 8.1.2, we define the category $\widetilde{\Omega}(\Gamma)$ as the group completion of the category $\widetilde{\operatorname{span}}(\Gamma)$ with objects finite Γ -sets, i.e. finite direct sums of finite permutation modules isomorphic to $R(\Gamma/H)$.

A morphism from X to Y is a span $X \leftarrow Z := \coprod_{i=1}^{n} R(\Gamma/H_i) \to Y$ together with a one-dimensional symmetric bilinear form $\langle \alpha_i \rangle \in \mathrm{GW}(\bar{k}^{H_i})$ for each *i*. Composition and addition are defined analogously to Construction 8.1.2; extension of scalars yields an additive category $\widetilde{\Omega}_R(\Gamma)$.

Remark 8.2.6. Let $\Gamma := \operatorname{Gal}(\bar{k}/k)$ be the absolute Galois group of a perfect field k of characteristic $\neq 2$ for a fixed algebraic closure \bar{k} . The equivalence of categories Γ -sets $\simeq \operatorname{Sm}_k^0$ induces equivalences of additive categories

$$\Omega(\Gamma) \simeq \Omega(k)$$

and

$$\tilde{\Omega}(\Gamma) \simeq \tilde{\Omega}(k)$$

and similarly after extending scalars to R.

Remark 8.2.7. let $\Gamma := \operatorname{Gal}(k/k)$ be the absolute Galois group of a perfect field k of characteristic $\neq 2$ for a fixed algebraic closure \bar{k} . One could (forcably) define the additive category $\operatorname{perm}(\Gamma; R)$ as follows. Objects are free R-modules of the form R(X) for X in Γ -sets. The morphism group between additive generators $R(\Gamma/K)$ and $R(\Gamma/H)$ is defined as

$$\operatorname{Hom}_{\widetilde{\operatorname{perm}}(\Gamma;R)}(R(\Gamma/K), R(\Gamma/H)) := \operatorname{GW}(\bar{k}^K \otimes \bar{k}^H) \otimes R.$$

Since $\operatorname{Hom}_{\widetilde{\operatorname{perm}}(\Gamma;R)}(R(\Gamma/K), R(\Gamma/H)) \cong \widetilde{\operatorname{CH}}^0(\operatorname{Spec}(\bar{k}^K) \times \operatorname{Spec}(\bar{k}^H)) \otimes R$, we set the composition as the restriction of the usual composition in $\widetilde{\operatorname{Cor}}_{k,R}^0$ introduced in [CF22, Section 4.2]. Immediately, one could generalize [BG23a, Proposition 6.14] to obtain an equivalence of *R*-linear tensor categories

$$\widetilde{\operatorname{Cor}}_k^0 \otimes R \simeq \widetilde{\operatorname{perm}}(\Gamma; R).$$

Then, we could define *cohomological MW Mackey functors* to be additive functors

$$(\widetilde{\Omega}_R(\Gamma))^{\mathrm{op}} \to R\text{-}\mathrm{Mod}$$
.

factoring (conjecturally uniquely) through $\widetilde{\text{perm}}(\Gamma; R)$ and denote the resulting abelian categories of cohomological MW Mackey functors by $\widetilde{\text{Mack}}_{R}^{\text{coh}}(\Gamma)$. It would immediately follow that we have equivalences of abelian categories

$$\widetilde{\operatorname{Mack}}_{R}^{\operatorname{coh}}(\Gamma) \simeq \widetilde{\operatorname{Mack}}_{R}^{\operatorname{coh}}(k) \simeq \operatorname{Sh}_{\operatorname{Nis}}(\widetilde{\operatorname{Cor}}_{k,R}^{0}).$$

However, the results only transfer directly from $\widetilde{\operatorname{Cor}}_{k,R}^0$ and $\widetilde{\Omega}(k)$, and not much is won by this definition of $\widetilde{\operatorname{perm}}(\Gamma; R)$.

We have not found a sensible way how to equip morphisms of R-modules with symmetric bilinear forms without using the equivalence $\operatorname{Sm}_k^0 \simeq \Gamma$ -sets. For this reason, we do not have an intrinsic candidate for $\widetilde{\operatorname{perm}}(\Gamma; R)$ and especially not for $\widetilde{\operatorname{Perm}}(\Gamma; R)$ or $\widetilde{\operatorname{Mod}}(\Gamma; R)$.

In the end, our computations in Chapter 9 do not rely on representation theoretical results as opposed to computations in [BG22a].

Chapter 9

The Balmer Spectrum of Artin Milnor–Witt Motives

We now want to compute Balmer spectra for initial examples. In Section 9.1, we determine, for example, the spectrum of (geometric) Artin Milnor–Witt motives over algebraically closed base fields. Then, Section 9.2 treats the base field \mathbb{R} with coefficients in $\mathbb{Z}/2$. We will assume all base fields k (and sometimes L) to be infinite. Following Remark 7.3.4, it may be possible to drop this assumption in the future.

9.1 Algebraically Closed Fields and First Computations

As expected, for algebraically closed fields with arbitrary coefficients, the derived category of Milnor–Witt Artin motives is equivalent to Voevodsky's derived category of Artin motives.

Lemma 9.1.1. Let k be an algebraically closed field of characteristic $\neq 2$ and R a commutative ring. Then, we have an equivalence of tensor triangulated categories $DAM^{gm}(k; R) \simeq \widetilde{DAM}^{gm}(k; R)$. If Conjecture 7.3.10 holds, then $DAM(k; R) \simeq \widetilde{DAM}(k; R)$.

Proof. First, we show that $\widetilde{\operatorname{Cor}}_{k,R}^0 \simeq \operatorname{Cor}_{k,R}^0$. Let $X, Y \in \operatorname{Sm}_k^0$ and $T \subset X \times Y$ an admissible subset. Then, $X \times Y$ is zero-dimensional by Remark 8.1.1. It follows from [Fas08, Remarque 10.2.16] that

$$\widetilde{\operatorname{CH}}^0_T(X \times Y) \otimes R \cong \operatorname{CH}^0_T(X \times Y) \otimes R,$$

since k is algebraically closed. In particular, $\widetilde{\text{Cor}}_{k,R}^0 \simeq \text{Cor}_{k,R}^0$ by [CF22, Remark 4.3.3]. The result now follows from Proposition 7.3.3 and Corollary 7.3.12.

Corollary 9.1.2. Let k be an algebraically closed field of characteristic $\neq 2$ and R a commutative ring. The Balmer spectra of $\widetilde{DAM}^{gm}(k; R)$ and $DAM^{gm}(k; R)$ are homeomorphic to $\operatorname{Spec}(R)$.

Proof. Follows from Lemma 9.1.1 and the observation from [BG22b, Corollary 2.9] that $\operatorname{Spc}(\operatorname{DAM}^{\operatorname{gm}}(k; R)) = \operatorname{Spc}(\operatorname{D}^{\operatorname{perf}}(R))$.

Moreover, we obtain a refinement of [BG22b, Proposition 2.8].

Proposition 9.1.3. Let L be a finite separable field extension of k, both L and k of characteristic $\neq 2$, and K a perfect field such that the trace form $(\operatorname{Tr}_{k}^{L})_{1}$ is invertible in $\operatorname{GW}(k) \otimes K$. Then, the map

$$\operatorname{Spc}(\widetilde{\operatorname{DAM}}^{\operatorname{gm}}(L;K)) \to \operatorname{Spc}(\widetilde{\operatorname{DAM}}^{\operatorname{gm}}(k;K))$$

induced by the extension of scalars functor

$$\widetilde{\operatorname{Ext}}_k^L: \widetilde{\operatorname{Cor}}_{k,K}^0 \to \widetilde{\operatorname{Cor}}_{L,K}^0$$

is surjective.

Proof. We follow an argument similar to the standard argument in the proof of [BG22b, Proposition 2.8]. The extension functor $\widetilde{\operatorname{Ext}}_k^L$ admits a left adjoint $\widetilde{\operatorname{Res}}_k^L$ with unit η and counit ϵ which is also a right adjoint with unit $\tilde{\eta}$ and counit $\tilde{\epsilon}$, see [CF22, Section 6.2]. It is a tensor functor by [CF22, Section 6.2, p.32].

We obtain adjunctions after applying $K_b((-)^{\natural})$ on both sides. Using [CF22, Lemma 6.2.2], we see that the composition

$$\epsilon \circ \tilde{\eta} : \mathrm{id}_{\mathrm{K}_b((\widetilde{\mathrm{Cor}}_{k,K}^0)^{\natural})} \to \widetilde{\mathrm{Res}}_k^L \circ \widetilde{\mathrm{Ext}}_k^L \to \mathrm{id}_{\mathrm{K}_b((\widetilde{\mathrm{Cor}}_{k,K}^0)^{\natural})}$$

is given by (degreewise) multiplication as a $(\mathrm{GW}(k) \otimes K)$ -module by the trace form $(\mathrm{Tr}_k^L)_1 \in \mathrm{GW}(k) \otimes K$. Since $(\mathrm{Tr}_k^L)_1$ is invertible in $\mathrm{GW}(k) \otimes K$ by assumption, $\epsilon \circ \tilde{\eta}$ is also invertible. Here, we use that composition in $\mathrm{Hom}_{\widetilde{\mathrm{Cor}}_{k,K}^0}(\mathrm{Spec}(k), \mathrm{Spec}(k))$ coincides with multiplication in $\mathrm{GW}(k) \otimes K$ by [Fas07, Theorem 7.6].

It follows that $\tilde{\eta}_A$ is injective for all objects A in $K_b((\widetilde{\operatorname{Cor}}_{k,K}^0)^{\natural})$. Consequently, the functor $\widetilde{\operatorname{Ext}}_k^L : K_b((\widetilde{\operatorname{Cor}}_{k,K}^0)^{\natural}) \to K_b((\widetilde{\operatorname{Cor}}_{L,K}^0)^{\natural})$ is faithful. Moreover, it is a tensor triangulated functor since it is a tensor functor before applying $K_b((-)^{\natural})$. The statement now follows from [Bal18, Theorem 1.3] and Proposition 7.3.3. \Box

Remark 9.1.4. Unlike in the non-oriented case, it is not an immediate corollary of this proposition that the spectrum of $\widetilde{DAM}^{gm}(\mathbb{R}; K)$ is homeomorphic to a point when 2 is invertible in K (compare [BG22b, Corollary 2.9]). The reason for this is that the field extension \mathbb{C}/\mathbb{R} does not fulfill the assumption that the trace form is invertible.

Recall from Example 8.1.8 (ii) that $(\operatorname{Tr}_{\mathbb{R}}^{\mathbb{C}})_1 = \langle 1, -1 \rangle \in \operatorname{GW}(\mathbb{R}) \otimes K$, which is indeed not invertible. It is not even a monomorphism when considered an morphism in $\operatorname{Hom}_{\operatorname{Cor}_{\mathbb{R},K}^0}(\operatorname{Spec}(\mathbb{R}), \operatorname{Spec}(\mathbb{R})) \cong \operatorname{GW}(\mathbb{R}) \otimes K$ since we have

$$\langle 1, -1 \rangle \langle 1 \rangle = \langle 1, -1 \rangle$$

and

$$\langle 1, -1 \rangle \langle -1 \rangle = \langle -1, 1 \rangle = \langle 1, -1 \rangle$$

by [Lam05, Proposition II.3.2] and [Scha85, Definition 1.4/1.7], but $\langle 1 \rangle \neq \langle -1 \rangle \in$ GW(\mathbb{R}) $\otimes K$. Here, we used that the composition in Hom_{$\widetilde{Cor}_{\mathbb{R},K}^0$} (Spec(\mathbb{R}), Spec(\mathbb{R})) coincides with the multiplication in GW(\mathbb{R}) $\otimes K$ by [Fas07, Theorem 7.6].

9.2 The Base Field \mathbb{R} and Coefficients in \mathbb{F}_2

Let $K = \mathbb{F}_2$ in the following. We want take initial steps to compute the Balmer spectrum of geometric Artin Milnor–Witt motives $\widetilde{DAM}^{gm}(\mathbb{R}; K)$ over the base field \mathbb{R} with coefficients in K.

Recall from Proposition 7.3.3 that we have a categorical equivalence

$$\widetilde{\mathrm{DAM}}^{\mathrm{gm}}(\mathbb{R};K)\simeq \mathrm{K}_b((\widetilde{\mathrm{Cor}}^0_{\mathbb{R},K})^{\natural})$$

since \mathbb{R} is perfect infinite of characteristic $\neq 2$.

Remark 9.2.1. In the non-oriented case, these categories are equivalent to the bounded homotopy category of the abelian category KC_2 -mod of finitely generated KC_2 -modules by [BG23a, Proposition 6.14] since every KC_2 -module is already a permutation module. Moreover, [BG22a, Remark 10.5] recalls that there is a categorical equivalence between KC_2 -mod and the category of Artin Chow motives $AM(\mathbb{R}, K)$. Its oriented analogon is precisely $\widetilde{Cor}_{\mathbb{R},K}^0$, which is why $\widetilde{AM}(\mathbb{R}, K)$ would also be a suitable notation for $\widetilde{Cor}_{\mathbb{R},K}^0$.

Remark 9.2.2. An additive category is a Krull-Schmidt category if the endomorphism ring of every indecomposable object is a local ring, see [Rin84, p.52]. By definition, every object in $\widetilde{\operatorname{Cor}}_{\mathbb{R},K}^0$ and $\operatorname{Cor}_{\mathbb{R},K}^0$ is a finite direct sum of copies of $\operatorname{Spec}(\mathbb{R})$ and $\operatorname{Spec}(\mathbb{C})$. The latter is a Krull-Schmidt category, see [BG22b, Remark 3.1]. It follows that $\operatorname{End}_{\widetilde{\operatorname{Cor}}_{\mathbb{R},K}^0}(\operatorname{Spec}(\mathbb{C})) \cong \operatorname{End}_{\operatorname{Cor}_{\mathbb{R},K}^0}(\operatorname{Spec}(\mathbb{C}))$ is a local ring. Moreover, the endomorphism ring $\operatorname{End}_{\widetilde{\operatorname{Cor}}_{\mathbb{R},K}^0}(\operatorname{Spec}(\mathbb{R})) \cong K[C_2]$ is local as well. Consequently, $\widetilde{\operatorname{Cor}}_{\mathbb{R},K}^0$ is a Krull-Schmidt category with indecomposable objects $\operatorname{Spec}(\mathbb{R})$ and $\operatorname{Spec}(\mathbb{C})$.

The category $\operatorname{Cor}^0_{\mathbb{R},K} \simeq KC_2$ -mod is abelian, but problems appear for its oriented counterpart.

Lemma 9.2.3. The category $\widetilde{\operatorname{Cor}}_{\mathbb{R},K}^0$ is not abelian or the forgetful functor π : $\widetilde{\operatorname{Cor}}_{\mathbb{R},K}^0 \to \operatorname{Cor}_{\mathbb{R},K}^0$ is not exact.

Proof. Suppose $\widetilde{\operatorname{Cor}}_{\mathbb{R},K}^0$ is abelian and π is exact. We want to show that the morphism $H := \langle 1, -1 \rangle \in \operatorname{GW}(\mathbb{R}) \otimes K \cong \operatorname{Hom}_{\widetilde{\operatorname{Cor}}_{\mathbb{R},K}^0}(\operatorname{Spec}(\mathbb{R}), \operatorname{Spec}(\mathbb{R}))$ does not have a kernel.

Since π is exact by assumption, it commutes with kernels and we have

$$\pi(\ker(H)) = \ker(\pi(H)) = \ker([\operatorname{Spec}(\mathbb{R}) \xrightarrow{0} \operatorname{Spec}(\mathbb{R})])$$
$$= [\operatorname{Spec}(\mathbb{R}) \xrightarrow{\operatorname{id}} \operatorname{Spec}(\mathbb{R})] \in \operatorname{Hom}_{\operatorname{Cor}^{0}_{\mathbb{R},K}}(\operatorname{Spec}(\mathbb{R}), \operatorname{Spec}(\mathbb{R})).$$

Since $K = \mathbb{F}_2$, the identity morphism on $\operatorname{Spec}(\mathbb{R})$ has exactly two lifts under π in $\operatorname{Hom}_{\operatorname{Cor}_{\mathbb{R},K}^0}(\operatorname{Spec}(\mathbb{R}), \operatorname{Spec}(\mathbb{R})) \cong \operatorname{GW}(\mathbb{R}) \otimes K$, namely $\langle 1 \rangle$ and $\langle -1 \rangle$. Recall that composition in $\operatorname{Hom}_{\operatorname{Cor}_{\mathbb{R},K}^0}(\operatorname{Spec}(\mathbb{R}), \operatorname{Spec}(\mathbb{R}))$ coincides with multiplication in $\operatorname{GW}(\mathbb{R}) \otimes K$ by [Fas07, Theorem 7.6]. Since $\langle 1 \rangle \cdot \langle 1, -1 \rangle = \langle -1 \rangle \cdot \langle 1, -1 \rangle =$ $\langle 1, -1 \rangle \neq 0$, the diagram



does not commute for $\kappa \in \{\langle 1 \rangle, \langle -1 \rangle\}$, which contradicts the definition of a kernel in an abelian category, see [MacL71, Section VIII.1].

Example 9.2.4. Let $\langle 1, -1 \rangle \in \operatorname{Hom}_{\widetilde{\operatorname{Cor}}_{\mathbb{R},K}^{0}}(\operatorname{Spec}(\mathbb{R}), \operatorname{Spec}(\mathbb{R})) \cong \operatorname{GW}(\mathbb{R}) \otimes K$. We want to show that $\langle 1, -1 \rangle \circ \langle 1, -1 \rangle = 0 \in \operatorname{Hom}_{\widetilde{\operatorname{Cor}}_{\mathbb{R},K}^{0}}(\operatorname{Spec}(\mathbb{R}), \operatorname{Spec}(\mathbb{R})) \cong \operatorname{GW}(\mathbb{R}) \otimes K$.

The composition in $\operatorname{Hom}_{\widetilde{\operatorname{Cor}}_{\mathbb{R},K}^{0}}(\operatorname{Spec}(\mathbb{R}), \operatorname{Spec}(\mathbb{R}))$ coincides with the multiplication in $\operatorname{GW}(\mathbb{R}) \otimes K$ by [Fas07, Theorem 7.6]. By [Scha85, §2, Definition 1.4/1.7], the multiplication $\langle 1, -1 \rangle \cdot \langle 1, -1 \rangle$ in $\operatorname{GW}(\mathbb{R}) \otimes K$ is defined as

$$\langle 1, -1 \rangle \cdot \langle 1, -1 \rangle := \langle 1, -1, -1, 1 \rangle,$$

which equals $2\langle 1 \rangle + 2\langle -1 \rangle$. Since K is of characteristic 2, it follows that $\langle 1, -1 \rangle \cdot \langle 1, -1 \rangle = 0 \in \text{GW}(\mathbb{R}) \otimes K$.

We want to take first steps in computing the Balmer spectrum of

$$\widetilde{\mathrm{DAM}}^{\mathrm{gm}}(\mathbb{R};K)\simeq \mathrm{K}_b((\widetilde{\mathrm{Cor}}^0_{\mathbb{R},K})^{\natural}).$$

It suffices to consider the tensor triangulated category $K_b(\widetilde{\operatorname{Cor}}_{\mathbb{R},K}^0)$ since idempotentcompletion does not change the Balmer spectrum by [Bal05b, Proposition 3.13].

In order to determine some points of $\operatorname{Spc}(\operatorname{K}_b(\operatorname{Cor}^0_{\mathbb{R},K}))$, one can begin with computing preimages of tensor prime ideals under the tensor triangulated forgetful functor

$$\pi^*: \mathrm{K}_b(\widetilde{\mathrm{Cor}}^0_{\mathbb{R},K}) \to \mathrm{K}_b(\mathrm{Cor}^0_{\mathbb{R},K}),$$

which is defined by applying $\pi : \widetilde{\operatorname{Cor}}^0_{\mathbb{R},K} \to \operatorname{Cor}^0_{\mathbb{R},K}$ at each stage by [BC+22, (MW1)]. It induces a continuous map

$$\operatorname{Spc}(\pi^*) : \operatorname{Spc}(\operatorname{K}_b(\operatorname{Cor}^0_{\mathbb{R},K})) \to \operatorname{Spc}(\operatorname{K}_b(\operatorname{\widetilde{Cor}}^0_{\mathbb{R},K}))$$

 $\mathcal{P} \mapsto (\pi^*)^{-1}(\mathcal{P})$

by [Bal05b, Proposition 3.6].

The left-hand side consists of the three tensor prime ideals $\langle M(\operatorname{Spec}(\mathbb{C}))\rangle, \langle S\rangle$, and $\langle S, M(\operatorname{Spec}(\mathbb{C}))\rangle$ as shown in [BG22a, Theorem 3.14]. Here, $M(\operatorname{Spec}(\mathbb{C}))$ is $\operatorname{Spec}(\mathbb{C})$ considered a complex concentrated in degree 0 and

 $S = \dots \to 0 \to \operatorname{Spec}(\mathbb{R}) \xrightarrow{\eta} \operatorname{Spec}(\mathbb{C}) \xrightarrow{\epsilon} \operatorname{Spec}(\mathbb{R}) \to 0 \to \dots$

is concentrated in degrees 0, 1, and 2, see [BG22a, (3.4)].

Notation 9.2.5. The notation $\langle C \rangle$ for a set of objects C in $K_b(\operatorname{Cor}_{\mathbb{R},K}^0)$ stands for the thick tensor ideal generated by C, i.e., the smallest thick tensor ideal containing C.

We abuse notation here by writing $M(\operatorname{Spec}(\mathbb{C}))$ for an object in $\operatorname{K}_b(\operatorname{Cor}_{\mathbb{R},K}^0)$, which originally denotes the motive of $\operatorname{Spec}(\mathbb{C})$ in $\operatorname{DM}(\mathbb{R}; K)$. Similarly, we denote by $\widetilde{M}(\operatorname{Spec}(\mathbb{C}))$ not only the Milnor-Witt motive of $\operatorname{Spec}(\mathbb{C})$ in $\widetilde{\operatorname{DM}}(\mathbb{R}; K)$, but also the corresponding complex in $\operatorname{K}_b(\operatorname{Cor}_{\mathbb{R},K}^0)$, i.e., the complex $\operatorname{Spec}(\mathbb{C})$ concentrated in degree 0. Translating their notation from the setting of representation theory to algebraic geometry, some authors, e.g. in [BG22a, Section 3], use the notation $\operatorname{Spec}(\mathbb{C})$ for the complex $\operatorname{Spec}(\mathbb{C})$ concentrated in degree 0. We choose a different notation to better identify whether me mean an object in the oriented or nonoriented case.

As a consequence, in order to determine some points of $\operatorname{Spc}(\operatorname{K}_b(\operatorname{Cor}^0_{\mathbb{R},K}))$, we need to compute

$$(\pi^*)^{-1}(\langle M(\operatorname{Spec}(\mathbb{C}))\rangle), (\pi^*)^{-1}(\langle S\rangle), \text{ and } (\pi^*)^{-1}(\langle S, M(\operatorname{Spec}(\mathbb{C}))\rangle).$$

In the light of [BG22a, Corollary 3.15], this amounts to computing the kernels of (see loc.cit. for the notation in the third composition)

$$\mathrm{K}_{b}(\widetilde{\mathrm{Cor}}^{0}_{\mathbb{R},K}) \xrightarrow{\pi^{*}} \mathrm{K}_{b}(\mathrm{Cor}^{0}_{\mathbb{R},K}) \xrightarrow{\mathrm{ext}^{\mathbb{R}}_{\mathbb{C}}} \mathrm{K}_{b}(\mathrm{Cor}^{0}_{\mathbb{C},K}) \simeq \mathrm{K}_{b}(K\operatorname{-mod}),$$

 $\mathrm{K}_{b}(\widetilde{\mathrm{Cor}}^{0}_{\mathbb{R},K}) \xrightarrow{\pi^{*}} \mathrm{K}_{b}(\mathrm{Cor}^{0}_{\mathbb{R},K}) \xrightarrow{q} \mathrm{D}_{b}(\mathrm{Cor}^{0}_{\mathbb{R},K}) \simeq \mathrm{D}_{b}(KC_{2}\operatorname{-}\mathrm{mod}) \xrightarrow{\mathrm{stab}} KC_{2}\operatorname{-}\mathrm{stab},$

and

$$\mathrm{K}_{b}(\widetilde{\mathrm{Cor}}^{0}_{\mathbb{R},K}) \xrightarrow{\pi^{*}} \mathrm{K}_{b}(\mathrm{Cor}^{0}_{\mathbb{R},K}) \simeq \mathrm{K}_{b}(KC_{2}\operatorname{-}\mathrm{mod}) \xrightarrow{\mathrm{K}_{b}(\mathrm{stab})} \mathrm{K}_{b}(K\operatorname{-}\mathrm{mod}),$$

where $\operatorname{ext}_{\mathbb{C}}^{\mathbb{R}}$ denotes the functor induced by extension of scalars.

Let us focus on some intereseting objects in the category $\operatorname{Ch}_b(\operatorname{Cor}^0_{\mathbb{R},K})$ that have multiple or no lifts in $\operatorname{Ch}_b(\widetilde{\operatorname{Cor}}^0_{\mathbb{R},K})$.

Example 9.2.6. Let $K = \mathbb{F}_2$.

(i) The composition $\operatorname{Spec}(\mathbb{R}) \xrightarrow{\eta} \operatorname{Spec}(\mathbb{C}) \xrightarrow{\epsilon} \operatorname{Spec}(\mathbb{R})$ in $\operatorname{Cor}^{0}_{\mathbb{R},K}$ appearing in the acyclic complex S has a unique lift

$$\operatorname{Spec}(\mathbb{R}) \xrightarrow{\langle 1 \rangle} \operatorname{Spec}(\mathbb{C}) \xrightarrow{\langle 1 \rangle} \operatorname{Spec}(\mathbb{R})$$

to $\widetilde{\operatorname{Cor}}^0_{\mathbb{R},K}$ since there are isomorphisms

$$\operatorname{Hom}_{\operatorname{\widetilde{Cor}}^{0}_{\mathbb{R},K}}(\operatorname{Spec}(\mathbb{R}),\operatorname{Spec}(\mathbb{C})) \cong \operatorname{GW}(\mathbb{C}) \otimes K$$
$$\cong K \cong \operatorname{Hom}_{\operatorname{Cor}^{0}_{\mathbb{R},K}}(\operatorname{Spec}(\mathbb{R}),\operatorname{Spec}(\mathbb{C}))$$

and

$$\operatorname{Hom}_{\operatorname{\widetilde{Cor}}^{0}_{\mathbb{R},K}}(\operatorname{Spec}(\mathbb{C}),\operatorname{Spec}(\mathbb{R})) \cong \operatorname{GW}(\mathbb{C}) \otimes K$$
$$\cong K \cong \operatorname{Hom}_{\operatorname{Cor}^{0}_{\mathbb{R},K}}(\operatorname{Spec}(\mathbb{C}),\operatorname{Spec}(\mathbb{R})).$$

However, we have seen in Example 8.1.8 (ii) that the composition is $\langle 1, -1 \rangle \neq 0 \in \mathrm{GW}(\mathbb{R})$. Consequently, the complex S does not give rise to a complex in $\mathrm{Ch}_b(\mathrm{Cor}^0_{\mathbb{R},K})$.

(ii) On the other hand, the composition

$$\operatorname{Spec}(\mathbb{R}) \xrightarrow{\langle 1,-1 \rangle} \operatorname{Spec}(\mathbb{R}) \xrightarrow{\langle 1,-1 \rangle} \operatorname{Spec}(\mathbb{R})$$

in $\widetilde{\mathrm{Cor}}^0_{\mathbb{R},K}$ yields a complex (the composition is 0 by Example 9.2.4)

$$H^{\bullet}: \dots \to 0 \to \operatorname{Spec}(\mathbb{R}) \xrightarrow{\langle 1, -1 \rangle} \operatorname{Spec}(\mathbb{R}) \xrightarrow{\langle 1, -1 \rangle} \operatorname{Spec}(\mathbb{R}) \to 0 \to \dots,$$

which is not chain homotopy equivalent by the considerations below to

$$C^{\bullet} : \dots \to 0 \to \operatorname{Spec}(\mathbb{R}) \xrightarrow{0} \operatorname{Spec}(\mathbb{R}) \xrightarrow{0} \operatorname{Spec}(\mathbb{R}) \to 0 \to \dots$$

The forgetful functor $\pi : \widetilde{\operatorname{Cor}}^0_{\mathbb{R},K} \to \operatorname{Cor}^0_{\mathbb{R},K}$ sends both compositions $\langle 1, -1 \rangle \circ \langle 1, -1 \rangle$ and $0 \circ 0$ to

$$\operatorname{Spec}(\mathbb{R}) \xrightarrow{0} \operatorname{Spec}(\mathbb{R}) \xrightarrow{0} \operatorname{Spec}(\mathbb{R}),$$

which gives rise to a complex in $\operatorname{Ch}_b(\operatorname{Cor}^0_{\mathbb{R},K})$.

Let us check that H^{\bullet} and C^{\bullet} are indeed not chain homotopy equivalent. If they were, we could find chain maps $f: H^{\bullet} \to C^{\bullet}$ and $g: C^{\bullet} \to H^{\bullet}$ such that $f \circ g \sim \operatorname{id}_{C^{\bullet}}$ and $g \circ f \sim \operatorname{id}_{H^{\bullet}}$. In order to make the diagrams mandatory for chain maps appearing in

$$0 \longrightarrow \operatorname{Spec}(\mathbb{R}) \xrightarrow{H} \operatorname{Spec}(\mathbb{R}) \xrightarrow{H} \operatorname{Spec}(\mathbb{R}) \longrightarrow 0$$
$$f_1 \downarrow \uparrow^{g_1} \qquad f_2 \downarrow \uparrow^{g_2} \qquad f_3 \downarrow \uparrow^{g_3}$$
$$0 \longrightarrow \operatorname{Spec}(\mathbb{R}) \xrightarrow{0} \operatorname{Spec}(\mathbb{R}) \xrightarrow{0} \operatorname{Spec}(\mathbb{R}) \longrightarrow 0$$
commutative, the concrete description of the Hom-sets in $\widetilde{\operatorname{Cor}}_{\mathbb{R},\mathbb{F}_2}^0$ implies that $g_1, g_2, f_2, f_3 \in \{0, \langle 1, -1 \rangle\}$. In particular, $g_2 \circ f_2 = f_2 \circ g_2 = 0$.

A chain homotopy equivalence between id_{C^\bullet} and $f\circ g$ is a collection of maps $h_i:C^i\to C^{i-1}$

$$0 \longrightarrow \operatorname{Spec}(\mathbb{R}) \xrightarrow{0} \operatorname{Spec}(\mathbb{R}) \xrightarrow{0} \operatorname{Spec}(\mathbb{R}) \longrightarrow 0$$

$$f_1 \circ g_1 \bigsqcup_{\langle 1 \rangle} \xrightarrow{h_2} 0 \bigsqcup_{\langle 1 \rangle} \xrightarrow{h_3} f_3 \circ g_3 \bigsqcup_{\langle 1 \rangle} \langle 1 \rangle$$

$$0 \longrightarrow \operatorname{Spec}(\mathbb{R}) \xrightarrow{0} \operatorname{Spec}(\mathbb{R}) \xrightarrow{0} \operatorname{Spec}(\mathbb{R}) \longrightarrow 0$$

such that, in particular, $\langle 1 \rangle - 0 = 0 \circ h_2 + h_3 \circ 0 = 0$, which is a contradiction.

Lemma 9.2.7. Let $K = \mathbb{F}_2$. The image of $\langle S \rangle$ under

$$\operatorname{Spc}(\pi^*) : \operatorname{Spc}(\operatorname{K}_b(\operatorname{Cor}^0_{\mathbb{R},K})) \to \operatorname{Spc}(\operatorname{K}_b(\operatorname{\widetilde{Cor}}^0_{\mathbb{R},K}))$$

is contained in supp $(\overline{M}(\operatorname{Spec}(\mathbb{C})))$.

Proof. It follows from [Bal05b, Proposition 3.6] and [BG22a, Theorem 3.14] that

$$(\operatorname{Spc}(\pi^*))^{-1}(\operatorname{supp}_{\operatorname{K}_b(\operatorname{\widetilde{Cor}}^0_{\mathbb{R},K})}(\widetilde{M}(\operatorname{Spec}(\mathbb{C})))) = \operatorname{supp}_{\operatorname{K}_b(\operatorname{Cor}^0_{\mathbb{R},K})}(\pi^*(\widetilde{M}(\operatorname{Spec}(\mathbb{C})))) = \sup_{\operatorname{K}_b(\operatorname{Cor}^0_{\mathbb{R},K})}(M(\operatorname{Spec}(\mathbb{C}))) = \langle S \rangle.$$

Now, we can apply $\operatorname{Spc}(\pi^*)$ on both sides and obtain

$$\operatorname{Spc}(\pi^*)(\langle S \rangle) = \operatorname{Spc}(\pi^*)(\operatorname{Spc}(\pi^*)^{-1}(\operatorname{supp}_{\operatorname{K}_b(\operatorname{\widetilde{Cor}}^0_{\mathbb{R},K})}(\widetilde{M}(\operatorname{Spc}(\mathbb{C}))))) \\ \subset \operatorname{supp}_{\operatorname{K}_b(\operatorname{\widetilde{Cor}}^0_{\mathbb{R},K})}(\widetilde{M}(\operatorname{Spcc}(\mathbb{C}))).$$

We can compute the preimage of $\langle M(\operatorname{Spec}(\mathbb{C})) \rangle$ under π^* .

Lemma 9.2.8. Let $K = \mathbb{F}_2$. The image of $\langle M(\operatorname{Spec}(\mathbb{C})) \rangle$ under

$$\operatorname{Spc}(\pi^*) : \operatorname{Spc}(\operatorname{K}_b(\operatorname{Cor}^0_{\mathbb{R},K})) \to \operatorname{Spc}(\operatorname{K}_b(\operatorname{\widetilde{Cor}}^0_{\mathbb{R},K}))$$

is the prime ideal $\langle \widetilde{M}(\operatorname{Spec}(\mathbb{C})) \rangle$.

Proof. Since $\pi^*(M(\operatorname{Spec}(\mathbb{C}))) = M(\operatorname{Spec}(\mathbb{C})) \in \langle M(\operatorname{Spec}(\mathbb{C})) \rangle$, we have that $\widetilde{M}(\operatorname{Spec}(\mathbb{C})) \in (\pi^*)^{-1}(\langle M(\operatorname{Spec}(\mathbb{C})) \rangle)$. Consequently, we obtain $\langle \widetilde{M}(\operatorname{Spec}(\mathbb{C})) \rangle \subset (\pi^*)^{-1}(\langle M(\operatorname{Spec}(\mathbb{C})) \rangle) = \operatorname{Spc}(\pi^*)(\langle M(\operatorname{Spec}(\mathbb{C})) \rangle)$.

For the other inclusion, we recall that $\operatorname{Cor}^{0}_{\mathbb{R},K}$ and $\operatorname{\widetilde{Cor}}^{0}_{\mathbb{R},K}$ have the same objects and the additive generators satisfy $\operatorname{Spec}(\mathbb{C}) \otimes_{\operatorname{Spec}(\mathbb{R})} \operatorname{Spec}(\mathbb{R}) \cong \operatorname{Spec}(\mathbb{C})$ and $\operatorname{Spec}(\mathbb{C}) \otimes_{\operatorname{Spec}(\mathbb{R})} \operatorname{Spec}(\mathbb{C}) \cong \operatorname{Spec}(\mathbb{C}) \oplus \operatorname{Spec}(\mathbb{C})$. Hence, complexes in $\langle M(\operatorname{Spec}(\mathbb{C})) \rangle$ and $\langle \widetilde{M}(\operatorname{Spec}(\mathbb{C})) \rangle$ have only (possibly 0) copies of $\operatorname{Spec}(\mathbb{C})$ in each degree. Note

that the tensor unit $M(\operatorname{Spec}(\mathbb{R}))$ considered a complex concentrated in degree 0 cannot appear as a direct summand of an object in $\langle \widetilde{M}(\operatorname{Spec}(\mathbb{C})) \rangle$. Otherwise, the thick tensor prime ideal $\operatorname{Spc}(\pi^*)(\langle M(\operatorname{Spec}(\mathbb{C})) \rangle)$ would not be proper, which is a contradiction. Analogously, $M(\operatorname{Spec}(\mathbb{R}))$ cannot appear as a direct summand of an object in $\langle M(\operatorname{Spec}(\mathbb{C})) \rangle$. The differentials are arbitrary morphisms in $\operatorname{Cor}_{\mathbb{R},K}^0$ and $\widetilde{\operatorname{Cor}}_{\mathbb{R},K}^0$, respectively, such that the composition of two consecutive differentials is 0.

Let $\widetilde{C} \in (\pi^*)^{-1}(\langle M(\operatorname{Spec}(\mathbb{C})) \rangle)$, i.e., $\pi^*(\widetilde{C}) = C \in \langle M(\operatorname{Spec}(\mathbb{C})) \rangle$. The functor π^* is given by applying the additive forgetful functor π degreewise, which is the identity on objects. Hence, $C_i = \widetilde{C}_i$ in each degree *i*. In particular, $\widetilde{C} \in \langle \widetilde{M}(\operatorname{Spec}(\mathbb{C})) \rangle$.

Remark 9.2.9. Applying Proposition 7.3.3 and using the fact that idempotent completion does not change the Balmer spectrum by [Bal05b, Proposition 3.13], we have computed one point of the Balmer spectrum $\operatorname{Spc}(\widetilde{\operatorname{DAM}}^{\operatorname{gm}}(\mathbb{R};K))$ for K of characteristic 2. The next step is to compute the images under $\operatorname{Spc}(\pi^*)$ of $\langle S \rangle$ and $\langle S, M(\operatorname{Spec}(\mathbb{C})) \rangle$, which we conjecture to be $\langle 0 \rangle$ and $\langle \widetilde{M}(\operatorname{Spec}(\mathbb{C})) \rangle$, respectively. In future work, one could try to find tensor prime ideals in $\operatorname{Spc}(K_b(\widetilde{\operatorname{Cor}}^0_{\mathbb{R},K}))$ that are not images under $\operatorname{Spc}(\pi^*)$.

However, one can show that the functor $\pi : \widetilde{\operatorname{Cor}}_{\mathbb{R},K}^{0} \to \operatorname{Cor}_{\mathbb{R},K}^{0}$ detects \otimes nilpotence for morphisms. That is, if $\pi(f) = 0$ for a morphism f in $\widetilde{\operatorname{Cor}}_{\mathbb{R},K}^{0}$, then $f^{\otimes n} = 0$ for some $n \geq 1$. One can also show that this property transfers to the level of chain complexes, but it is not clear whether it transfers to the level of homotopy categories as well. If it does, one can apply [Bal18, Theorem 1.3] and conclude that the induced map on spectra $\operatorname{Spc}(\pi^*)$ is surjective.

In this case, computing the its image already yields a complete description of the Balmer spectrum $\operatorname{Spc}(K_b(\operatorname{Cor}_{\mathbb{R},K}^0))$. If the above conjecture of the computation of the other preimages proves to be true, the spectrum would actually consist of *less* points than $\operatorname{Spc}(K_b(\operatorname{Cor}_{\mathbb{R},K}^0))$. This is unexpected, since "decorating with tilde" adds information and one would, in general, rather suspect a surjection in the opposite direction. However, we are in the special case of coefficients in a field K of characteristic 2. Therefore, unexpected results may appear. Still, the author expects $\operatorname{Spc}(\pi^*)$ not to be surjective in this case.

Remark 9.2.10. If the conjecture $\operatorname{Spc}(\pi^*)(\langle S \rangle) = \langle 0 \rangle$ proves to be true, it follows that $\langle 0 \rangle$ is a prime ideal in $\operatorname{K}_b(\operatorname{Cor}_{\mathbb{R},K}^0)$. In $\operatorname{K}_b(\operatorname{Cor}_{\mathbb{R},K}^0)$, this is not the case. We repeat the argument in $\operatorname{K}_b(\operatorname{perm}(C_2, K)) \simeq \operatorname{K}_b(\operatorname{Cor}_{\mathbb{R},K}^0)$ after applying Corollary 6.2.12. By Frobenius reciprocity,

$$kC_2 \otimes M \cong \operatorname{Ind}_*^{C_2} \operatorname{Res}_*^{C_2} M = \operatorname{Ind}_*^{C_2} 0 = 0$$

for any acyclic complex $M \in \langle S \rangle = \mathcal{K}_{b,ac}(\operatorname{Cor}^{0}_{\mathbb{R},K})$. Hence, in particular, $kC_2 \otimes S = 0$.

Bibliography

- [AF16] A. Asok, J. Fasel: Comparing Euler Classes. In: The Quarterly Journal of Mathematics 67, pp. 603–635. 2016.
- [BC+22] T. Bachmann, B. Calmès, F. Déglise, J. Fasel, P. A. Østvær: Milnor-Witt Motives. arXiv:2004.06634v2 [math.AG]. 2022.
- [Bal99] P. Balmer: Derived Witt Groups of a Scheme. In: Journal of Pure and Applied Algebra 141 (2), pp. 101–129. 1999.
- [Bal00] P. Balmer: Triangular Witt Groups. Part I: The 12-Term Localization Exact Sequence. In: K-Theory 19 (4), pp. 311–363. 2000.
- [Bal01] P. Balmer: Triangular Witt Groups. Part II: From Usual to Derived. In: Mathematische Zeitschrift 236 (2), pp. 351–382. 2001.
- [Bal02] P. Balmer: Presheaves of Triangulated Categories and Reconstruction of Schemes. In: Mathematische Annalen 324 (3), pp. 557–580. 2002.
- [Bal05a] P. Balmer: Witt Groups. In: E. M. Friedlander, D. R. Grayson (eds.): Handbook of K-Theory, Vol. 2, pp. 539–576. Springer Berlin-Heidelberg. 2005.
- [Bal05b] P. Balmer: The Spectrum of Prime Ideals in Tensor Triangulated Categories. In: Journal f
 ür die Reine und Angewandte Mathematik 588, pp. 149–168. 2005.
- [Bal07] P. Balmer: Supports and Filtrations in Algebraic Geometry and Modular Representation Theory. In: American Journal of Mathematics 129 (5), pp. 1227–1250. 2007.
- [Bal09] P. Balmer: Niveau Spectral Sequences on Singular Schemes and Failure of the Generalized Gersten Conjecture. In: Proceedings of the American Mathematical Society 137 (1), pp. 99–106. 2009.
- [Bal10a] P. Balmer: Spectra, Spectra, Spectra Tensor Triangular Spectra Versus Zariski Spectra of Endomorphism Rings. In: Algebraic and Geometric Topology 10 (3), pp. 1521–1563. 2010.
- [Bal10b] P. Balmer: Tensor Triangular Geometry. In: Proceedings of the International Congress of Mathematicians (ICM) 2, pp. 85–112. 2010.

- [Bal13] P. Balmer: Tensor Triangular Chow Groups. In: Journal of Geometry and Physics 72, pp. 3–6. 2013.
- [Bal18] P. Balmer: On the Surjectivity of the Map of Spectra Associated to a Tensor-Triangulated Functor. In: Bulletin of the London Mathematical Society 50 (3), pp. 487–495. 2018.
- [Bal19] P. Balmer: A Guide to Tensor-Triangular Classification. In: H. Miller (ed.): Handbook of Homotopy Theory, Chapter 4, pp.147–164. Chapman and Hall/CRC. 2019.
- [BBC09] P. Balmer, D. Benson, J. Carlson: Gluing Representations via Idempotent Modules and Constructing Endotrivial Modules. In: Journal of Pure and Applied Algebra 213 (2), pp. 173–193. 2009.
- [BG22a] P. Balmer, M. Gallauer: Three Real Artin-Tate Motives. In: Advances in Mathematics, 406, 108535. 2022.
- [BG22b] P. Balmer, M. Gallauer: The tt-Geometry of Permutation Modules. Part I: Stratification. arXiv:2210.08311v1 [math.RT]. 2022.
- [BG23a] P. Balmer, M. Gallauer: Permutation Modules, Mackey Functors, and Artin Motives. In: A. B. Buan, H. Krause, Ø. Solberg (eds.): Representation of Algebras and Related Structures: International Conference on Representations of Algebras, ICRA 2020, November 9-25, 2020, Chapter 3, pp. 37–75. EMS Press. 2023
- [BG23b] P. Balmer, M. Gallauer: The tt-Geometry of Permutation Modules. Part II: Twisted Cohomology. arXiv:2307.04398v1 [math.RT]. 2022.
- [BG24] P. Balmer, M. Gallauer: *The Spectrum of Artin Motives*. arXiv:2401.02722v1 [math.AG]. 2023.
- [BS01] P. Balmer, M. Schlichting: Idempotent Completion of Triangulated Categories. In: Journal of Algebra 236 (2), pp. 819–834. 2001.
- [BW02] P. Balmer, C. Walter: A Gersten-Witt Spectral Sequence for Regular Schemes. In: Annales Scientifiques de l'École Normale Supérieure 35 (1), pp. 127–152. 2002.
- [BM00] J. Barge, F. Morel: Groupe de Chow des Cycles Orientés et Classe d'Euler des Fibrés Vectoriels. In: Comptes Rendus de l'Académie des Sciences Paris, 330, pp. 287–290. 2000.
- [Ben98a] D.J. Benson: Representations and Cohomology: I. Basic Representation Theory of Finite Groups and Associative Algebras, Second Edition. Cambridge Studies in Advanced Mathematics 30, Cambridge University Press, Cambridge. 1998.

- [Ben98b] D.J. Benson: Representations and Cohomology: II. Cohomology of Groups and Modules, Second Edition. Cambridge Studies in Advanced Mathematics 31, Cambridge University Press, Cambridge. 1998.
- [BCR97] D. J. Benson, J. F. Carlson, J. Rickard: Thick Subcategories of the Stable Module Category. In: Fundamenta Mathematicae 153 (1), pp. 59– 80. 1997.
- [BIK12] D. J. Benson, S. Iyengar, H. Krause: Representation of Finite Groups: Local Cohomology and Support. Oberwolfach Seminars 43, Birkhäuser Verlag, Basel. 2012.
- [CF22] B. Calmès, J. Fasel: The Category of Finite Milnor-Witt Correspondences. In: Milnor-Witt Motives, arXiv:2004.06634v2 [math.AG]. 2022.
- [CH09] B. Calmès, J. Hornbostel: Tensor-Triangulated Categories and Duality.
 In: Theory and Applications of Categories 22 (6), pp. 136–198. 2009.
- [CH11] B. Calmès, J. Hornbostel: Push-Forwards for Witt Groups of Schemes.
 In: Commentarii Mathematici Helvetici 86 (2), pp. 437--468. 2011.
- [Car96] J. F. Carlson: Modules and Group Algebras. Notes by R. Suter. Lectures in Mathematics ETH Zürich, Birkhäuser Verlag, Basel. 1996.
- [CD15] D.-C. Cisinski, F. Déglise: Integral Mixed Motives in Equal Characteristic. In: Documenta Mathematica Extra Volume: Alexander S. Merkurjev's Sixtieth Birthday, pp. 145–194. 2015.
- [CD16] D.-C. Cisinski, F. Déglise: Étale Motives. In: Compositio Mathematica 152 (3), pp. 556–666. 2016.
- [Deg23] F. Déglise: Notes on Milnor-Witt K-Theory. arXiv:2305.18609v1 [math.AG]. 2023.
- [DF22] F. Déglise, J. Fasel: Milnor-Witt Motivic Complexes. In: Milnor-Witt Motives, arXiv:2004.06634v2 [math.AG]. 2022.
- [DS14] I. Dell'Ambrogio, G. Stevenson: Even More Spectra: Tensor Triangular Comparison Maps via Graded Commutative 2-Rings. In: Applied Categorical Structures 22, pp. 169–210. 2014.
- [Eis95] D. Eisenbud: Commutative Algebra with a View Toward Algebraic Geometry. Graduate Texts in Mathematics, Springer New York, NY. 1995.
- [Fas07] J. Fasel: The Chow-Witt Ring. In: Documenta Mathematica 12, pp. 275–312. 2007.
- [Fas08] J. Fasel: Groupes de Chow-Witt. In: Mémoires de la Société Mathématique de France 113. 2008.

- [Fas13] J. Fasel: The Projective Bundle Theorem for I^j-Cohomology. In: Journal of K-Theory 11 (2), pp. 413–464. 2013.
- [Fas20] J. Fasel: Lectures on Chow-Witt groups. In: Motivic Homotopy Theory and Refined Enumerative Geometry, pp. 83-121. In: Contemporary Mathematics 745. American Mathematical Society, Providence RI. 2020.
- [FO22] J. Fasel, P.-A. Østvær: A Cancellation Theorem for Milnor-Witt Correspondences. In: Milnor-Witt Motives, arXiv:2004.06634v2 [math.AG]. 2022.
- [FS09] J. Fasel, V. Srinivas: Chow-Witt Groups and Grothendieck-Witt Groups of Regular Schemes. In: Advances in Mathematics 221, pp. 302–329. 2009.
- [FY23] J. Fasel, N. Yang: On Tate Milnor-Witt Motives. arXiv:2307.03424v3 [math.AG]. 2023.
- [Ful98] W. Fulton: Intersection Theory, Second Edition. Ergebnisse der Mathematik und ihrer Grenzgebiete, Springer New York, NY. 1998.
- [Gal19] M. Gallauer: tt-Geometry of Tate Motives over Algebraically Closed Fields. In: Compositio Mathematica 155, pp. 1888–1923. 2019.
- [GN03] S. Gille, A. Nenashev: Pairings in Triangular Witt Theory. In: Journal of Algebra 261 (2), pp. 292–309. 2003.
- [Hal16] J. Hall: The Balmer Spectrum of a Tame Stack. In: Annals of K-Theory 1 (3), pp. 259–274. 2016.
- [Hap88] D. Happel: Triangulated Categories in the Representation Theory of Finite Dimensional Algebras. London Mathematical Society Lecture Note Series 119, Cambridge University Press, Cambridge. 1988.
- [Har77] R. Hartshorne: Algebraic Geometry. Graduate Texts in Mathematics. Springer New York, NY. 1977.
- [He58] A. Heller: Homological Algebra in Abelian Categories. In: Annals of Mathematics 68 (3), pp. 484–525. 1958.
- [He60] A. Heller: The Loop-Space Functor in Homological Algebra. In: Transactions of the American Mathematical Society 96, pp. 382–394. 1960.
- [HO18] J. Heller, K. M. Ormsby: Primes and Fields in Stable Motivic Homotopy Theory. In: Geometry and Topology 22, pp. 2187–2218. 2018.
- [Hor08] J. Hornbostel: Oriented Chow groups, Hermitian K-Theory and the Gersten Conjecture. In: Manuscripta Mathematica 125, pp. 273–284. 2008.

- [HPS97] M. Hovey, J. H. Palmieri, N. P. Strickland: Axiomatic Stable Homotopy Theory. In: Memoirs of the American Mathematical Society 128, No. 610, 1997.
- [HW19] J. Hornbostel, M. Wendt: Chow-Witt Rings of Classifying Spaces for Symplectic and Special Linear Groups. In: Journal of Topology 12 (3), pp. 916–966. 2019.
- [HS98] M. J. Hopkins, J. H. Smith: Nilpotence and Stable Homotopy Theory. II. In: Annals of Mathematics, 148 (1), pp. 1–49. 1998.
- [KY15] B. Kahn, T. Yamazaki: Correction to "Voevodsky's Motives and Weil Reciprocity", Duke Math. J. 162 (2013), 2751–2796. In: Duke Mathematical Journal 164, pp. 2093–2098. 2015.
- [KS06] M. Kashiwara, P. Schapira: Categories and Scheaves. Grundlehren der mathematischen Wissenschaften 332, Springer-Verlag Berlin-Heidelberg. 2006.
- [Kel99] B. Keller: On the Cyclic Homology of Exact Categories. In: Journal of Pure and Applied Algebra 136 (1), pp. 1–56. 1999.
- [Kel96] B. Keller: Derived Categories and their Uses. In: M. Hazewinkel (ed.): Handbook of Algebra, Vol. 1, pp. 671–701. Elsevier/North-Holland, Amsterdam. 1996.
- [Kel17] S. Kelly: Voevodsky Motives and ldh-Descent. Astérisque 391, Société Mathématique de France. 2017.
- [Kl16a] S. Klein: Chow Groups of Tensor Triangulated Categories. In: Journal of Pure and Applied Algebra 220 (4), pp. 1343–1381. 2016.
- [Kl16b] S. Klein: Intersection Products for Tensor Triangular Chow Groups. In: Journal of Algebra 449, pp. 497–538. 2016.
- [Lam05] T.Y. Lam: Introduction to Quadratic Forms over Fields. Graduate Studies in Mathematics 67, American Mathematical Society, Providence, RI. 2005.
- [MacL71] S. Mac Lane: Categories for the Working Mathematician, Second Edition. Graduate Texts in Mathematics, Springer New York, NY. 1971.
- [May01] J.P. May: The Additivity of Traces in Triangulated Categories. In: Advances in Mathematics 163, pp. 34–73. 2001.
- [MVW06] C. Mazza, V. Voevodsky, C. Weibel: Lecture Notes on Motivic Cohomology. Clay Mathematics Monographs 2, American Mathematical Society, Providence, RI. 2006.

- [Milne20] J. S. Milne: Algebraic Number Theory (v3.08). Available at: https://www.jmilne.org/math/CourseNotes/ANT.pdf. 2020.
- [Mil70] J. Milnor: Algebraic K-Theory and Quadratic Forms. In: Inventiones Mathematicae 9, pp.318–344. 1969/70.
- [MH73] J. Milnor, D. Husemoller: Symmetric Bilinear Forms. Ergebnisse der Mathematik und ihrer Grenzgebiete, 2. Folge, Band 73. Springer Berlin-Heidelberg. 1973.
- [Mor12] F. Morel: A¹-Algebraic Topology over a Field. Lecture Notes in Mathematics, Springer Berlin-Heidelberg. 2012.
- [Nee05] A. Neeman: The K-Theory of Triangulated Categories. In: Handbook of K-Theory, Vol. 1, pp. 1011–1078. Springer Berlin-Heidelberg. 2005.
- [Nee92] A. Neeman: The Connection Between the K-theory Localization Theorem of Thomason, Trobaugh and Yao and the Smashing Subcategories of Bousfield and Ravenel. In: Annales Scientifiques de l'École Normale Supérieure 25 (5), pp. 547—566. 1992.
- [PW86] C. Pedrini, C. Weibel: K-Theory and Chow Groups on Singular Varieties. In: Applications of Algebraic K-Theory to Algebraic Geometry and Number Theory Part I (Boulder, Colo., 1983), Contemporary Mathematics, American Mathematical Society, Providence, RI. 1986.
- [Pe13] T. J. Peter: Prime Ideals of Mixed Artin-Tate Motives. In: Journal of K-Theory 11 (2), pp. 331–349. 2013.
- [PS14] K. Ponto, M. Shulman: Traces in Symmetric Monoidal Categories. In: Expositiones Mathematicae 32 (3), pp. 248–273. 2014.
- [QSS79] H.-G. Quebbemann, W. Scharlau, M. Schulte: Quadratic and Hermitian Forms in Additive and Abelian Categories. In: Journal of Algebra 59, pp. 264–289. 1979.
- [Qui73] D. Quillen: Higher Algebraic K-Theory I. In: Bass, H. (ed.): Algebraic K-Theory I: Higher K-Theories, pp. 85–147. Lecture Notes in Mathematics 341, Springer, Berlin-Heidelberg. 1973.
- [Ric89] J. Rickard: Derived Categories and Stable Equivalences. In: Journal of Pure and Applied Algebra 61, pp. 303–317. 1989.
- [Rin84] C. M. Ringel: *Tame Algebras and Integral Quadratic Forms*. Lecture Notes in Mathematics, Springer-Verlag Berlin-Heidelberg. 1984.
- [Scha85] W. Scharlau: *Quadratic and Hermitian Forms*. Grundlehren der mathematischen Wissenschaften 270, Springer-Verlag Berlin-Heidelberg. 1985.

- [Sch06] M. Schlichting: Negative K-Theory of Derived Categories. In: Mathematische Zeitschrift 253, pp. 97–134. 2006.
- [Sch10a] M. Schlichting: Hermitian K-Theory of Exact Categories. In: Journal of K-Theory 5, pp. 105–165. 2010.
- [Sch10b] M. Schlichting: The Mayer-Vietoris Principle for Grothendieck-Witt Groups of Schemes. In: Inventiones Mathematicae 179, pp. 349–433. 2010.
- [Sch11] M. Schlichting: Higher Algebraic K-Theory. In: G. Cortiñas (ed.): Topics in Algebraic and Topological K-Theory, Lecture Notes in Math., Springer Berlin-Heidelberg. 2011.
- [Sch17] M. Schlichting: Hermitian K-Theory, Derived Equivalences, and Karoubi's Fundamental Theorem. In: Journal of Pure and Applied Algebra 221 (7), pp. 1729–1844. 2017.
- [SGA4] Théorie des topos et cohomologie étale des schémas. Tome 1: Théorie des topos. Séminaire de Géométrie Algébrique du Bois Marie 1963/64 (SGA 4), M. Artin, A. Grothendieck, et J. L. Verdier. Avec la collaboration de N. Bourbaki, P. Deligne et B. Saint-Donat. Lecture Notes in Mathematics 269, Springer Berlin-Heidelberg. 1972.
- [SGA6] Théorie des Intersections et Théorème de Riemann-Roch. Séminaire de Géométrie Algébrique du Bois Marie 1966/67 (SGA 6), Dirigé par P. Berthelot, A. Grothendieck, et L. Illusie. Avec la collaboration de D. Ferrand, J. P. Jouanolou, O. Jussila, S. Kleiman, M. Raynaud et J.P. Serre. Lecture Notes in Mathematics 225, Springer Berlin-Heidelberg. 1971.
- [Ste14] G. Stevenson: Subcategories of Singularity Categories via Tensor Actions. In: Compositio Mathematica 150 (2), pp. 229–272. 2014.
- [TaWa91] H. Tachikawa, T. Wakamatsu: Cartan Matrices and Grothendieck Groups of Stable Categories. In: Journal of Algebra 144, pp. 390–398. 1991.
- [ThWe95] J. Thévenaz, P. Webb: The Structure of Mackey Functors. In: Transactions of the American Mathematical Society 347 (6), pp. 1865-1961. 1995.
- [Tho97] R. W. Thomason: The Classification of Triangulated Subcategories. In: Compositio Mathematica 105 (1), pp. 1–27. 1997.
- [TT90] R. W. Thomason, T. Trobaugh: Higher Algebraic K-Theory of Schemes and of Derived Categories. In: P. Cartier et al. (eds): The Grothendieck Festschrift, Volume III: A Collection of Articles Written in Honor of

the 60th Birthday of Alexander Grothendieck, pp. 247–435. Birkhäuser Boston, MA. 1990.

- [Voe00] V. Voevodsky: Triangulated Categories of Motives over a Field. In: Cycles, Transfers, and Motivic Homology Theories, pp. 188–238. Annals of Mathematical Studies 143, Princeton University Press, Princeton, NJ. 2000.
- [Wal03] C. Walter: Grothendieck–Witt Groups of Triangulated Categories. Available at: https://www.maths.ed.ac.uk/~v1ranick/papers/trigw.pdf. 2003.
- [Wei13] C. A. Weibel: The K-Book. An Introduction to Algebraic K-Theory. Graduate Studies in Mathematics 145. American Mathematical Society, Providence, Rhode Island. 2013.
- [Wei95] C. A. Weibel: An Introduction to Homological Algebra. In: Cambridge Studies in Advanced Mathematics 38. Cambridge University Press. 1995.
- [Ya16] S. Yang: K-Theoretic Chow Groups of Derived Categories of Schemes on a Question by Green-Griffiths. arXiv:1311.0066v3 [math.AG]. 2016.
- [Yos83] T. Yoshida: On G-Functors (II): Hecke Operators and G-Functors. In: Journal of the Mathematical Society of Japan 35 (1), pp.179–190. 1983.